UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): February 15, 2024 (February 11, 2024)

DRAFTKINGS INC.

(Exact name of registrant as specified in its charter)

Nevada (State or other jurisdiction of incorporation)

001-41379 (Commission File Number) 87-2764212 (IRS Employer Identification No.)

222 Berkeley Street, 5th Floor

Boston, MA 02116
(Address of principal executive offices, including zip code)

Registrant's telephone number, including area code: (617) 986-6744			
Not Applicable (Former name or former address, if changed since last report)			
Check the appropriate box below if the Form 8-K filing is inter	nded to simultaneously satisfy the filing obliq	gation of the registrant under any of the following provisions:	
Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)			
Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)			
Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17	CFR 240.14d-2(b))		
Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17	CFR 240.13e-4(c))		
ecurities registered pursuant to Section 12(b) of the Act:			
cearities registered parsault to section 12(b) of the rect.			
Title of each class	Trading Symbol(s)	Name of each exchange on which registered	
•	Trading Symbol(s) DKNG	Name of each exchange on which registered The Nasdaq Stock Market LLC	
Title of each class	DKNG	The Nasdaq Stock Market LLC	
Title of each class Class A common stock, par value \$0.0001 per share indicate by check mark whether the registrant is an emerging growth company	DKNG	The Nasdaq Stock Market LLC	
Title of each class Class A common stock, par value \$0.0001 per share indicate by check mark whether the registrant is an emerging growth company in the c	DKNG DKNG y as defined in Rule 405 of the Securities Act	The Nasdaq Stock Market LLC of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities	
Title of each class Class A common stock, par value \$0.0001 per share indicate by check mark whether the registrant is an emerging growth company exchange Act of 1934 (§240.12b-2 of this chapter). Simerging growth company f an emerging growth company, indicate by check mark if the registrant has e	DKNG DKNG y as defined in Rule 405 of the Securities Act	The Nasdaq Stock Market LLC of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities	
Title of each class Class A common stock, par value \$0.0001 per share indicate by check mark whether the registrant is an emerging growth company exchange Act of 1934 (§240.12b-2 of this chapter). Simerging growth company f an emerging growth company, indicate by check mark if the registrant has e	DKNG DKNG y as defined in Rule 405 of the Securities Act	The Nasdaq Stock Market LLC of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities	

Item 1.01. Entry into a Material Definitive Agreement.

On February 11, 2024, DraftKings Inc., a Nevada corporation (the "Company" or "DraftKings"), JackPocket Inc., a Delaware corporation ("Jackpocket"), DraftKings Holdings Inc., a Nevada corporation and a direct wholly-owned subsidiary of DraftKings ("DK HoldCo"), Fortune Merger Sub Inc., a Delaware corporation and a direct wholly-owned subsidiary of DK HoldCo ("Merger Sub I"), Fortune Merger Sub LLC, a Delaware limited liability company and a direct wholly-owned subsidiary of DK HoldCo ("Merger Sub II" and, together with Merger Sub I, the "Merger Subs") and Shareholder Representative Services LLC, a Colorado limited liability company, solely in its capacity as representative, agent and attorney-in-fact of the Jackpocket securityholders, entered into an agreement and plan of merger and plan of reorganization (the "Merger Agreement").

The Merger Agreement and the transactions contemplated therein (the "<u>Transactions</u>") were (i) approved and declared fair and advisable to, and in the commercial interests of, DraftKings by the board of directors of DraftKings and (ii) approved and declared advisable and fair to, and in the best interests of, Jackpocket and its stockholders by the board of directors of Jackpocket (the "Jackpocket Board").

Merger Agreement

Transaction Structure & Merger Consideration

On the terms and subject to the conditions set forth in the Merger Agreement, Merger Sub I will merge with and into Jackpocket (the "Initial Merger"), and upon consummation of the Initial Merger, Merger Sub I will cease to exist and Jackpocket will become a direct wholly-owned subsidiary of DK HoldCo. As part of the same overall series of related transactions, promptly following the Initial Merger, Jackpocket (as the surviving company of the Initial Merger (the "Initial Surviving Company")) will merge with and into Merger Sub II (the "Subsequent Merger" and, together with the Initial Merger, the "Mergers"), and upon the consummation of the Subsequent Merger, the Initial Surviving Company will cease to exist, and Merger Sub II will survive as a direct wholly-owned subsidiary of DK HoldCo and change its name to "Jackpocket LLC".

On the terms and subject to the conditions set forth in the Merger Agreement, DraftKings has agreed to acquire Jackpocket for aggregate consideration of \$750,000,000, of which (a) approximately \$412,500,000 will be in the form of cash (the "Cash Consideration"), subject to customary purchase price adjustments, and (b) approximately \$337,500,000 will be in the form of shares (the "Stock Consideration") of Class A common stock of DraftKings, par value \$0.0001 per share (the "Class A Common Stock"); provided that (i) holders of shares of Jackpocket capital stock held by unaccredited investors will receive additional Cash Consideration in lieu of Stock Consideration in respect of their Jackpocket capital stock and (ii) holders of shares of Jackpocket capital stock held by accredited investors will receive merger consideration that reflects a corresponding reduction in Cash Consideration and increase in Stock Consideration in respect of their Jackpocket capital stock. The Stock Consideration will be subject to a collar mechanism based on the volume-weighted average of the closing price per share of the Class A Common Stock for the thirty (30) consecutive full trading days ending on the second trading day immediately preceding the closing date of the Initial Merger (the "Closing Trading Price"). In the event that the Closing Trading Price is greater than or equal to \$31.68 but less than or equal to \$42.86 or less than \$31.68, the aggregate Stock Consideration will be a fixed number of shares of Class A Common Stock and approximately 10,654,149 shares of Class A Common Stock and approximately 10,654,149 shares of Class A Common Stock and approximately 10,654,149 shares of Class A Common Stock and the Cash Consideration will be decreased to the minimum extent necessary such that the Stock Consideration comprises at least 40% of the aggregate consideration as so determined, subject to certain limitations described in the Merger Agreement. DraftKings intends to issue the shares of Class A Common Stock described herein in

Treatment of Jackpocket Options

At the Initial Effective Time (as defined in the Merger Agreement), each outstanding option to purchase shares of Jackpocket common stock, \$0.00001 par value (the "Jackpocket Shares", and each such option to purchase Jackpocket Shares a "Jackpocket Option"), under the Jackpocket Inc. 2013 Equity Incentive Plan, whether vested or unvested, by virtue of the Initial Merger and without any required action on the part of the holder thereof or the parties to the Merger Agreement, will be treated as follows:

- Each outstanding and unexercised Jackpocket Option that, as of immediately prior to the Initial Effective Time, has a per share exercise price that is less than the Per Share Closing Consideration (as defined in the Merger Agreement) and that is not subject to performance vesting conditions (collectively, the "In-the-Money Options") will be converted into an option to purchase a number of shares of Class A Common Stock, as determined in accordance with the terms of the Merger Agreement. Except as provided in the Merger Agreement, the In-the-Money Options will be governed by the same terms and conditions (including vesting and exercisability terms) as were applicable immediately prior to the Initial Effective Time.
- Each outstanding and unexercised Jackpocket Option that, as of immediately prior to the Initial Effective Time, has a per share exercise price that is equal to or greater than the Per Share Closing Consideration will be cancelled for no consideration.
- Each outstanding and unvested Jackpocket Option that is subject to performance vesting conditions as of immediately prior to the Initial Effective Time will be cancelled for no consideration or payment.

Representations and Warranties and Covenants

The Merger Agreement contains customary representations and warranties from DraftKings, Jackpocket, DK HoldCo and the Merger Subs and each of the foregoing has agreed to customary covenants, including, among others, covenants relating to the conduct of its business prior to the closing and the use of commercially reasonable efforts to consummate the Transactions.

Following the execution of the Merger Agreement, Jackpocket delivered to DraftKings written consents executed by Jackpocket stockholders holding a majority of the voting power of the issued and outstanding shares of Jackpocket capital stock and a majority of the issued and outstanding Jackpocket preferred stock (excluding the holders of Jackpocket Series D-3 preferred stock) adopting the Merger Agreement (the "Required Stockholder Approval").

Under the terms of the Merger Agreement, DraftKings will, within fifteen business days of the Closing, file a resale registration statement on Form S-3 or other appropriate form covering the resale of the shares of Class A Common Stock to be issued to eligible Jackpocket stockholders in the Transactions.

Among other things, the Merger Agreement also prohibits Jackpocket from soliciting, recommending or consummating any competing acquisition proposal with any third party. In addition, the Jackpocket Board may not change its recommendation that its stockholders approve the Transactions (a "Change of Recommendation"), and Jackpocket must notify DraftKings if it receives a competing acquisition proposal within 24 hours of receipt of such proposal.

Closing Conditions

The Closing is subject to customary closing conditions, including: (i) the adoption of the Merger Agreement by the holders of the requisite number of shares of Jackpocket capital stock, (ii) expiration or early termination of the waiting period (and any extension thereof) under the Hart-Scott-Rodino Antitrust Improvement Act of 1976 and the receipt of certain other regulatory approvals identified by the parties, (iii) the absence of any law or governmental order making illegal or prohibiting the consummation of the Transactions or causing the Transactions to be rescinded following their consummation, (iv) the accuracy of the representations and warranties contained in the Merger Agreement (subject to customary materiality qualifiers) of the other party, (v) performance in all material respects by the other party of its covenants under the Merger Agreement, (vi) the absence of any

Material Adverse Effect (as defined in the Merger Agreement) on the other party, (vii) continued employment of at least 80% of certain Jackpocket employees identified by the parties, which must include certain employees specified by the parties, (viii) the continued operations of Jackpocket and its subsidiaries in certain jurisdictions specified by the parties, (ix) Jackpocket's delivery to DraftKings of support agreements from Jackpocket stockholders comprising no less than ninety percent (90%) of the outstanding Jackpocket capital stock (which shall include, among other things, agreements not to transfer all or a portion of the Class A Common Stock received by such stockholders in the Transactions prior to the six-month anniversary of the Closing), (x) Jackpocket's delivery to DraftKings of written consents from Jackpocket stockholders sufficient to satisfy the Required Stockholder Approval (which has been received prior to the date of this Current Report on Form 8-K), (xi) Jackpocket's delivery to DraftKings of consents, amendments, waivers or other arrangements specified by the parties with respect to certain contracts identified by the parties and (xii) delivery by the parties of certain other customary closing agreements and certificates.

Termination Rights

The Merger Agreement may be terminated by DraftKings or Jackpocket under certain circumstances, including (i) by mutual written consent of DraftKings and Jackpocket, (ii) by either DraftKings or Jackpocket if (x) the Closing has not occurred on or before November 11, 2024, subject to two extensions of three months each under certain circumstances, or (y) the non-terminating party breaches the Merger Agreement such that certain closing conditions would not be satisfied, subject to a cure period or (iii) by DraftKings if (x) the Jackpocket Board has made a Change of Recommendation or (y) Jackpocket fails to timely deliver to DraftKings the written consents from Jackpocket stockholders sufficient to satisfy the Required Stockholder Approval.

The summary of the Merger Agreement and the transactions contemplated thereby in this Current Report on Form 8-K does not purport to be complete and is qualified in its entirety by the terms and conditions of the Merger Agreement, a copy of which is attached hereto as Exhibit 2.1 and is incorporated herein by reference.

Item 2.02. Results of Operations and Financial Condition.

On February 15, 2024, DraftKings issued a press release announcing the Company's financial results for the quarter and year ended December 31, 2023.

A copy of DraftKings' press release is attached hereto as Exhibit 99.1 and is hereby incorporated by reference in this Item 2.02. The information and exhibit contained in this Item 2.02 is being furnished and shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, nor shall it be deemed incorporated by reference in any filing under the Securities Act of 1933, as amended, except as shall be expressly set forth by specific reference in such filing.

Item 3.02. Unregistered Sale of Equity Securities.

The disclosure under Item 1.01 of this Current Report on Form 8-K relating to the Merger Agreement and the issuance of Class A Common Stock thereunder is incorporated into this Item 3.02 by reference.

Item 7.01. Regulation FD Disclosure.

On February 15, 2024 DraftKings and Jackpocket issued a joint press release, a copy of which is attached hereto as Exhibit 99.2, announcing DraftKings' and Jackpocket's entry into the Merger Agreement and related matters.

The information in this Item 7.01, including Exhibit 99.2, is being furnished and shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, nor shall it be deemed incorporated by reference in any filing under the Securities Act of 1933, as amended, except as shall be expressly set forth by specific reference in such filing. This Current Report on Form 8-K will not be deemed an admission as to the materiality of any information of the information in this Item 7.01, including Exhibit 99.2.

Cautionary Statement Regarding Forward-Looking Statements

The statements in this Current Report on Form 8-K may contain forward-looking statements within the meaning of Section 21E of the Securities Exchange Act of 1934, as amended, and the Private Securities Litigation Reform Act of 1995, including statements about the Company, Jackpocket and their respective industries that involve substantial risks and uncertainties. All statements, other than statements of historical fact, contained in this Current Report on Form 8-K, including statements regarding guidance, DraftKings' and Jackpocket's consummation of the Transactions and future results of operations or financial condition, strategic plans and focus, user growth and engagement, product initiatives, and the objectives and expectations of management for future operations (including launches in new jurisdictions and the expected timing thereof), are forward-looking statements. In some cases, you can identify forward-looking statements because they contain words such as "anticipate," "believe," "confident," "contemplate," "continue," "could," "estimate," "expect," "forecast," "going to," "intend," "may," "plan," "poised," "protential," "predict," "project," "propose," "should," "target," "will," or "would" or the negative of these words or other similar terms or expressions. DraftKings cautions you that the foregoing may not include all of the forward-looking statements made in this press release.

You should not rely on forward-looking statements as predictions of future events. DraftKings has based the forward-looking statements contained in this press release primarily on its current expectations and projections about future events and trends, including the current macroeconomic environment, that it believes may affect its and Jackpocket's business, financial conditions, and prospects. These forward-looking statements are not guarantees of future performance, conditions or results, and involve a number of known and unknown risks, uncertainties, assumptions and other important factors, many of which are outside DraftKings' and Jackpocket's control and that could cause actual results or outcomes to differ materially from those discussed in the forward-looking statements. Important factors, among others, that may affect actual results or outcomes include, but are not limited to, the outcome of any legal proceedings that may be instituted against DraftKings and Jackpocket following the announcement of the Merger Agreement and the Transactions; the inability to complete the Transactions, including due to failure to obtain the requisite approvals of applicable regulatory authorities or the failure to satisfy certain other conditions to closing in the Merger Agreement; the occurrence of any event, change or other circumstance that could give rise to the termination of the Merger Agreement or could otherwise cause the Transactions to fail to close; the risk that the Transactions disrupt current plans or operations as a result of the announcement and consummation of the Transactions; the ability to recognize the anticipated benefits of the Transactions, which may be affected by, among other things, competition and the ability of DraftKings and Jackpocket to grow and manage growth and retain its key employees; costs related to the Transactions, braftKings' and Jackpocket's abilities to execute their respective business plans and meet their respective projections; potential litigation involving DraftKings or Jack

Item 9.01. Financial Statements and Exhibits.

(d) Exhibits.

Exhibit Number

<u>2.1</u>	Agreement and Plan of Merger and Plan of Reorganization, dated as of February 11, 2024, by and among DraftKings Inc., DraftKings Holdings Inc., Fortune Merger Sub Inc., Fortune Merger Sub LLC, JackPocket Inc. and Shareholder Representative Services LLC.
<u>99.1</u>	Press Release, dated as of February 15, 2024, reporting financial results for the fourth quarter and year ended December 31, 2023,
<u>99.2</u>	Press Release, dated as of February 15, 2024, issued by DraftKings Inc. and JackPocket Inc.

^{*}Annexes, schedules and/or exhibits have been omitted pursuant to Item 601(b)(2) of Regulation S-K. The Company agrees to furnish supplementally a copy of any omitted attachment to the SEC on a confidential basis upon request.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: February 15, 2024 DRAFTKINGS INC.

By:

Name:

/s/ R. Stanton Dodge
R. Stanton Dodge
Chief Legal Officer and Secretary Title:

AGREEMENT AND PLAN OF MERGER AND PLAN OF REORGANIZATION

by and among

DRAFTKINGS INC.,

DRAFTKINGS HOLDINGS INC.,

FORTUNE MERGER SUB INC.,

FORTUNE MERGER SUB LLC,

JACKPOCKET INC.,

and

SHAREHOLDER REPRESENTATIVE SERVICES LLC,

Dated as of February 11, 2024

TABLE OF CONTENTS

		Page
ARTICLE I TH	HE MERGERS; CLOSING; INITIAL EFFECTIVE TIME; SUBSEQUENT EFFECTIVE TIME; PAYMENTS AT CLOSING	3
1.1	The Initial Merger; Effects of the Initial Merger	3
1.2	The Subsequent Merger; Effects of the Subsequent Merger	3
1.3	Closing	3
1.4	Initial Merger Effective Time and Subsequent Merger Effective Time	3
1.5	Governing Documents	4
1.6	Directors and Officers of the Initial Surviving Company	4
1.7	Directors and Officers of the Subsequent Surviving Company	4
1.8	Additional Actions	4
ARTICLE II P	URCHASE PRICE; EFFECT ON SHARES	5
2.1	Effect on Capital Stock and Merger Sub I	5
2.2	Effect of Subsequent Merger on Initial Surviving Company	6
2.3	Estimated Closing Statement.	7
2.4	Purchase Price Holdbacks	8
2.5	Closing Payment Procedures.	9
2.6	Post-Closing Statement	10
2.7	Post-Closing Adjustment Payments	13
2.8	Withholding Rights	13
2.9	No Interest	14
2.10	Treatment of Equity Awards	14
2.11	Treatment of Company Warrants	15
2.12	Equitable Adjustments	15
ARTICLE III F	REPRESENTATIONS AND WARRANTIES OF THE COMPANY	15
3.1	Organization, Good Standing and Qualification	16
3.2	Capital Structure.	16
3.3	Authority; Approval	18
3.4	Governmental Filings; No Violations; Certain Contracts, Etc.	18
3.5	Financial Statements; Filings.	19
3.6	Absence of Certain Changes	20

3.7

3.8 3.9 Legal Proceedings......21

Employee Benefits......21

		· · · · · · · · · · · · · · · · · · ·	Page
	3.10	Labor Matters	22
	3.11	Compliance with Laws; Licenses.	23
	3.12	Takeover Statutes	26
	3.13	Material Contracts	26
	3.14	Real Property.	29
	3.15	Environmental Matters	29
	3.16	Taxes	30
	3.17	Intellectual Property; IT Assets; Data Privacy	32
	3.18	Personal Property	34
	3.19	Insurance	35
	3.20	Suppliers and Gaming Authorities	35
	3.21	Related Party Transactions	35
	3.22	Bank Accounts	35
	3.23	Brokers and Finders	36
	3.24	Private Placement	36
	3.25	Accredited Investors	36
	3.26	Trade Allowances	36
	3.27	Working Capital	36
	3.28	Products	36
	3.29	Information	37
	3.30	Non-Reliance	37
	3.31	No Other Representations or Warranties	37
RTIC	LE IV F	REPRESENTATIONS AND WARRANTIES OF PARENT, DK HOLDCO, MERGER SUB I AND MERGER SUB II	37
	4.1	Organization, Good Standing and Qualification	38
	4.2	Parent Common Stock	38
	4.3	Authority; Approval	38
	4.4	Governmental Filings; No Violations.	38
	4.5	Public Filings	39
	4.6	Legal Proceedings	39
	4.7	Available Funds	40
	4.8	Merger Sub I and DK HoldCo	40
	4.9	Merger Sub II	40

	· · · · · · · · · · · · · · · · · · ·	Page
4.10	Brokers and Finders	40
4.11	Tax Matters	40
4.12	Capitalization	41
4.13	Absence of Certain Changes	41
4.14	Non-Reliance	41
4.15	No Other Representations or Warranties	41
ARTICLE V C	COVENANTS	42
5.1	Interim Operations.	42
5.2	Cooperation; Status Updates	46
5.3	Regulatory Filings and Approvals	46
5.4	No Solicitation of Acquisition Proposals or Change in Recommendation	49
5.5	Required Stockholder Approval	50
5.6	Stockholder Support Agreements	51
5.7	Information Statement	51
5.8	Third-Party Consents	52
5.9	Access and Reports; Retention of Books and Records	52
5.10	Publicity	53
5.11	Employee Benefits	53
5.12	Indemnification; Directors' and Officers' Insurance.	55
5.13	Confidentiality	56
5.14	Tax Matters	56
5.15	Takeover Statutes	61
5.16	R&W Insurance Policy.	61
5.17	Transfer of Private Placement Shares; Registration Statement	61
5.18	Director and Officer Resignations	65
5.19	Delivery of Data Room	65
5.20	Company Indebtedness	65
5.21	Lobbying	66
5.22	Continued Licensure	66
5.23	Further Assurances	66
ARTICLE VI	CONDITIONS	66
6.1	Conditions to Each Party's Obligation to Consummate the Transactions	66

Page

6.2	Conditions to Obligation of Parent, DK HoldCo, Merger Sub and Merger Sub I	66
6.3	Conditions to Obligation of the Company	69
ARTICLE V	II TERMINATION	
7.1	Termination by Mutual Written Consent	
7.2	Termination by Either Parent or the Company	
7.3	Termination by Parent	
7.4	Termination by the Company	71
7.5	Effect of Termination and Abandonment	71
7.6	Bridge Loan	71
7.7	Treatment of Extension Payment	73
ARTICLE V	III SURVIVAL; INDEMNIFICATION	73
8.1	Survival	73
8.2	Indemnification by Company Stockholders	74
8.3	Claim Procedures.	76
8.4	Recoveries	77
8.5	Indemnity Payments.	78
8.6	Exclusive Remedies	79
8.7	No Effect on R&W Insurance Policy	79
ARTICLE IX	K STOCKHOLDER REPRESENTATIVE	79
9.1	Acknowledgement	79
9.2	Stockholder Representative Expense Amount	81
9.3	Exculpation and Indemnification of Stockholder Representative	81
9.4	Reliance	82
9.5	Replacement of the Stockholder Representative	82
ARTICLE X	MISCELLANEOUS AND GENERAL	82
10.1	Modification or Amendment; Waiver	82
10.2	Expenses	82
10.3	Counterparts	83
10.4	Governing Law and Venue; Submission to Jurisdiction; Selection of Forum; Waiver of Trial by Jury; Specific Performance	83
10.5	Notices	84
10.6	Entire Agreement	85
10.7	No Third Party Beneficiaries	85

	P	age
10.8	Obligations of Parent and of the Company	85
10.9	Severability	85
10.10	Interpretation; Construction.	86
10.11	Successors and Assigns	87
10.12	Fulfillment of Obligations	87
10.13	Waiver of Conflicts; Privilege	87

SCHEDULES AND EXHIBITS

Schedule I: List of Voting Company Stockholders

Schedule II: List of Certain Voting Company Stockholders
Schedule III: List of Certain Holders of Company Options

Exhibit A: Definitions
Exhibit B: Written Consent
Exhibit C: Support Agreement

Exhibit D: Option Acknowledgment Agreement
Exhibit E: Illustrative Estimated Closing Statement
Exhibit F: Illustrative Working Capital Calculation
Exhibit G: Form of Non-Disclosure Agreement

Exhibit H: Management Incentive Plan

Exhibit I: Company Tax Representation Letter
Exhibit J: Parent Tax Representation Letter
Exhibit K: Selling Stockholder Questionnaire

AGREEMENT AND PLAN OF MERGER AND PLAN OF REORGANIZATION

This AGREEMENT AND PLAN OF MERGER AND PLAN OF REORGANIZATION (as amended or restated from time to time, this "Agreement"), dated as of February 11, 2024, is entered into by and among DraftKings Inc., a Nevada corporation ("Parent"), DraftKings Holdings Inc., a Nevada corporation and a direct and wholly-owned subsidiary of Parent ("DK HoldCo"), Fortune Merger Sub Inc., a Delaware corporation and a direct and wholly-owned subsidiary of DK HoldCo ("Merger Sub I"), Fortune Merger Sub LLC, a Delaware limited liability company and a direct and wholly-owned subsidiary of DK HoldCo ("Merger Sub II"), JackPocket Inc., a Delaware corporation (the "Company"), and Shareholder Representative Services LLC, a Colorado limited liability company, solely in its capacity as representative, agent and attorney-in-fact of the Company Securityholders (the "Stockholder Representative"). All of the signatories to this Agreement are collectively referred to as the "Parties" and individually as a "Party."

RECITALS

WHEREAS, the Parties intend that, on the terms and subject to the conditions set forth in this Agreement, Merger Sub I shall merge with and into the Company (the "<u>Initial Merger</u>"), with the Company surviving the Initial Merger as the surviving corporation (the "<u>Initial Surviving Company</u>"), pursuant to the provisions of the General Corporation Law of the State of Delaware (the "<u>DGCL</u>");

WHEREAS, the Parties intend that, on the terms and subject to the conditions set forth in this Agreement, promptly following the consummation of the Initial Merger, the Initial Surviving Company shall merge with and into Merger Sub II (the "Subsequent Merger", together with the Initial Merger, the "Mergers"), with Merger Sub II surviving the Subsequent Merger as the surviving company (the "Subsequent Surviving Company"), pursuant to the provisions of the DGCL and the Delaware Limited Liability Company Act (the "DLLCA");

WHEREAS, the board of directors of the Company (the "Company Board") has unanimously (a) approved and declared advisable this Agreement and the transactions contemplated by this Agreement (the "Transactions"), (b) determined that this Agreement and the Transactions are fair to, and in the best interests of, the Company and the holders of shares of the Company's common stock, \$0.00001 par value ("Common Stock"), Series Seed Preferred Stock, \$0.00001 par value ("Seed Stock"), Series A Preferred Stock, \$0.00001 par value ("Class A Stock"), Series B Preferred Stock, \$0.00001 par value ("Class B Stock"), Series B-1 Preferred Stock, \$0.00001 par value ("Class B-1 Stock"), Series C-1 Preferred Stock, \$0.00001 par value ("Class C-2 Stock"), Series C-3 Preferred Stock, \$0.00001 par value ("Class C-3 Stock"), Series C-3 Preferred Stock, \$0.00001 par value ("Class C-2 Stock, the "Class C Preferred Stock"), Series D-1 Preferred Stock, \$0.00001 par value ("Class D-1 Stock"), Series D-2 Preferred Stock, \$0.00001 par value ("Class D-2 Stock") and Series D-3 Preferred Stock, \$0.00001 par value ("Class D-3 Stock"), and together with the Common Stock, Seed Stock, Class A Stock, Class B Stock, Class B-1 Stock, Class C-1 Stock, Class C-2 Stock, Class C-3 Stock, Class D-1 Stock and Class D-2 Stock, the "Company Capital Stock"), (c) directed that this Agreement be submitted to the holders of Company Capital Stock (the "Company Stockholders") for their adoption and (d) resolved to recommend that the Company Stockholders adopt this Agreement;

WHEREAS, DK HoldCo, as the sole shareholder of Merger Sub I, has (a) determined that this Agreement and the Transactions are fair and advisable to, and in the commercial interests of, DK HoldCo and Merger Sub I, (b) approved and adopted this Agreement, and (c) authorized and approved the Transactions, including Merger Sub I's execution and delivery of this Agreement;

WHEREAS, DK HoldCo, as the sole member of Merger Sub II, has (a) determined that this Agreement and the Transactions are fair and advisable to, and in the commercial interests of, DK HoldCo and Merger Sub II, (b) approved and adopted this Agreement, and (c) authorized and approved the Transactions, including Merger Sub II's execution and delivery of this Agreement;

WHEREAS, the board of directors of Parent has (a) determined that this Agreement and the Transactions are fair and advisable to, and in the commercial interests of, Parent, (b) approved and adopted this Agreement, and (c) authorized and approved the Transactions, including Parent's execution and delivery of this Agreement;

WHEREAS, for U.S. federal and applicable state and local income Tax purposes, the Parties intend that (i) the Initial Merger and the Subsequent Merger be treated as a single integrated transaction that qualifies as a "reorganization" within the meaning of Section 368(a) of the Internal Revenue Code of 1986 (the "Code"), and the Treasury regulations promulgated thereunder, (ii) Parent, DK HoldCo and the Company each be a party to such reorganization within the meaning of Section 368(b) of the Code and (iii) this Agreement constitute a "plan of reorganization" within the meaning of Treasury regulations sections 1.368-2(g) and 1.368-3(a) (clauses (i)-(iii), collectively, the "Intended Tax Treatment");

WHEREAS, as an inducement to Parent's, DK HoldCo's, Merger Sub I's and Merger Sub II's willingness to enter into this Agreement, (i) prior to the end of the Initial Stockholder Consent Delivery Period, and as a condition to the obligations of Parent, DK HoldCo, Merger Sub I and Merger Sub II hereunder, the Company Stockholders listed on Schedule I constituting the Required Stockholder Approval shall have executed and delivered written consents in the form attached hereto as Exhibit B (each, a "Written Consent"), (ii) prior to the Closing Date, Company Stockholders holding no less than ninety percent (90%) of the issued and outstanding shares of Company Capital Stock shall have executed and delivered support agreements in the form attached hereto as Exhibit C (each, a "Support Agreement"), which, in the case of certain Company Stockholders so designated on Schedule II, shall include customary covenants regarding confidentiality, non-competition and non-solicitation (with such covenants to be effective as of the Closing), (iii) as a condition to payment of any amount owed to or the issuance of any securities to holders of Company Options hereunder, holders of Company Options shall have first executed and delivered option acknowledgment agreements in the form attached hereto as Exhibit D (each, an "Option Acknowledgment Agreement"), which, in the case of certain holders of Company Options so designated on Schedule III, shall include customary covenants regarding confidentiality, non-competition and non-solicitation (with such covenants to be effective as of the Closing) and (iv) concurrently with the execution and delivery of this Agreement, the Designated Stockholder has executed and delivered a Support Agreement and the Designated Optionholder has executed and delivered an Option Acknowledgment Agreement, which, in each case, include customary covenants regarding confidentiality, non-competition and non-solicitation (with such covenants to be effective as of the Closing);

WHEREAS, concurrently with the execution of this Agreement, and as a condition and inducement to Parent's, DK HoldCo's, Merger Sub I's and Merger Sub II's willingness to enter into this Agreement, the Designated Stockholder shall have executed an offer of employment letter (an "Offer Letter"), which, for the avoidance of doubt, shall include customary covenants regarding confidentiality and non-solicitation (with such covenants to be effective as of the Closing); and

WHEREAS, the Parties desire to make certain representations, warranties, covenants and agreements in connection with the transactions contemplated by this Agreement.

NOW, THEREFORE, in consideration of the promises, and of the representations, warranties, covenants and agreements contained herein, and for other good and valuable consideration, the receipt

and sufficiency of which are hereby acknowledged, accepted and agreed to, the Parties, intending to be legally bound, hereby agree as follows:

ARTICLE I

THE MERGERS; CLOSING; INITIAL EFFECTIVE TIME; SUBSEQUENT EFFECTIVE TIME; PAYMENTS AT CLOSING

- 1.1 The Initial Merger; Effects of the Initial Merger. On the terms and subject to the conditions set forth in this Agreement and the applicable provisions of the DGCL, (a) at the Initial Effective Time, Merger Sub I will be merged with and into the Company in accordance with the DGCL and the separate corporate existence of the Merger Sub I shall thereupon cease, (b) the Company will be the Initial Surviving Company in the Initial Merger and, from and after the Initial Effective Time, will be a direct wholly-owned subsidiary of DK HoldCo, which, in turn, will, from and after Initial Effective Time and until the Subsequent Effective Time, be a direct wholly-owned subsidiary of Parent, and the separate corporate existence of the Company with all of its rights, privileges, immunities, powers and franchises will continue unaffected by the Initial Merger as provided in the DGCL, and (c) from and after the Initial Effective Time, the Initial Merger shall have such other effects set forth in this Agreement and the applicable provisions of the DGCL.
- 1.2 The Subsequent Merger; Effects of the Subsequent Merger. On the terms and subject to the conditions set forth in this Agreement and the applicable provisions of the DGCL and the DLLCA, (a) immediately following the effectiveness of the Initial Merger, the Initial Surviving Company will be merged with and into Merger Sub II in accordance with the DGCL and the DLLCA and the separate corporate existence of Initial Surviving Company will thereupon cease, (b) Merger Sub II will be the Subsequent Surviving Company in the Subsequent Merger and, from and after the Subsequent Effective Time, will be a direct wholly-owned limited liability company subsidiary of DK HoldCo, which, in turn, will, from and after Subsequent Effective Time, be a direct wholly-owned subsidiary of Parent and the separate existence of Merger Sub II with all of its rights, privileges, immunities, powers and franchises will continue unaffected by the Subsequent Merger, as provided in the DGCL and the DLLCA, and (c) from and after the Subsequent Effective Time, the Subsequent Merger will have the effects set forth in this Agreement and the applicable provisions of the DGCL and the DLLCA.
- 1.3 Closing. The closing of the Initial Merger (the "Closing") shall take place by remote communication and by the exchange of signatures by electronic transmission and to the extent a physical location is required or exchange by electronic transmission is not practicable, at the offices of Sullivan & Cromwell LLP, 125 Broad Street, New York, New York 10004, at 9:00 a.m. prevailing Eastern Time on the third (3rd) Business Day following the day on which the last to be satisfied or waived of the conditions set forth in Article VI (other than those conditions that by their nature are to be satisfied or waived at the Closing, but subject to the satisfaction or waiver of those conditions) have been satisfied or waived in accordance with this Agreement or at such other date, time or place (including by means of remote communication) as the Parties may mutually agree in writing (the date on which the Closing actually occurs, the "Closing Date").
- 1.4 <u>Initial Merger Effective Time and Subsequent Merger Effective Time.</u> On the date of the Closing, the Company and DK HoldCo shall (a) cause the certificate of merger relating to the Initial Merger (the "<u>Certificate of Initial Merger</u>") to be executed and filed with the Secretary of State of the State of Delaware as provided in Section 251 of the DGCL and (b) deliver and tender, or cause to be delivered or tendered, as applicable, any Taxes and fees and make all other filings or recordings required under the DGCL in connection with such filing of the Certificate of Initial Merger and the consummation of the Initial Merger, which shall become effective at the date and time when the Certificate of Initial

Merger has been executed and filed pursuant to this Section 1.4, or at such later date and time as may be agreed by the Parties in writing and specified in the Certificate of Initial Merger so executed and filed (such date and time, as applicable, the "Initial Effective Time"). Immediately following the Initial Effective Time, DK HoldCo shall (i) cause the certificate of merger relating to the Subsequent Merger (the "Certificate of Subsequent Merger") to be executed and delivered for filing with the Secretary of State of the State of Delaware as provided in Section 18-209 of the DLLCA and Section 251 of the DGCL and (ii) deliver and tender, or cause to be delivered or tendered, as applicable, any Taxes and fees and make all other filings or recordings required under the DGCL and the DLLCA in connection with such filing of the Certificate of Subsequent Merger and the consummation of the Subsequent Merger, which shall become effective at the date and time when the Certificate of Subsequent Merger has been executed and filed pursuant to this Section 1.4, or at such later date and time as may be agreed by the Parties in writing and specified in the Certificate of Subsequent Merger so executed and filed (such date and time, as applicable, the "Subsequent Effective Time").

- 1.5 Governing Documents. At the Initial Effective Time, the certificate of incorporation and bylaws of the Initial Surviving Company shall be amended and restated to be identical to those of Merger Sub I, as in effect immediately prior to the Initial Effective Time, except that the name of the Initial Surviving Company shall be "Jackpocket Inc." At the Subsequent Effective Time, the certificate of formation and limited liability company operating agreement of Merger Sub II, as in effect immediately prior to the Subsequent Effective Time, shall each remain in effect as the certificate of formation and limited liability company operating agreement of the Subsequent Surviving Company, except that the name of the Subsequent Surviving Company shall be "Jackpocket LLC".
- 1.6 Directors and Officers of the Initial Surviving Company. The Parties shall take all actions necessary so that, effective as of the Initial Effective Time, (i) the directors of Merger Sub I immediately prior to the Initial Effective Time shall become and constitute the only directors of the Initial Surviving Company, each to hold office until his or her successor has been duly elected or appointed and qualified or until his or her earlier death, resignation or removal in accordance with the articles of incorporation and bylaws of the Initial Surviving Company and applicable Law, and (ii) the officers of Merger Sub I immediately prior to the Initial Effective Time shall become and constitute the only officers of the Initial Surviving Company, each to hold office until his or her successor has been duly elected or appointed and qualified or until his or her earlier death, resignation or removal in accordance with the articles of incorporation and the bylaws of the Initial Surviving Company and applicable Law.
- 1.7 Directors and Officers of the Subsequent Surviving Company. The Parties shall take all actions necessary so that, effective as of the Subsequent Effective Time, (i) the managers of Merger Sub II immediately prior to the Subsequent Effective Time shall become and constitute the managers of the Subsequent Surviving Company, each to hold office until his or her successor has been duly elected or appointed and qualified or until his or her earlier death, resignation or removal in accordance with the certificate of formation and limited liability company agreement of the Subsequent Surviving Company and applicable Law, and (ii) the officers of Merger Sub II immediately prior to the Subsequent Effective Time shall become and constitute the only officers of the Subsequent Surviving Company, each to hold office until his or her successor has been duly elected or appointed and qualified or until his or her earlier death, resignation or removal in accordance with the limited liability company agreement of the Subsequent Surviving Company and applicable Law.
- 1.8 <u>Additional Actions</u>. The Subsequent Surviving Company may, at any time after the Subsequent Effective Time, take any action, including executing and delivering any document, in the name and on behalf of any of the Company, Merger Sub I, the Initial Surviving Company or Merger Sub II, in order to consummate the Transactions.

ARTICLE II

PURCHASE PRICE: EFFECT ON SHARES

- 2.1 <u>Effect on Capital Stock and Merger Sub I.</u> At the Initial Effective Time, as a result of the Initial Merger and without any action on the part of any of the Parties or any holder of Company Capital Stock:
- (a) Merger Sub I. Each share of common stock, par value \$0.0001 per share, of Merger Sub I that is issued and outstanding immediately prior to the Initial Effective Time shall be converted into one share of common stock, par value \$0.0001 per share, of the Initial Surviving Company.

(b) Conversion of Company Capital Stock.

- (i) Each issued and outstanding share of Company Capital Stock held of record by an Accredited Investor (other than Excluded Shares, if any) shall be converted into and shall thereafter represent the right of the holder thereof to receive the following: (1) a number of shares of Parent Common Stock equal to the Per Share Portion multiplied by the Merger Stock Consideration, as adjusted pursuant to Section 2.1(b)(iv) (the "Accredited Investor Closing Stock Consideration") and (2) an amount in cash equal to the Per Share Portion multiplied by the Estimated Merger Cash Consideration, as adjusted pursuant to Section 2.1(b)(iv) (the "Accredited Investor Closing Cash Consideration") (clauses (1) and (2) collectively, the "Accredited Investor Company Capital Stock Closing Consideration").
- (ii) Each issued and outstanding share of Company Capital Stock held of record by a Non-Accredited Investor (other than the Excluded Shares) shall be converted into and shall thereafter represent the right of the holder thereof to receive an amount in cash equal to the following: (1) the product of (A) the Per Share Portion, multiplied by (B) the Merger Stock Consideration, multiplied by (C) the Closing Parent Trading Price (the "Non-Accredited Investor Closing Stock Consideration") and (2) an amount in cash equal to the Per Share Portion multiplied by the Estimated Merger Cash Consideration (the "Non-Accredited Investor Closing Cash Consideration") (clauses (1) and (2) collectively, the "Non-Accredited Investor Company Capital Stock Closing Consideration" and together with the Accredited Investor Company Capital Stock Closing Consideration, the "Company Capital Stock Closing Consideration", and the Company Capital Stock Closing Consideration to be paid in respect of each share of Company Capital Stock, as applicable the "Per Share Closing Consideration").
- (iii) At the Initial Effective Time, all shares of Company Capital Stock issued and outstanding immediately prior to the Initial Effective Time shall no longer be outstanding and shall automatically be cancelled and retired and shall cease to exist, and each certificate (a "Certificate") formerly representing any of the shares of Company Capital Stock (other than Excluded Shares) thereof shall cease to have any rights with respect thereto except the right to receive the Per Share Merger Consideration.
- (iv) To account for the fact that Non-Accredited Investors are not entitled to receive any Merger Stock Consideration hereunder: (1) the aggregate Accredited Investor Closing Cash Consideration shall be reduced by the aggregate amount of cash consideration included in the Non-Accredited Investor Closing Stock Consideration; and (2) the aggregate Accredited Investor Closing Stock Consideration shall be increased by the number of shares of Parent Common Stock

equal to (A) the aggregate Per Share Portion of the Fully Diluted Non-Option Shares held by Non-Accredited Investors, multiplied by (B) the Merger Stock Consideration.

- (c) <u>Cancellation of Excluded Shares</u>. Each Excluded Share shall cease to be outstanding, be cancelled without payment of any consideration therefor and shall cease to exist, subject to any rights the holder thereof may have under Section 2.1(d).
- Subject to the last sentence of this Section 2.1(d), no (d) Appraisal Rights. Dissenting Stockholder shall be entitled to receive the Per Share Merger Consideration with respect to the Dissenting Shares, and each Dissenting Stockholder shall be entitled to receive only the payment provided by Section 262 of the DGCL with respect to the Dissenting Shares owned by such Dissenting Stockholder and such Dissenting Stockholder shall cease to have any other rights with respect to such Dissenting Shares. The Company shall give Parent (i) prompt written notice and copies of any written demands for appraisal, actual, attempted or purported withdrawals of such demands, and any other instruments served pursuant to (or purportedly pursuant to) applicable Law that are received by the Company relating to the Company Stockholders' demands of appraisal, and (ii) the opportunity to direct negotiations and Proceedings with respect to demands for appraisal, including any determination to make any payment or deposit with respect to any of the Dissenting Stockholders with respect to any of their Dissenting Shares under Section 262(h) of the DGCL prior to the entry of judgment in the Proceedings regarding appraisal; provided that, after the Closing, the Stockholder Representative, on behalf of the Company Securityholders, shall be entitled, at the Company Securityholders' expense, to participate in all such negotiations and Proceedings, and Parent shall consider in good faith input from the Stockholder Representative. The Company shall not, except with the prior written consent of Parent (which shall not be unreasonably withheld, conditioned or delayed), and Parent shall not, except with the prior written consent of the Company (prior to the Closing) or the Stockholder Representative (from and after the Closing) (in each case which shall not be unreasonably withheld, conditioned or delayed) make any payment or deposit with respect to any demands for appraisal, offer to settle or settle any such demands or approve any withdrawal of any such demands or agree, authorize or commit to do any of the foregoing. If any Dissenting Stockholder shall have effectively withdrawn or otherwise waived or lost the right under Section 262 of the DGCL with respect to any Dissenting Shares, such Dissenting Shares shall thereupon be converted into the right to receive the Per Share Merger Consideration with respect to such Company Capital Stock pursuant to this Article II.
- (e) No Fractional Shares. Notwithstanding anything to the contrary contained in this Agreement, no fractional shares of Parent Common Stock shall be issued to the holders of Company Capital Stock in connection with the Initial Merger and any such fractional share interests shall not entitle the owner thereof to vote or to any other rights of a holder of shares of Parent Common Stock. Parent shall pay to each Company Stockholder that would otherwise have been entitled to receive a fraction of a share of Parent Common Stock (after aggregating all shares represented by the Certificates delivered by such holder), in lieu thereof and upon surrender thereof, an amount in cash (without interest) equal to the product obtained by multiplying (i) the fraction of a share of Parent Common Stock to which such holder would otherwise be entitled (after taking into account all shares of Company Capital Stock held by such holder immediately prior to the Initial Effective Time) by (ii) the Closing Parent Trading Price. All calculations pursuant to or in connection with this Agreement or the Payments Administrator Agreement involving the determination of shares of Parent Common Stock, including the calculation of Aggregate Stock Consideration and Per Share Closing Consideration shall be calculated to the nearest sixth decimal place.
- 2.2 <u>Effect of Subsequent Merger on Initial Surviving Company</u>. At the Subsequent Effective Time, as a result of the Subsequent Merger and without any action on the part of any of the Parties or any holder of stock of the Initial Surviving Company, each share of common stock, par value \$0.0001 per

share, of the Initial Surviving Company issued and outstanding immediately prior to the Subsequent Effective Time shall be converted into one limited liability company unit of the Subsequent Surviving Company.

2.3 Estimated Closing Statement.

- (a) <u>Delivery of the Estimated Closing Statement and Funds Flow Memorandum</u>. No later than five (5) Business Days prior to the Closing Date, the Company shall prepare and deliver to Parent:
 - (i) a written statement (the "<u>Estimated Closing Statement</u>"), prepared in accordance with <u>Exhibit E</u> (which sets forth an illustrative calculation of the Estimated Closing Statement as of the date of the Most Recent Balance Sheet (as defined below)), setting forth the Company's good faith calculation of the Estimated Merger Cash Consideration and each component thereof, including as separate line items (a) Net Working Capital (which shall be prepared in a manner consistent with the illustrative calculation set forth on <u>Exhibit F</u>) and the amount of the Net Working Capital Difference, (b) Cash on Hand, (c) Closing Indebtedness, and (d) (1) Company Transaction Expenses and (2) the Company Transaction Expenses that are Company Transaction Expenses Paid by Parent, together with reasonably detailed supporting documentation used by the Company (which shall be reasonably satisfactory to Parent) in calculating the amounts set forth in the Estimated Closing Statement;
 - (ii) the Funds Flow Memorandum, prepared by the Company in good faith based on the Company's books and records; and
 - (iii) a certificate of the Chief Executive Officer and Chief Financial Officer of the Company, dated as of the date of delivery of the Estimated Closing Statement, certifying that such officers have reviewed the Estimated Closing Statement and the underlying supporting documentation and that based on such review and such persons' understanding of the Company's business and financial position, the Estimated Merger Cash Consideration represents such officers' good faith estimate thereof.
 - (iv) <u>Cooperation</u>. Promptly following the Company's delivery of the Estimated Closing Statement and the Funds Flow Memorandum to Parent, the Company shall, and shall cause its Subsidiaries to, upon reasonable prior notice from Parent, (A) provide Parent and its authorized representatives, in connection with Parent's review of the Estimated Closing Statement and the Funds Flow Memorandum, with access during normal business hours to the facilities, books and records and work papers of the Company and (B) cooperate with and assist Parent and its authorized representatives in connection with the review of such materials, including by making available its and its Affiliates' employees, accountants and other personnel to the extent reasonably requested.
 - (v) Objection to the Estimated Closing Statement and Funds Flow Memorandum and Resolution of Disputes. If Parent notifies the Company in writing of an objection to the (A) Estimated Closing Statement or any of the amounts included in the calculation of the Estimated Merger Cash Consideration set forth therein and/or (B) the Funds Flow Memorandum (which written notice shall include the specific items in the Estimated Closing Statement and/or Funds Flow Memorandum that are in dispute and the nature and amount of any disagreement so identified) prior to the Business Day immediately prior to the Closing Date, then Parent and the Company shall seek in good faith to agree to revisions to the Estimated Closing Statement and/or Funds Flow Memorandum to resolve such objection and, to the extent Parent and Company are able to agree on any revisions, the Company shall update and redeliver the Estimated Closing Statement and/or the

Funds Flow Memorandum to reflect any such agreements, along with an updated officer's certificate required by Section 2.3(a)(iii) not later than the Business Day immediately prior to the Closing Date. If Parent has validly provided notice of an objection to the Estimated Closing Statement and/or the Funds Flow Memorandum pursuant to this Section 2.3(a)(v) and Parent and the Company fail to mutually agree upon revisions to the Estimated Closing Statement and/or the Funds Flow Memorandum on or prior to the Business Day immediately prior to the Closing Date, then (A) neither Parent nor the Company shall delay the Closing because of such failure and (B) the amounts set forth in the Estimated Closing Statement, without any adjustment, shall be the amounts used in the determination of the Estimated Merger Cash Consideration. The agreement of the Parties to revisions to the Estimated Closing Statement and/or the Funds Flow Memorandum or the failure of the Parties to agree to such revisions shall not constitute a waiver or limitation of a Party's rights and obligations pursuant to Section 2.7.

2.4 Purchase Price Holdbacks.

- (a) Purchase Price Adjustment Holdback. At the Initial Effective Time, Parent shall deposit, or cause to be deposited, \$5,000,000 of the Aggregate Cash Consideration with Computershare Trust Company, N.A. (in such capacity, the "Escrow Agent"), which shall constitute a holdback used to satisfy, among other matters, amounts owed to Parent from the Company Securityholders pursuant to Section 2.7(a) (the "Purchase Price Adjustment Holdback"). Parent will pay, or cause to be paid, all fees, costs and expenses of the Escrow Agent. Upon joint written instructions from Parent and the Stockholder Representative (or otherwise in accordance with the procedures set forth in the Escrow Agreement), the Escrow Agent shall release the Purchase Price Adjustment Holdback in accordance with Section 2.7 and the Escrow Agreement to the Payments Administrator. The Payments Administrator shall distribute the portion of any Per Share Post-Closing Adjustment Amount that is distributable to the Company Securityholders in accordance with Section 2.7, the Payments Administrator Agreement and the Funds Flow Memorandum. Parent will pay, or cause to be paid, all fees, costs and expenses of the Payments Administrator will be paid by DK HoldCo.
- Retention Holdback. At the Initial Effective Time, Parent shall deposit, or cause to be deposited, \$1,875,000 (the "Retention Holdback Amount") of the Aggregate Cash Consideration with the Escrow Agent, which shall constitute a holdback used to satisfy, among other matters, any indemnification amounts owed to Parent or any other Parent Indemnified Party from the Company Stockholders. Upon joint written instructions from Parent and the Stockholder Representative (or otherwise in accordance with the procedures set forth in the Escrow Agreement), an amount, if any, that is in excess of eighty percent (80%) of the initial aggregate Retention Holdback Amount shall be released by the Escrow Agent to the Payments Administrator within ten (10) Business Days following the date that is twelve (12) months after the Closing Date (the "First Release Date"), and the remainder of the Retention Holdback, if any, shall be released by the Escrow Agent to the Payments Administrator within ten (10) Business Days following the date that is eighteen (18) months after the Closing Date (the "Second Release Date" and each of the Second Release Date and First Release Date, a "Release Date"), in each case, for further distribution by the Payments Administrator to the Company Securityholders no later than ten (10) Business Days following the applicable Release Date: provided that such portion of the Retention Holdback Amount as may be necessary in the reasonable good faith of Parent to satisfy any unresolved or unsatisfied claims for Losses validly made under Article VIII before the applicable Release Date shall remain in the Retention Holdback and the applicable portion of the Retention Holdback Amount that is subject to any such claimed Losses shall not be so released until such the amount of such Losses are finally resolved and satisfied in accordance with Article VIII. The Payments Administrator shall distribute the portion of any Per Share Retention Holdback Consideration that is distributable to the Company Securityholders in accordance with the terms of this Agreement, the Payments Administrator Agreement and the Funds Flow Memorandum promptly after its release from the Retention Holdback.

Reserve Holdback. At the Initial Effective Time, Parent shall deposit, or cause to be deposited, \$3,000,000 (the "Reserve Holdback Amount") of the Aggregate Cash Consideration with the Escrow Agent, which shall constitute a holdback (the "Reserve Holdback") used to satisfy, among other things, the matters set forth in Section 2.4(c) of the Company Disclosure Letter (the "Reserved Matter"). Upon joint written instructions from Parent and the Stockholder Representative (or otherwise in accordance with the procedures set forth in the Escrow Agreement), the Reserve Holdback Amount shall be released by the Escrow Agent to the Payments Administrator within ten (10) Business Days following the date that is eighteen (18) months after the Closing Date (the "Reserve Release Date") for further distribution by the Payments Administrator to the Company Securityholders no later than ten (10) Business Days following the Reserve Release Date; provided that such portion of the Reserve Holdback Amount as is necessary in the reasonable good faith of Parent to satisfy any unresolved or unsatisfied Third Party Claims for Losses validly made under Article VIII pursuant to any Third Party Claim or communication made by any Governmental Entity in respect of the State of Texas with respect to the Reserved Matter after the date hereof and before the Reserve Release Date shall remain in the Reserve Holdback and the applicable portion of the Reserve Holdback Amount that is subject to any such claimed Losses shall not be so released until the amount of such claimed Losses are finally resolved and satisfied in accordance with Article VIII. The Payments Administrator shall distribute the portion of any Per Share Reserve Holdback Consideration that is distributable to the Company Securityholders in accordance with the terms of this Agreement, the Payments Administrator Agreement and the Funds Flow Memorandum promptly after its release from the Reserve Holdback.

2.5 Closing Payment Procedures.

- (a) Payments Administrator. On or prior to the Closing Date, Parent, the Stockholder Representative and Computershare N.A. (in such capacity, the "Payments Administrator") shall enter into an agreement with customary terms in form and substance reasonably acceptable to each of Parent and the Stockholder Representative (the "Payments Administrator Agreement") pursuant to which, among other things, on the Closing Date, Parent shall deposit, or cause to be deposited, the Closing Payments Fund with the Payments Administrator for the benefit of the Company Securityholders, to be held and disbursed by the Payments Administrator in accordance with this Agreement, the Funds Flow Memorandum prepared by the Company and approved by Parent (such approval not to be unreasonably withheld, conditioned or delayed), and the Payments Administrator Agreement. The Payments Administrator shall hold the Closing Payments Fund for the sole purpose of delivering the Per Share Closing Consideration to the Company Securityholders. Any portion of the amount deposited by Parent (or caused to be deposited by Parent) that remains unclaimed by the applicable Company Securityholders twelve (12) months after the Initial Effective Time will, to the extent permitted by applicable Law, be returned to Parent or its designee upon demand.
- (b) <u>Closing Payments Fund</u>. At the Closing, Parent shall deposit, or cause to be deposited, with the Payments Administrator, for the benefit of the Company Securityholders, the following (collectively, the "<u>Closing Payments Fund</u>"):
 - (i) an amount in cash equal to the Estimated Merger Cash Consideration;
 - (ii) the Aggregate Stock Consideration; and
 - (iii) an amount in cash equal to the product obtained by multiplying (x) the aggregated fractional shares of Parent Common Stock payable to Company Stockholders who are Accredited Investors as part of the applicable Accredited Investor Closing Stock Consideration by (y) the Closing Parent Trading Price.

- (c) <u>Stockholder Representative Expense Amount</u>. At the Closing, Parent shall deposit, or cause to be deposited, in an account designated by the Stockholder Representative, an amount in cash equal to the Stockholder Representative Expense Amount, to be solely held, safeguarded and disbursed by the Stockholder Representative pursuant to Section 9.2.
- To the extent that the Company does not deliver Exchange Procedures. Certificates (or affidavits of loss in lieu of the Certificates as provided in Section 2.5(f)) to the Payments Administrator on or before the Closing Date, then as promptly as practicable after the Initial Effective Time, the Subsequent Surviving Company shall cause the Payments Administrator to mail to each holder of record of Company Capital Stock (other than holders of Excluded Shares) (i) a letter of transmittal ("Letter of Transmittal") and (ii) instructions for effecting the surrender of the Certificates (or affidavits of loss in lieu of the Certificates as provided in Section 2.5(f)) in exchange for the Per Share Merger Consideration. Upon surrender of a Certificate (or affidavit of loss in lieu of the Certificate as provided in Section 2.5(f)) to the Payments Administrator in accordance with this Section 2.5(d) or the terms of such Letter of Transmittal, duly executed, the holder of such Certificate shall be entitled to receive in exchange therefor the Per Share Merger Consideration in respect of each share of Company Capital Stock represented by such Certificate (or affidavit of loss in lieu of the Certificate as provided in Section 2.5(f)), and the Certificate so surrendered shall forthwith be cancelled. No interest will be paid or accrued on any amount payable upon due surrender of the Certificates. In the event of a transfer of ownership of a share of Company Capital Stock that is not registered in the transfer records of the Company, the aggregate Per Share Merger Consideration to be exchanged upon due surrender of the Certificate may be issued to such transferee if the Certificate formerly representing such share of Company Capital Stock is presented to the Payments Administrator, accompanied by all documents required to evidence and effect such transfer and to evidence that any applicable stock Transfer Taxes have been paid or are not applicable.
- (e) <u>Transfers</u>. From and after the Initial Effective Time, there shall be no transfers on the stock transfer books of the Company of the shares of Company Capital Stock that were outstanding immediately prior to the Initial Effective Time. If, after the Initial Effective Time, any Certificate is presented to the Subsequent Surviving Company, Parent or the Payments Administrator for transfer, it shall be cancelled and exchanged for the aggregate Per Share Merger Consideration to which the holder of the Certificate is entitled pursuant to this Article II.
- (f) Lost, Stolen or Destroyed Certificates. In the event any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such Certificate to be lost, stolen or destroyed and, if required by the Payments Administrator, the posting by such Person of a bond in customary amount and upon such terms as may be required by the Payments Administrator as indemnity against any claim that may be made against it or the Subsequent Surviving Company with respect to such Certificate, the Payments Administrator shall issue to such Person the aggregate Per Share Merger Consideration with respect to the number of shares of Company Capital Stock represented by such lost, stolen or destroyed Certificate.
- (g) <u>Company Transaction Expenses Paid by Parent</u>. At the Closing, Parent will pay, or cause to be paid, the Company Transaction Expenses Paid by Parent on behalf of the Company in accordance with the Funds Flow Memorandum.

2.6 Post-Closing Statement.

(a) <u>Delivery of the Post-Closing Statement</u>. No later than ninety (90) days after the Closing Date, Parent shall deliver to the Stockholder Representative the Post-Closing Statement, together with supporting documentation used by Parent in calculating the amounts set forth therein.

Access to Information. From and after delivery of the Post-Closing Statement until the determination of the Final Closing Statement, Parent shall, upon reasonable prior notice by the Stockholder Representative to Parent, (A) provide the Stockholder Representative and its Representatives (including accounting and legal advisors) with reasonable access during normal business hours to the facilities, books and records and work papers of the Subsequent Surviving Company or such copies thereof as may be reasonably requested by the Stockholder Representative (subject to Stockholder Representative and each of Stockholder Representative's advisors being granted access entering into a non-disclosure agreement with the Subsequent Surviving Company in the form attached hereto as Exhibit G); provided that, following the determination of the Final Closing Statement and upon Parent's written request, all such copies must be returned to the Subsequent Surviving Company or destroyed within fifteen (15) Business Days of such written request (and, if destroyed, with such destruction certified in writing to Parent and the Subsequent Surviving Company by the Stockholder Representative or the Stockholder Representative's advisors, as applicable) and (B) cooperate with and assist the Stockholder Representative and its Representatives (including accounting and legal advisors) in connection with the review of such materials, including by making available its employees, accountants and other personnel to the extent reasonably requested.

(c) Dispute Resolution.

- Notice of Objection. If the Stockholder Representative has any objections to the Post-Closing Statement or any of the amounts included in the calculation of the Merger Cash Consideration set forth therein, it shall deliver to Parent a written statement (a "Notice of Objection") setting forth in reasonable detail the particulars of such disagreement (including the specific items in the Post-Closing Statement that are in dispute and the nature and amount of any disagreement so identified) not later than thirty (30) days after its receipt of the Post-Closing Statement (such thirty (30) day period, the "Review Period"). If the Stockholder Representative fails to deliver a Notice of Objection within the Review Period, the Post-Closing Statement and the amounts set forth therein shall be deemed to have been accepted by the Stockholder Representative and shall be deemed final and binding upon all of the Parties, and shall be deemed the Final Closing Statement. If the Stockholder Representative delivers a Notice of Objection to Parent within the Review Period, (i) any items set forth in the Post-Closing Statement that are not objected to by the Stockholder Representative in such Notice of Objection shall be deemed to have been accepted by the Stockholder Representative and shall be deemed final and binding upon all of the Parties and (ii) the Stockholder Representative and Parent shall work in good faith to resolve the Stockholder Representative's objections set forth in such Notice of Objection within the thirty (30) day period following Parent's receipt thereof.
- (ii) Selection of the Accountant. In the event that Parent and the Stockholder Representative are unable to resolve in writing all of the Stockholder Representative's objections in the Notice of Objection within the thirty (30) day period (or such other period as may be agreed by Parent and the Stockholder Representative) following Parent's receipt of such Notice of Objection, the resolution of all such unresolved items ("Disputed Items") shall be submitted to a nationally recognized firm of independent certified public accountants mutually selected by Parent and the Stockholder Representative to resolve any remaining disagreements and determine the Final Merger Cash Consideration. If a mutually selected independent accountant is not appointed within thirty-five (35) Business Days from the date on which a Notice of Objection is delivered (or such longer period as may be agreed by Parent and the Stockholder Representative), Parent and Stockholder Representative shall draw lots to determine the independent accountant; provided that it shall be the New York main office of one of the "big four" accounting firms (such firm as is ultimately selected pursuant to the aforementioned procedures being the "Accountant"). The Stockholder Representative and Parent shall execute any agreement reasonably required by the Accountant for its engagement hereunder.

- Representative shall, promptly (but in any event within thirty five (35) days) following the formal engagement of the Accountant, provide the Accountant (copying the other upon submission) with a single written submission setting forth its respective calculations of and assertions regarding the Disputed Items (which submissions the Accountant shall promptly distribute to the other Party) and upon receipt thereof, each of the Stockholder Representative and Parent shall be entitled (no later than twenty (20) days following receipt of the other Party's initial submission) to submit to the Accountant a single written response to such other Party's initial submission setting forth such Party's objections or rebuttals to the calculations and/or assertions set forth in such initial submission (which responses the Accountant shall promptly distribute to the other applicable Party). There shall be no ex parte communications between the Stockholder Representative (or its Representatives) or Parent (or its Representatives), on the one hand, and the Accountant, on the other hand, relating to the Disputed Items and unless requested by the Accountant in writing, no Party may present any additional information or arguments to the Accountant, either orally or in writing.
- Accountant's Determination. The Accountant shall be instructed to render to both Parent and the Stockholder Representative in writing its determination with respect to (A) the Disputed Items and (B) the allocation of the Accountant's fees pursuant to Section 2.6(c)(v) as soon as reasonably practicable (which the Parties agree shall not be later than thirty (30) days following the date the Accountant receives each Party's comments on the other's initial submission or, if no comments are received, the later date of its receipt of each Party's initial submission). The Accountant shall act as an expert and not as an arbitrator to determine solely the Disputed Items based solely on the submissions and responses of Parent, on the one hand, and the Stockholder Representative, on the other hand, without independent investigation and in accordance with this Agreement. In resolving any disputed amount in connection with its determination of the Final Merger Cash Consideration, or any component thereof, the Accountant may not assign a value to any Disputed Item greater than the greatest value for such Disputed Item claimed by either Party or less than the smallest value for such item claimed by either Party. The Parties agree that the determination of the Accountant with respect to any Disputed Items is not intended to permit the introduction of different accounting methods, policies, principles, practices, procedures, classifications or estimation methodologies for the purposes of determining the Final Merger Cash Consideration, and the Accountant shall not conduct an independent investigation but shall instead base its determination on the written submissions of the Parties delivered pursuant to and in accordance with Section 2.6(c)(iii) with respect to the Disputed Items. The determination of the Accountant in accordance with this Section 2.6(c)(iv) shall be binding and final for purposes of this Agreement. The Post-Closing Statement resulting from the determinations with respect to the Disputed Items made by the Accountant in accordance with this Section 2.6(c)(iv) shall be deemed the Final Closing Statement.
- (v) Accountant's Fees and Expenses. The Accountant shall allocate its costs and expenses between Parent and the Stockholder Representative based upon the percentage of the aggregate contested amount submitted to the Accountant that is ultimately awarded to Parent, on the one hand, or the Company Securityholders, on the other hand, such that Parent bears a percentage of such costs and expenses equal to the percentage of the contested amount awarded to the Company Securityholders and the Stockholder Representative bears a percentage of such costs and expenses equal to the percentage of the contested amount awarded to Parent (with such amount to be deducted from the Stockholder Representative Expense Amount, to the extent funds are available therein, and to the extent the full amount of the funds are not available therein, then from (i) first, the Purchase Price Adjustment Holdback, (ii) second, the Reserve Holdback and (iii) third, the Retention Holdback). The Accountant's allocation shall be final and binding on the Parties. For the avoidance of doubt, the fees and disbursements of the Representatives of each Party incurred in connection with the preparation or

review of the Post-Closing Statement and any Notice of Objection as well as any submissions and responses to the Accountant, as applicable, shall be borne by such Party.

2.7 <u>Post-Closing Adjustment Payments</u>.

- (a) Within five (5) Business Days following the determination of the Final Closing Statement:
 - (i) if the Post-Closing Adjustment is a negative number, then Parent and the Stockholder Representative shall provide joint written instructions to the Escrow Agent and Payments Administrator to release, and the Escrow Agent and Payments Administrator shall release, (A) an amount in cash equal to the absolute value of the Post-Closing Adjustment from the Purchase Price Adjustment Holdback to Parent or its designee, to the extent funds are available therein, and to the extent the full amount of the funds are not available therein, then from (1) first, the Reserve Holdback and (2) second, the Retention Holdback; and (B) the remainder (if any) of the Purchase Price Adjustment Holdback to the Company Securityholders as the Per Share Post-Closing Adjustment Amount in accordance with their Fully Diluted Pro Rata Shares;
 - (ii) if the Post-Closing Adjustment is a positive number, then (A) Parent shall deposit, or cause to be deposited, with the Payments Administrator for further distribution to the Company Securityholders an amount in cash equal to the Post-Closing Adjustment and (B) Parent and the Stockholder Representative shall provide joint written instructions to the Escrow Agent and Payments Administrator to release, and the Escrow Agent and Payments Administrator shall release, an amount in cash equal to the sum of (I) the Post-Closing Adjustment and (II) the Purchase Price Adjustment Holdback to the Company Securityholders as the Per Share Post-Closing Adjustment Amount in accordance with their Fully Diluted Pro Rata Shares; and
 - (iii) if the Post-Closing Adjustment is zero, then Parent and the Stockholder Representative shall provide joint written instructions to the Escrow Agent and Payments Administrator to release, and the Escrow Agent and Payments Administrator shall release, the Purchase Price Adjustment Holdback to the Company Securityholders as the Per Share Post-Closing Adjustment Amount in accordance with their Fully Diluted Pro Rata Shares.
- (b) <u>Tax Matters</u>. Any payment made pursuant to Section 2.7(a) shall, for Tax purposes, be deemed to be an adjustment to the consideration payable to the Company Securityholders. Such amounts shall be paid in immediately available funds, pursuant to the instructions previously delivered by Parent and the Stockholder Representative.
- (c) <u>Exclusive Remedy</u>. Notwithstanding anything to the contrary in this Agreement, the process set forth in this Section 2.7 shall be the sole and exclusive remedy of the Parties for any disputes related to items required to be included or reflected in the calculation of the Final Aggregate Cash Consideration, except in the case of actual fraud.
- 2.8 <u>Withholding Rights</u>. Each of Parent and the Subsequent Surviving Company (each, a "Payor") shall be entitled to deduct and withhold from any amounts otherwise payable pursuant to this Agreement such amounts as it is required to deduct and withhold with respect to the making of such payment under the Code, or any other provision of applicable state, local or foreign Tax law. Each Payor shall reasonably cooperate with the Company Securityholders in order to eliminate or to reduce any such deduction or withholding, including providing a reasonable opportunity for the Company Securityholders to provide forms or other evidence that would mitigate, reduce or eliminate such deduction or withholding. To the extent that amounts are so withheld by an applicable Payor and paid to the

appropriate Governmental Entity in accordance with applicable Tax Law, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding was made.

2.9 <u>No Interest</u>. Notwithstanding anything to the contrary contained herein, no interest shall accumulate on any Per Share Merger Consideration.

2.10 Treatment of Equity Awards.

- (a) <u>Treatment of Options.</u> At the Initial Effective Time, each outstanding option to purchase shares of Common Stock (a "<u>Company Option</u>") under the JackPocket, Inc. 2013 Equity Incentive Plan (as amended, the "<u>Stock Plan</u>"), whether vested or unvested (other than Company Performance Options that are Unvested Company Options, which shall be cancelled in accordance with Section 2.10(b)), shall, by virtue of the Initial Merger and without any further action on the part of the holders thereof or the Parties hereto, shall be treated as follows:
 - In-the-Money Options: Each Company Option that is outstanding and unexercised immediately prior to the Initial Effective Time that has a per share exercise price that is less than the Per Share Closing Consideration (each, an "In-the-Money Option") shall cease to represent an option to purchase shares of Common Stock and shall be converted into an option to purchase a number of shares of Parent Common Stock equal to the product (rounded down to the nearest whole number) of (x) the number of shares of Common Stock subject to such In-the-Money Option immediately prior to the Initial Effective Time and (y) the Equity Award Conversion Amount, at an exercise price per share (rounded up to the nearest whole cent) equal to (A) the exercise price per share of Common Stock of such In-the-Money Option immediately prior to the Initial Effective Time divided by (B) the Equity Award Conversion Amount; provided, however, that the exercise price and the number of shares of Parent Common Stock purchasable pursuant to the In-the-Money Options shall be determined in a manner consistent with the requirements of Section 409A of the Code; provided, further, that in the case of any In-the-Money Option to which Section 422 of the Code applies, the exercise price and the number of shares of Parent Common Stock purchasable pursuant to such option shall be determined in accordance with the foregoing, subject to such adjustments as are necessary in order to satisfy the requirements of Section 424(a) of the Code; provided, further, that except as specifically provided above, following the Initial Effective Time, each In-the-Money Option (other than Company Performance Options that are Unvested Company Options, which shall be cancelled in accordance with Section 2.10(b)) shall continue to be governed by the same terms and conditions (including vesting and exercisability terms) as were applicable to such In-the-Money Option immediately prior to the Initial Effective Time.
 - (ii) <u>Underwater Options</u>: Each Company Option that is outstanding and unexercised immediately prior to the Initial Effective Time that has a per share exercise price that is equal to or greater than the Per Share Closing Consideration (each, an "<u>Underwater Option</u>") shall no longer be outstanding and shall be cancelled at the Initial Effective Time for no consideration or payment, without further action by the holder of such Underwater Option, and no payment shall be made in respect of any Underwater Option following the Initial Effective Time.
- (b) <u>Treatment of Unvested Company Performance Options</u>. Each Unvested Company Option that is a Company Performance Option shall no longer be outstanding and shall be cancelled at the Initial Effective Time for no consideration or payment, without further action by the holder of such Company Performance Option, and no payment shall be made in respect of any Company Performance Option following the Initial Effective Time.

- (c) <u>Company Actions</u>. At or prior to the Initial Effective Time, the Company, the Company Board and the compensation committee of the Company Board, as applicable, shall adopt any resolutions and take any actions that are necessary or appropriate to effectuate the provisions of this Section 2.10. The Company shall take all actions necessary to ensure that, from and after the Initial Effective Time, neither Parent nor the Initial Surviving Company will be required to deliver shares of Common Stock or other capital stock of the Company to any person in connection with Company Options.
- (d) <u>Parent Actions</u>. Parent shall take all corporate actions that are necessary for the treatment of Company Options pursuant to this Section 2.10, including the reservation, issuance and listing of Parent Common Stock, in accordance with the transactions contemplated by this Section 2.10. If registration of the shares of Parent Common Stock issuable under the options to purchase shares of Parent Common Stock contemplated by this Section 2.10 is required under the Securities Act, Parent shall file with the SEC on, or as promptly as practicable after, the Closing Date a registration statement on Form S-8 with respect to such Parent Common Stock.
- (e) Option Acknowledgment Agreement. As a condition to payment of any amount or the issuance of any securities pursuant to this Section 2.10, each holder of Company Options (other than holders of Company Options who also hold Company Capital Stock, who shall execute a Support Agreement) shall be required, and the Company shall use its commercially reasonable efforts to cause such Person, to execute and deliver an Option Acknowledgment Agreement. The Company shall deliver an Option Acknowledgment Agreement to each holder of Company Options (other than holders of Company Options who also hold Company Capital Stock, who shall execute a Support Agreement) for execution as promptly as reasonably practicable following the DK Deal Announcement and shall provide Parent with a copy of each Option Acknowledgment Agreement promptly following its execution.
- 2.11 <u>Treatment of Company Warrants</u>. Each Company Warrant that is outstanding and not exercised immediately prior to the Initial Effective Time shall no longer be outstanding and shall be cancelled as of the Initial Effective Time and not receive or have the right to receive any consideration.
- 2.12 Equitable Adjustments. If at any time during the period between the date hereof and the Initial Effective Time, any change in the issued and outstanding shares of Parent Common Stock or securities convertible or exchangeable into or exercisable for shares of Parent Common Stock or the issued and outstanding shares of Company Capital Stock or securities convertible or exchangeable into or exercisable for shares of Company Capital Stock shall occur by reason of any reclassification, recapitalization, stock split or combination, exchange or readjustment of shares, or any stock dividend thereon with a record date during such period, the Per Share Merger Consideration or any number or amount contained in this Agreement which is based on the price of Parent Common Stock or the number of shares of Parent Common Stock shall be equitably adjusted to the extent necessary to provide the same economic effect to the Company Stockholders, holders of Company Options and Parent as contemplated by this Agreement prior to such reclassification, recapitalization, stock split or combination, exchange or readjustment of shares, or stock dividend thereon; provided, however, that nothing in this Section 2.12 shall be construed to permit the Parties or any other Person to take any action prohibited by the terms of this Agreement.

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company (which shall include for purposes of this Article III, unless the context requires otherwise, each Subsidiary of the Company) hereby represents and warrants to Parent, DK HoldCo,

Merger Sub I and Merger Sub II as of the date hereof and as of the Closing (or in the case of representations and warranties that speak of a specified date, as of such specified date), except as disclosed in the corresponding sections or subsections of the Company Disclosure Letter, as follows:

Organization, Good Standing and Qualification. Each of the Company and its Subsidiaries (a) is a legal entity duly organized, validly existing and in good standing under the Laws of its respective jurisdiction of organization, (b) has all requisite corporate or similar power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted and (c) is qualified to do business and, to the extent such concept is applicable, is in good standing as a foreign corporation or other legal entity in each jurisdiction where the ownership, leasing or operation of its assets or properties or conduct of its business requires such qualification, except in the case of clause (b) or (c) where the failure to be so qualified or in good standing, or to have such power or authority, would not reasonably be expected to (i) be material to the Company and its Subsidiaries (taken as a whole) or (ii) prevent, materially delay or materially impair the consummation of the Transactions. The Company has made available to Parent complete and correct copies of the Company's Certificate of Incorporation (the "Certificate of Incorporation") and the Company's and its Subsidiaries' certificates of incorporation and bylaws or comparable governing documents, each as amended through the date hereof (collectively, "Organizational Documents"), and each as made available to Parent is in full force and effect. Since the date of its incorporation, the Company has been and currently is in compliance and current with all filings, including the filing of the Certificate of Incorporation and fees required to be paid under the DGCL, except as would not reasonably be expected to be material to the Company and its Subsidiaries, taken as a whole. The Company is not in violation of any of the provisions of its Organizational Documents. None of the Subsidiaries of the Company is in violation of any of the provisions of its Organizational Documents in any material respect. Section 3.1 of the Company Disclosure Letter contains a correct and complete list of each jurisdiction in which the Company and its Subsidiaries are organized and qualified to do business.

3.2 Capital Structure.

As of 5:00 P.M. Eastern Time on February 9, 2024 (the "Capitalization Time"), the authorized capital stock of the Company consists of 293,551,440 shares of Company Capital Stock, of which (i) 167,000,000 shares of Common Stock were authorized and 15,060,151 shares of Common Stock were issued and outstanding, (ii) 10,940,623 shares of Seed Stock were authorized and 10,940,623 shares of Seed Stock were issued and outstanding, (iii) 13,690,611 shares of Class A Stock were authorized and 13,690,611 shares of Class A Stock were issued and outstanding, (iv) 24,050,634 shares of Class B Stock were authorized and 24,050,634 shares of Class B Stock were issued and outstanding, (v) 5,962,810 shares of Class B-1 Stock were authorized and 5,962,810 shares of Class B-1 Stock were issued and outstanding, (vi) 27,850,682 shares of Class C-1 Stock were authorized and 27,850,682 shares of Class C-1 Stock were issued and outstanding, (vii) 3,046,173 shares of Class C-2 Stock were authorized and 3,046,173 shares of Class C-2 Stock were issued and outstanding, (viii) 4,951,057 shares of Class C-3 Stock were authorized and 4,951,057 shares of Class C-3 Stock were issued and outstanding, (ix) 25,843,198 shares of Class D-1 Stock were authorized and 25,843,198 shares of Class D-1 Stock were issued and outstanding, (x) 6,080,739 shares of Class D-2 Stock were authorized and 6,080,739 shares of Class D-2 Stock were issued and outstanding, (xi) 4,134,913 shares of Class D-3 Stock were authorized and 1,231,318 shares of Class D-3 Stock were issued and outstanding and (xii) 23,225,504 shares of Common Stock were reserved for issuance upon the exercise of outstanding Company Options, including 15,572,180 shares of Common Stock reserved for issuance upon the exercise of outstanding Company Performance Options. As of the date of this Agreement, no shares of Company Capital Stock are held by the Company in treasury. From the Capitalization Time until the time at which this Agreement is executed, no Company Capital Stock or securities convertible into or exercisable for Company Capital Stock (including for the avoidance of doubt Company Options,

Company PSUs or Company RSUs) have been repurchased or redeemed or issued. All of the outstanding shares of Company Capital Stock have been duly authorized and are validly issued, fully paid and nonassessable. Other than as a result of the exercise of any Company Options in the Ordinary Course and the 2,010,423 shares of Common Stock reserved for issuance pursuant to future grants under the Stock Plan, the Company has no shares of Company Capital Stock reserved for issuance. Other than the Company Options, no equity-based awards (including any cash awards where the amount of payment is determined, in whole or in part, based on the value of any capital stock of the Company or any Subsidiary) are outstanding. Section 3.2(a) of the Company Disclosure Letter sets forth a complete and accurate list of each of the Company Stockholders and the number and type of shares of Company Capital Stock held by each Company Stockholder as of the date of this Agreement.

- (b) Each of the outstanding shares of capital stock or other securities of each of the Company's Subsidiaries is duly authorized, validly issued, fully paid and nonassessable and owned by the Company or by a direct or indirect wholly owned Subsidiary of the Company, free and clear of any Liens (other than any transfer restrictions imposed by federal and state securities Laws or the Company's Organizational Documents). All issued and outstanding shares of Company Capital Stock have been issued in compliance with the Company's Organizational Documents and applicable Law.
- (c) The Company does not have outstanding any bonds, debentures, notes or other obligations, the holders of which have the right to vote (or convert into or exercise for securities having the right to vote) with the Company Stockholders on any matter.
- (d) The Company Capital Stock is not subject to any voting trust agreements, proxies, or other Contracts with respect to the voting, purchase, repurchase, dividend rights, disposition or transfer of Company Capital Stock. As of the Initial Effective Time, all agreements, including stockholders and voting agreements, between or among the Company and the Company Stockholders shall have been terminated and shall be of no further force or effect.
- (e) Other than the Company Options that are outstanding as of the date hereof, there are no preemptive or outstanding rights, options, warrants, agreements, arrangements or commitments under which the Company or any of its Subsidiaries is or may become obligated to sell, or giving any Person a right to acquire or in any way dispose of, any shares of Company Capital Stock or other securities of the Company or any of its Subsidiaries or any securities or obligations exercisable or exchangeable for, or convertible into, or valued by reference to, any securities of the Company or any of its Subsidiaries, and no securities or obligations evidencing such rights are authorized, issued or outstanding.
- (f) Section 3.2(f) of the Company Disclosure Letter sets forth, other than de minimis inaccuracies, a true, complete and correct document containing the information described in clauses (a) and (b) of the Funds Flow Memorandum, with the calculations set forth therein assuming the date hereof as the Closing Date for illustrative purposes.
- (g) All of the information with respect to the Company Stockholders and the allocation of the Aggregate Purchase Price set forth in the Funds Flow Memorandum, when delivered in accordance with Section 2.3(a), will, other than de minimis inaccuracies, be accurate and complete, will set forth a true, correct and complete calculation of the entitlement of each Company Stockholder and each holder of a Company Option under the Company's Organizational Documents, and will comply with all of the requirements set forth in the Company's Organizational Documents, Contracts to which the Company or any of its Subsidiaries is party and the DGCL. Except as set forth in the Funds Flow Memorandum, no Person is entitled to receive any portion of the Aggregate Purchase Price.

- (h) Section 3.2(h) of the Company Disclosure Letter sets forth (i) each of the Company's Subsidiaries and the ownership interest of the Company (or a Subsidiary thereof) in each such Subsidiary and (ii) the Company's or its Subsidiaries' capital stock, equity interest or other direct or indirect ownership interest in any other Person.
- (i) Section 3.2(i) of the Company Disclosure Letter sets forth, for each Company Option, a true, correct and complete list of the holder, type of award, grant date, number of shares, vesting schedule (including any acceleration provisions), exercise price and expiration date, in each case as of the date hereof.
- (j) Each Company Option (i) was granted in compliance with all applicable Laws and the terms and conditions of the Stock Plan, (ii) has an exercise price per share equal to or greater than the fair market value of a share of Common Stock on the date of such grant and (iii) does not trigger any liability for the holder thereof under Section 409A of the Code.

3.3 Authority; Approval.

- (a) The Company has all requisite corporate power and authority and has taken all corporate action necessary in order to execute, deliver and perform its obligations under this Agreement and to consummate the Transactions, subject only to adoption of this Agreement by the holders of (i) a majority of the voting power of the issued and outstanding shares of Company Capital Stock and (ii) a majority of the Company Preferred Stock (excluding the Class D-3 Stock) in each case entitled to approve such matter by written consent or at a stockholders' meeting duly called and held for such purpose (the "Required Stockholder Approval"). Other than the Required Stockholder Approval, there are no votes, approvals, consents or other proceedings of the equityholders or members or any governing body of the Company or any of its Subsidiaries necessary in connection with the execution and delivery of, or the performance by the Company of its obligations under, this Agreement or the consummation of the Transactions. The holders of Company Capital Stock are entitled to act by written consent with respect to duly obtaining the Required Stockholder Approval in accordance with the DGCL and the Company's Organizational Documents. This Agreement has been duly executed and delivered by the Company and constitutes a valid and binding agreement of the Company enforceable against the Company in accordance with its terms, subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar Laws of general applicability relating to or affecting creditors' rights and to general equity principles (the "Bankruptcy and Equity Exception").
- (b) The Company Board has unanimously (i) approved and declared advisable this Agreement and the Transactions, (ii) determined that this Agreement and the Transactions are fair to, and in the best interests of, the Company and the Company Stockholders, (iii) directed that this Agreement be submitted to the Company Stockholders for their adoption and (iv) resolved to recommend that the Company Stockholders adopt this Agreement (the "Board Recommendation").

3.4 Governmental Filings; No Violations; Certain Contracts, Etc.

(a) Other than the expirations of waiting periods and the filings, notices, reports, consents, registrations, approvals, permits and authorizations (i) contemplated by Section 1.4, (ii) required under the HSR Act and any applicable requirements of other Antitrust Laws, and (iii) listed on Section 3.4(a) of the Company Disclosure Letter, no expirations of waiting periods under applicable Laws are required and no notices, reports or other filings are required to be made by the Company with, nor are any consents, registrations, approvals, permits or authorizations required to be obtained by the Company from, any Governmental Entity in connection with the execution, delivery and performance of this Agreement by the Company or the consummation of the Transactions, or in connection with the

continuing operation of the business of the Company following the date of this Agreement, except those that the failure to make or obtain would not reasonably be expected to (x) be material to the Company and its Subsidiaries, taken as a whole or (y) prevent, materially delay or materially impair the consummation of the Transactions.

(b) The execution, delivery and performance of this Agreement by the Company do not, and the consummation of the Transactions will not, conflict with, or result in any breach or violation of or default (with or without notice, lapse of time, or both) under, or give rise to any right of termination, loss of rights, adverse modification of provisions, cancellation or acceleration of any obligation under, or result in the creation of a Lien on any of the assets of the Company or any of its Subsidiaries under, any provision of (i) the Organizational Documents of the Company or any of its Subsidiaries, (ii) any Material Contract or (iii) assuming (solely with respect to performance of this Agreement and consummation of the Transactions) compliance with the matters referred to in Section 3.4(a), any Laws to which the Company or any of its Subsidiaries is subject, except, in the case of clauses (ii) and (iii) above, for any breach, violation, default, termination, loss, adverse modification, cancellation, acceleration or creation that would not, reasonably be expected to be material to the Company and its Subsidiaries, taken as a whole. Section 3.4(b) of the Company Disclosure Letter sets forth a correct and complete list of Material Contracts pursuant to which consents, waivers or notices are or may be required to be obtained or provided, as applicable, prior to consummation of the Transactions (whether or not subject to the exception set forth with respect to clause (ii) above).

3.5 Financial Statements; Filings.

- (a) Set forth in Section 3.5(a) the Company Disclosure Letter are correct and complete copies of the following financial statements (collectively, the "<u>Financial Statements</u>"): (i) the audited balance sheets of the Company and its consolidated Subsidiaries as of December 31 for each of the fiscal years ended December 31, 2022, 2021 and 2020, and the related audited consolidated statements of income (loss), consolidated statements of stockholders' equity and consolidated statements of cash flows for the twelve (12) month periods then ended and (ii) the unaudited balance sheet of the Company and its consolidated Subsidiaries as of December 31, 2023 (the "<u>Most Recent Balance Sheet</u>") and the related unaudited consolidated statements of income (loss), stockholders' equity and cash flows for the nine-month period then ended.
- (b) The Financial Statements (including the related notes and schedules thereto) fairly present in all material respects the consolidated financial position of the Company and its consolidated Subsidiaries as of the date or period set forth therein and the consolidated balance sheets, consolidated statements of income (loss), consolidated statements of stockholders' equity (deficit) and consolidated statements of cash flows included in the Financial Statements (including any related notes and schedules thereto) fairly present in all material respects the financial condition, results of operations, changes in stockholders' equity and cash flows of the Company and its consolidated Subsidiaries for the periods set forth therein, in each case in accordance with GAAP, consistently applied during the periods involved.
- (c) The Company maintains a system of internal accounting controls sufficient to provide reasonable assurances that (i) transactions are executed in accordance with management's general or specific authorizations; (ii) transactions are recorded as necessary to permit preparation of financial statements in accordance with GAAP and to maintain asset accountability; (iii) access to assets is permitted only in accordance with management's general or specific authorization; and (iv) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences.

- (d) The Company has disclosed, based on its most recent evaluation of its internal accounting controls by its chief executive officer and its chief financial officer, to the Company Board (i) any significant deficiencies or material weaknesses in the design or operation of internal controls that would be reasonably expected to adversely affect the Company's ability to record, process, summarize and report financial information for inclusion in the applicable consolidated financial statements and (ii) any actual fraud, whether or not material, that involves any Employees who have (or had) a significant role in the Company's internal controls over financial reporting. No material written, or, to the Company's Knowledge, oral complaints from any source regarding accounting, internal accounting controls or auditing matters have been received by the Company. The Company has made available to Parent a summary of all substantive complaints or concerns relating to other matters made through the Company's whistleblower hot-line or equivalent system for receipt of Employee concerns regarding possible violations of accounting practices or Law.
- (e) Since January 1, 2021 (the "Applicable Date"), the Company has timely filed all reports, statements and other documents, together with any amendments required to be made with respect thereto, that it was required to file under any applicable Law with any applicable Governmental Entity, and has paid all fees and assessments due and payable in connection therewith, except where the failure to file such report, statement or other document or to pay such fees or assessments would not reasonably be expected to be material to the Company and its Subsidiaries, taken as a whole. As of their respective dates, each of such reports, statements and other documents, including the financial statements, exhibits and schedules thereto, complied in all material respects with all Laws of the applicable Governmental Entity with which they were filed.
- (f) None of the Company or any of its Subsidiaries has any outstanding Indebtedness for borrowed money.
 - 3.6 Absence of Certain Changes. Since January 1, 2023 through the date of this Agreement:
- (a) the Company and its Subsidiaries have conducted their respective businesses only in the Ordinary Course;
- (b) there has not been any material damage, destruction or other casualty loss with respect to any material asset or property owned, leased or otherwise used by the Company or any of its Subsidiaries, whether or not covered by insurance;
- (c) neither the Company nor any of its Subsidiaries has taken, or agreed, committed, arranged, authorized or entered into any understanding to take, any action that, if taken on or after the date of this Agreement, would (without Parent's prior written consent) have constituted a breach of Section 5.1(b); and
- (d) there has not been any change, development, event, circumstance, condition, fact, occurrence or effect that, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect with respect to the Company and its Subsidiaries.
- 3.7 <u>No Undisclosed Liabilities.</u> The Company does not have any obligations or liabilities (whether accrued, absolute, contingent, unknown or otherwise) and there is no existing condition, situation or set of circumstances which would reasonably be expected to result in any claims against, or obligations or liabilities of, the Company or any of its Subsidiaries, other than liabilities that (a) have been specifically and adequately reserved against or reflected in the Financial Statements or (b) were incurred since the date of the Most Recent Balance Sheet in the Ordinary Course (it being understood that any liability arising as a result of any breach of Contract, breach of warranty, tortious conduct, litigation,

infringement, misappropriation or violation of applicable Law will be deemed to not have incurred in the Ordinary Course).

3.8 Legal Proceedings.

- (a) There are no Proceedings pending or, to the Knowledge of the Company, threatened in writing, against the Company or any of its Subsidiaries that would reasonably be expected to (i) be material to the Company and its Subsidiaries, taken as a whole or (ii) prevent, materially delay or materially impair the consummation of the Transactions.
- (b) Neither the Company nor any of its Subsidiaries is a party to or subject to the provisions of any Order (i) that restricts the manner in which the Company and its Subsidiaries conduct their respective businesses in any material respect or (ii) which would reasonably be expected to prevent, materially delay or materially impair the consummation of the Transactions.
- (c) To the Knowledge of the Company, there is no Proceeding or inquiry of any nature pending or threatened in writing against any current or former director of the Company (in their capacities as such or relating to their employment, services or relationship with the Company).

3.9 Employee Benefits.

- (a) Section 3.9(a) of the Company Disclosure Letter sets forth a correct and complete list of each material Benefit Plan.
- (b) With respect to each Benefit Plan, the Company has made available to Parent, to the extent applicable, correct and complete copies of (i) the Benefit Plan document, including any amendments thereto, and all related trust documents, insurance Contracts or other funding vehicle documents, (ii) a written description of such Benefit Plan if such plan is not set forth in a written document, (iii) the most recently prepared actuarial report and (iv) all material correspondence to or from any Governmental Entity received in the last three years with respect to any Benefit Plan.
- (c) (i) Each Benefit Plan (including any related trusts) has been established, operated and administered in material compliance with its terms and applicable Laws, including ERISA and the Code, (ii) all contributions or other amounts payable by the Company or any of its Subsidiaries with respect to each Benefit Plan in respect of current or prior plan years have been paid or accrued in accordance with GAAP and (iii) there are no pending or, to the Knowledge of the Company, claims (other than routine claims for benefits) or Proceedings threatened by a Governmental Entity by, on behalf of or against any Benefit Plan or any trust related thereto.
- (d) With respect to each ERISA Plan, the Company has made available to Parent, to the extent applicable, correct and complete copies of (i) the most recent summary plan description together with any summaries of all material modifications thereto, (ii) the most recent IRS determination or opinion letter and (iii) the two most recent annual reports (Form 5500 or 990 series and all schedules and financial statements attached thereto).
- (e) Each ERISA Plan that is intended to be qualified under Section 401(a) of the Code has been determined by the IRS to be qualified under Section 401(a) of the Code and, to the Knowledge of the Company, nothing has occurred that would adversely affect the qualification or tax exemption of any such Benefit Plan. With respect to any ERISA Plan, neither the Company nor any of its Subsidiaries has engaged in a transaction in connection with which the Company or any of its

Subsidiaries reasonably could be subject to either a civil penalty assessed pursuant to Section 409 or 502(i) of ERISA or a tax imposed pursuant to Section 4975 or 4976 of the Code.

- (f) Neither the Company nor any ERISA Affiliate has contributed (or had any obligation of any sort) in the last six years to a plan that is subject to Section 412 of the Code or Section 302 or Title IV of ERISA.
- (g) Neither the Company nor any ERISA Affiliate has maintained, established, participated in or contributed to, or is or has been obligated to contribute to, or has otherwise incurred any obligation or liability (including any contingent liability) under, any "multiemployer plan" within the meaning of Section 3(37) of ERISA in the last six years. No Benefit Plan is a "multiple employer welfare arrangement" (as defined in Section 3(40) of ERISA).
- (h) Except to the extent required by applicable Law, no Benefit Plan provides retiree or post-employment medical, disability, life insurance or other welfare benefits to any Person, and none of the Company or any of its Subsidiaries has any obligation to provide such benefits. To the extent that the Company or any of its Subsidiaries sponsors such plans, the Company or the applicable Subsidiary has reserved the right to amend, terminate or modify at any time each Benefit Plan that provides retiree or post-employment disability, life insurance or other welfare benefits to any Person.
- (i) Each Benefit Plan that is a "nonqualified deferred compensation plan" (within the meaning of Section 409A of the Code) is in documentary compliance with, and has been operated and administered in all material respects in compliance with, Section 409A of the Code and the guidance issued by the IRS provided thereunder. Neither the Company nor any of its Subsidiaries has any obligation to provide, and no Benefit Plan or other agreement provides any individual with the right to, a gross up, indemnification, reimbursement or other payment for any excise or additional Taxes, interest or penalties incurred pursuant to Section 409A or Section 4999 of the Code or due to the failure of any payment to be deductible under Section 280G of the Code.
- (j) Neither the execution and delivery of this Agreement nor the consummation of the Transactions could, either alone or in combination with another event, (i) entitle any Employee to severance pay or any material increase in severance pay, (ii) accelerate the time of payment or vesting, or increase the amount of compensation due to any such Employee, (iii) directly or indirectly cause the Company to transfer or set aside any assets to fund any material benefits under any Benefit Plan, (iv) otherwise give rise to any material liability under any Benefit Plan or (v) limit or restrict the right to merge, materially amend, terminate or transfer the assets of any Benefit Plan on or following the Closing.
- (k) Neither the execution and delivery of this Agreement nor the consummation of the Transactions could, either alone or in combination with another event, result in the payment of any amount that could, individually or in combination with any other such payment, constitute an "excess parachute payment" as defined in Section 280G(b)(1) of the Code.
- (I) No Benefit Plan is maintained outside the jurisdiction of the United States or covers any Employees who reside or work outside of the United States.

3.10 Labor Matters.

(a) Neither the Company nor any of its Subsidiaries is a party to or bound by any collective bargaining agreement or other agreement with a labor union or like organization, and to the Knowledge of the Company, there are no activities or proceedings by any individual or group of

individuals, including representatives of any labor organizations or labor unions, to organize any employees of the Company or any of its Subsidiaries.

- (b) There is no, and has not been any, strike, lockout, work stoppage, unfair labor practice or other material labor dispute, or material arbitration or grievance pending or, to the Knowledge of the Company, threatened. Each of the Company and its Subsidiaries is in compliance in all material respects with all applicable Laws respecting labor, employment and employment practices, terms and conditions of employment, wages and hours (including classification of employees and contractors, discrimination, harassment and equitable pay practices), and occupational safety and health. Neither the Company nor any of its Subsidiaries has incurred any liability or obligation under the Worker Adjustment and Retraining Notification Act and the regulations promulgated thereunder or any similar state or local Law that remains unsatisfied.
- (c) As of the date of this Agreement, none of the Company or any of its Subsidiaries has agreed to any settlement or entered into any settlement agreement related to allegations of sexual or racial harassment or sexual or race-based misconduct by, and to the Knowledge of the Company no allegations of sexual or racial harassment or sexual or race-based misconduct have been made against, any individual in his or her capacity as an Employee.
- (d) Section 3.10(d) of the Company Disclosure Letter sets forth a true and complete list as of the date of this Agreement of all Employees and contains the following information for each such Employee: (i) name or employee identification number; (ii) whether classified as exempt or non-exempt for wage and hour purposes, (iii) whether paid on a salary, hourly or a commission basis, (iv) current annual base salary, base hourly rate of pay or commission rate, as applicable, (v) full or part time status, (vi) current target annual incentive opportunity and actual bonus(es) paid within the past twelve (12) months, (vii) status (i.e., active or inactive and if inactive, type of leave and estimated duration), (viii) the dollar amount of accrued, unused paid time off or vacation balance, (ix) any transaction bonuses and other amounts to be paid to such Employee in connection with the Transactions, (x) corporate hire date and, if different, the most recent hire date and (xi) work location.

3.11 Compliance with Laws; Licenses.

- (a) The Company, its Subsidiaries and their respective officers, directors and Employees, and to the Knowledge of the Company, each of its and its Subsidiaries' other Representatives are, and since the Applicable Date have been, in compliance in all respects material to the Company and its Subsidiaries with all applicable Laws. Since the Applicable Date, neither the Company nor any of its Subsidiaries has received any written or, to the Knowledge of the Company, oral allegation any written notice from any Governmental Entity regarding any failure to comply with any Law, except as would not reasonably be expected to be material to the Company and its Subsidiaries, taken as a whole.
- (b) The Company and each of its Subsidiaries has obtained, holds and is in compliance with all Licenses and registrations necessary to conduct their respective businesses as presently conducted except as would not reasonably be expected to be material to the Company and its Subsidiaries, taken as a whole. Except as would not be reasonably expected to be material to the Company and its Subsidiaries, taken as a whole, all Licenses are valid and in full force and effect and are not subject to any Proceeding that would reasonably be likely to result in any modification, termination or revocation thereof.
- (c) Section 3.11(c) of the Company Disclosure Letter sets forth a correct and complete list of each License and identifies each Company Gaming License. The Company has made available to Parent prior to the date hereof true, complete and accurate copies of all Licenses.

- (d) Section 3.11(d) of the Company Disclosure Letter sets forth a correct and complete list of all of the jurisdictions in which the Company or any of its Subsidiaries operates as a professional fundraiser for any charitable organization and identifies each such professional fundraising registration or certificate. All of the Company's and its Subsidiaries' certificates or registrations for professional fundraising are in good standing. The Company, its Subsidiaries, and their respective officers, directors, managers, members, Employees and agents are, and since the Applicable Date have been, in compliance in all material respects with all Laws, rules, and guidance applicable to fundraising for charitable organizations. Since the Applicable Date, no professional fundraising certificate or registration of the Company nor any of its Subsidiaries has been suspended, revoked, withdrawn, or ceased to be effective.
- (e) Neither the Company nor any of its Subsidiaries operates as a charitable organization in any state.
- (f) No material Licenses will be varied, suspended, revoked or cease to be effective as a result of the Transactions. There is no fact or circumstance that is reasonably expected to cause any License to be revoked, suspended, subject to a variation during its current term, or not to be renewed on terms which are no less advantageous to the Company or its relevant Subsidiary than the current terms of such License, and each action necessary to be undertaken by the Company and its Subsidiaries to enable the renewal or extension of each License that is subject to expiration has been duly taken.
- (g) Since the Applicable Date, neither the Company nor any of its Subsidiaries has (i) made any application for a License that has not been issued, granted or given or (ii) withdrawn any such application.
- (h) Since the Applicable Date, no License of the Company nor any of its Subsidiaries has been suspended, revoked, withdrawn, or ceased to be effective.
- (i) The Company is not in possession of (i) any legal opinion or legal advice in respect of any jurisdiction in which it currently operates or in any future jurisdiction in which it may operate in the United States, that questions the validity of, calls into question, or otherwise does not support the line of business of the Company or any of its Subsidiaries under the current Laws of such jurisdiction or (ii) any correspondence from any Governmental Entity in any jurisdiction in which the Company or any of its Subsidiaries operates, or in any other jurisdiction in the United States, that challenges, questions or otherwise provides that the proposed line of business of the Company or any of its Subsidiaries would not comply with the Laws in such jurisdiction. There are no Proceedings pending or, to the Knowledge of the Company, threatened against the Company or any of its Subsidiaries that challenge, question or otherwise allege that the proposed line of business of the Company or any of its Subsidiaries would not comply with the Laws in any jurisdiction in which the Company or any of its Subsidiaries operates or in any other jurisdiction in the United States.
- (j) To the Knowledge of the Company, where required to do so under Applicable Gaming Law or by any Gaming Regulatory Authority, all relevant directors, officers, managers, Employees, partners, and contractors of the Company and its Subsidiaries have obtained and hold personal management Licenses (or local equivalent License) and such Licenses are in full force and effect.
- (k) To the Knowledge of the Company, all matters that are required to be notified to the relevant Gaming Regulatory Authority in accordance with the terms and conditions of the applicable Company Gaming License have been so notified.

- (I) Since the Applicable Date, each of the Company and its Subsidiaries has adopted all policies and procedures necessary for compliance with Applicable Gaming Laws and the terms and conditions of the Company Gaming Licenses and has adhered in all material respects with the terms thereof.
- (m) Except as would not reasonably be expected to be material to the Company and its Subsidiaries, taken as a whole, all revocations of, or complaints, allegations or warnings directed at the Company or any of its Subsidiaries, inquiries or investigations regarding any Company Gaming License with respect to the Company or any of its Subsidiaries, in each case, as received or notified to the Company or any of its Subsidiaries in writing by a Gaming Regulatory Authority or any other statutory or Governmental Entity in respect of any such Company Gaming License within the five years prior to the date hereof, including copies of any relevant substantive correspondence, are described in reasonable detail in Section 3.11(m) of the Company Disclosure Letter.
- (n) Since the Applicable Date, there have been no material fines paid to, or other sanctions that have been, or threatened in writing to be, imposed on the Company or any of its Subsidiaries by, a Gaming Regulatory Authority or any other Governmental Entity in relation to the business of the Company or any of its Subsidiaries in relation to any Applicable Gaming Laws or any Company Gaming License.
- (o) To the Knowledge of the Company, the licensure of the Employees listed on Section 3.11(o) of the Company Disclosure Letter by Parent or any of its Subsidiaries under any Parent Gaming License would not be expected to cause any adverse modification to, or termination or revocation of, any Parent Gaming License.
- (p) None of the Company or any of its Subsidiaries has any operations, assets, properties, businesses or activities outside the United States.
- (q) The Company, its Subsidiaries and, to the Knowledge of the Company, their respective Representatives are in compliance with, and within the prior five-year period have complied with, (i) the FCPA and (ii) the Other Anti-Bribery Laws. Within the prior five-year period, none of the Company, any of its Subsidiaries or, to the Knowledge of the Company, any of their respective Representatives have paid, offered or promised to pay, or authorized or ratified the payment, directly or indirectly, of any monies or anything of value to an official or Representative (including anyone elected, nominated or appointed to be a Representative) of, or any Person acting in an official capacity for or on behalf of, any Governmental Entity (including any official or employee of any entity directly or indirectly owned or controlled by any Governmental Entity), any royal or ruling family member or any political party or candidate for public or political office for the purpose of influencing any act or decision of any such Governmental Entity or Person to obtain or retain business, or direct business to any Person or to secure any other improper benefit or advantage in each case in violation of the FCPA or any of the Other Anti-Bribery Laws.
- (r) No Proceeding by or before any Governmental Entity involving the Company or its Subsidiaries or their respective directors, officers or Employees or, to the Knowledge of the Company, its and its Subsidiaries' other Representatives relating to the Anti-Corruption Laws is pending or, to the Knowledge of the Company, threatened, except as would not reasonably be expected to be material to the Company and its Subsidiaries, taken as a whole.
- (s) The Company and its Subsidiaries (i) have instituted policies and procedures designed to ensure compliance with the FCPA and the Other Anti-Bribery Laws and (ii) have maintained such policies and procedures in full force and effect.

- (t) The Company and each of its Subsidiaries are in compliance in all material respects, and within the prior five-year period have been in compliance in all material respects, with relevant sanctions and export control Laws and regulations in jurisdictions in which the Company or any of its Subsidiaries do business or are otherwise subject to jurisdiction, including the United States International Traffic in Arms Regulations, the Export Administration Regulations, and United States sanctions Laws and regulations administered by the United States Department of the Treasury's Office of Foreign Assets Control.
- (u) None of the Company or any of its Subsidiaries have applied for a COVID-19 loan or applied for forgiveness of such loan.
- 3.12 <u>Takeover Statutes</u>. No "fair price," "moratorium," "interested stockholder," "control share acquisition," "business combination" or other similar anti-takeover statute or regulation enacted under state or federal law (each, a "<u>Takeover Statute</u>") or any anti-takeover provision in the Company's Organizational Documents is applicable to the Company, the Company Capital Stock, the Mergers or the other Transactions.

3.13 Material Contracts.

- (a) Section 3.13(a) of the Company Disclosure Letter sets forth a complete and correct list of each of the following Contracts, except for any Benefit Plan, to which the Company or any of its Subsidiaries is a party or by which the Company or any of its Subsidiaries or any of their respective assets is bound as of the date hereof (each, a "Material Contract"):
 - (i) each Lease pursuant to which the Company or any of its Subsidiaries currently leases or subleases real property to or from any Person;
 - (ii) each Contract (or group of related Contracts with respect to a single transaction or series of related transactions) pursuant to which the Company or any of its Subsidiaries currently leases personal property to or from any Person providing for lease payments in excess of \$50,000 per annum;
 - (iii) each Contract between the Company and an Employee or independent contractor providing for payments in excess of \$150,000 per annum;
 - (iv) each Contract (or group of related Contracts with respect to a single transaction or series of related transactions) that involves future payments, other residual liabilities, performance or services or delivery of goods or materials to or by the Company or any of its Subsidiaries of an amount or value reasonably expected to exceed \$200,000 in any future twelve (12) month period;
 - (v) each Contract or group of related Contracts with respect to a single transaction or series of related transactions with a Company Top Supplier or Company Top Gaming Authority;
 - (vi) each Contract concerning the establishment or operation of a partnership, strategic alliance, joint venture or limited liability company agreement or similar arrangement;
 - (vii) each Contract entered into at any time prior to the date hereof pursuant to which the Company or any of its Subsidiaries acquired another operating business;

(viii) each Contract that contains a put, call, right of first refusal, right of first offer or similar right pursuant to which the Company or any of its Subsidiaries could be required to, directly or indirectly, purchase or sell, as applicable, any securities, capital stock or other interests, assets or business of any Person;

- (ix) each Contract containing any standstill or similar agreement pursuant to which the Company or any of its Subsidiaries has agreed not to acquire assets or securities of another Person:
- (x) each Contract that prohibits (A) the payment of dividends or distributions in respect of the capital stock, membership interests, partnership interests or other equity interests of the Company or any of its Subsidiaries, (B) the pledging of the capital stock, membership interests, partnership interests or other equity interests of the Company or any of its Subsidiaries or (C) the incurrence of material indebtedness by the Company or any of its Subsidiaries;
- (xi) each Contract that limits or restricts, directly or indirectly, the freedom or ability of the Company or any of its Subsidiaries (or, after the Closing, Parent or any of its Subsidiaries) to (A) compete in any line of business or with any Person or engage in any line of business within any geographic area, (B) solicit or hire any Person or solicit business from any Person or (C) use, retain or access information of a Person pursuant to the terms of a non-disclosure or confidentiality agreement other than, in the case of this clause (C) those related to commercial transactions entered into in the Ordinary Course that are not material to the Company or any of its Subsidiaries;
- (xii) each Contract that could require the disposition of any material assets or line of business of the Company or its Subsidiaries (or, after the Closing, Parent or its Subsidiaries), in each case, with a fair market value in excess of \$50,000;
- (xiii) each Contract that relates to the direct or indirect acquisition or disposition of any securities, capital stock or other interests, assets or business (whether by merger, sale of stock, sale of assets or otherwise) in each case with a fair market value or purchase price in excess of \$150,000 or pursuant to which the Company or any of its Subsidiaries may be required to pay or may be entitled to receive any earn-out, deferred or other contingent payments in excess of \$50,000;
- (xiv) each Contract obligating the Company or any of its Subsidiaries to purchase or otherwise obtain any product or service exclusively from a single third party or granting any third party the exclusive right to develop, market, sell or distribute the Company's or any of its Subsidiaries' products or services;
- (xv) each Contract containing a "most favored nation" or similar provision in favor of any customer or other counterparty of the Company or any of its Subsidiaries or a limitation on the Company's or any of its Subsidiaries' ability to increase prices;
- (xvi) each Contract creating Indebtedness for borrowed money in excess of \$50,000;
- (xvii) each Contract creating or granting a material Lien (other than a Permitted Lien) on any assets of the Company or any of its Subsidiaries, other than purchase money security interests in connection with the acquisition of equipment in the Ordinary Course;

(xviii) each Contract between the Company or any of its Subsidiaries, on the one hand, and any shareholder, director, officer, consultant, employee or agent of the Company or any Person beneficially owning any of the outstanding shares of Company Capital Stock or any of their respective Affiliates, on the other hand (any such Contract, an "Affiliate Contract");

(xix) each Contract (including settlement arrangements) pursuant to which the Company or any of its Subsidiaries has agreed to settle or otherwise compromise any material pending or threatened action, suit, claim, investigation or other Proceeding;

(xx) each Contract providing for indemnification or any guarantee by the Company or any of its Subsidiaries of any Person or pursuant to which any indemnification or guarantee obligations of the Company or any of its Subsidiaries remain outstanding or otherwise survive as of the date hereof, except for any such Contract entered into in the Ordinary Course that does not contain any uncapped indemnification obligations of the Company or any of its Subsidiaries;

(xxi) each collective bargaining agreement or Contract with any labor union, staff association, works council or other similar agency or representative body certified or otherwise recognized for the purposes of bargaining collectively;

(xxii) each Contract that provides for, or relates to, any insurance, surety or other bond maintained by the Company pursuant to Applicable Gaming Laws;

(xxiii) each Contract pursuant to which (A) any license, covenant not to sue. release, waiver, immunity, option or other right is granted under any material Intellectual Property Rights owned by or purported to be owned by the Company or any of its Subsidiaries, (B) any Person has granted any license, covenant not to sue, release, waiver, immunity, option or other right under any Intellectual Property Rights to the Company or any of its Subsidiaries that is material to their businesses, (C) the Company or any of its Subsidiaries has assigned, transferred or sold, or agreed to assign, transfer or sell, any material Intellectual Property Rights to any Person, or (D) the Company or any of its Subsidiaries is subject to any obligation or covenant with respect to the use, sale, licensing, enforcement, prosecution or other exploitation of any material Intellectual Property Rights, including trademark co-existence or consent Contracts (collectively, the "Company IP Agreements"); provided that Company IP Agreements shall exclude (a) non-exclusive licenses for commercially available, offthe-shelf Software that have been granted on generally available terms; (b) non-exclusive licenses granted by the Company to customers in the Ordinary Course on the Company's standard forms; (c) non-exclusive licenses that are incidental to the commercial transaction for which such license is granted; (d) non-disclosure agreements; and (e) assignments of Intellectual Property Rights executed by Employees and contractors of the Company or its Subsidiaries;

(xxiv) each market access agreement, agreement relating to betting, gaming, gambling, sweepstakes or lottery courier services, gaming revenue share agreement or similar agreement;

(xxv) each Contract involving a commitment by the Company or any of its Subsidiaries to make capital expenditures in excess of \$100,000; and

(xxvi) each other Contract not specifically enumerated above that is material to the Company and its Subsidiaries (taken as a whole).

(b) The Company has previously delivered, or made available, to Parent current and complete copies of each written Material Contract and a written description of each oral Material

Contract. Each Material Contract is valid, binding and enforceable against the Company or its Subsidiaries, as the case may be, and, to the Knowledge of the Company, against each other party thereto, subject to the Bankruptcy and Equity Exception, and is in full force and effect. There is no breach or violation of, or default under, any such Material Contract by the Company or any of its Subsidiaries and, since the Applicable Date, no event has occurred that, with the lapse of time or the giving of notice or both, would constitute a default thereunder by the Company or any of its Subsidiaries or would permit or cause the termination, non-renewal or modification thereof or acceleration or creation of any right or obligation thereunder, in each case, except as would not reasonably be expected to be material to the Company and its Subsidiaries taken as a whole. To the Knowledge of the Company, no counterparty to any Material Contract is in breach, violation or default thereof.

3.14 Real Property.

- (a) Neither the Company nor any of its Subsidiaries owns any real property.
- (b) Section 3.14(b) of the Company Disclosure Letter sets forth a correct and complete list of all real property leased or subleased to the Company or any of its Subsidiaries (collectively, the "Leased Real Property") and a list of all leases, amendments, modifications, guarantees, and any other material documents (the "Leases") entered into by the Company or any of its Subsidiaries with respect to the Leased Real Property. Each of the Leases constitutes a legal, valid and binding obligation of the Company party thereto and, to the Knowledge of the Company, of the other parties thereto, subject to the Bankruptcy and Equity Exception.
- (c) Except in any such case as would not reasonably be expected to be material to the Company and its Subsidiaries taken as a whole, (i) the Company and its Subsidiaries, as applicable, have a valid leasehold interest in all Leased Real Property, free and clear of all Liens, except Permitted Liens, (ii) there exists no breach, violation or default under the Leases or, to the Knowledge of the Company, any circumstances or events, which with notice, lapse of time or both, would constitute a breach, violation or default under the Leases or permit acceleration of rent payments or termination or modification of such Leases by any third party thereunder, (iii) the Company or one of its Subsidiaries is in possession of the Leased Real Property and the Company's and its Subsidiaries' occupancy, use and operation of the Leased Real Property materially complies with all applicable Laws and the terms of the Leases, (iv) there are no pending or, to the Knowledge of the Company, threatened appropriation, condemnation, eminent domain or similar proceedings by any Governmental Entity relating to the Leased Real Property, and (v) there are no written or oral subleases, concessions, licenses, occupancy agreements or other Contracts or arrangement granting to any Person other than the Company or its Subsidiaries the right to use or occupy the Leased Real Property.
- (d) Neither the Company nor any Subsidiary owes any brokerage commissions or finder's fees with respect to any Lease. Neither the Company nor any of its Subsidiaries has subleased, licensed or otherwise granted any Person the right to use or occupy any Leased Real Property or any portion thereof.
- 3.15 Environmental Matters. Except for such matters that would not individually or in the aggregate reasonably be expected to be material to the Company and its Subsidiaries, taken as a whole, since the Applicable Date, (a) the Company and its Subsidiaries have complied at all times with all applicable Environmental Laws, (b) no property currently owned or operated by the Company or any of its Subsidiaries (including soils, groundwater, surface water, buildings or other structures) is contaminated with any Hazardous Substance which would reasonably be expected to require remediation or other action pursuant to any Environmental Law, (c) no property formerly owned or operated by the Company or any of its Subsidiaries was contaminated with any Hazardous Substance during or prior to

such period of ownership or operation which would reasonably be expected to require remediation or other action pursuant to any Environmental Law, (d) neither the Company nor any of its Subsidiaries is subject to liability for any Hazardous Substance disposal or contamination on any third party property, (e) neither the Company nor any of its Subsidiaries has been associated with any release or threat of release of any Hazardous Substance, (f) neither the Company nor any of its Subsidiaries has received any written. or, to the Knowledge of the Company, threatened, notice, demand, letter, claim or request for information alleging that the Company or any of its Subsidiaries may be in violation of or subject to liability under any Environmental Law which has not been fully resolved, (g) neither the Company nor any of its Subsidiaries is subject to any Order or any indemnity or other agreement with any third party relating to liability under any Environmental Law or relating to Hazardous Substances, (h) there are no other circumstances or conditions involving the Company or any of its Subsidiaries that would reasonably be expected to result in any claim, liability, investigation, cost or restriction on the ownership, use or transfer of any property pursuant to any Environmental Law and (i) the Company has delivered to Parent correct and complete copies of all environmental reports, studies, assessments, sampling data and other environmental information relating to the Company or its Subsidiaries or their respective current properties or operations.

3.16 Taxes.

- The Company and each of its Subsidiaries have (i) duly and timely filed all (a) (taking into account any extension of time within which to file) income and other material Tax Returns required to be filed by them (or any affiliated, consolidated, combined, unitary or similar group that includes the Company or any of its Subsidiaries) with the appropriate Tax authority and all such filed Tax Returns are complete and accurate in all material respects; (ii) have paid all income and other material Taxes required to be paid by them (or any affiliated, consolidated, combined, unitary or similar group that includes the Company or any of its Subsidiaries) whether or not shown on any Tax Returns; (iii) have withheld and paid all material Taxes required to have been withheld and paid by them in connection with amounts paid or owing to any Employee, stockholder, creditor, independent contractor or third party (each as determined for Tax purposes); (iv) have complied in all material respects with all information reporting (and related withholding) and record retention requirements relating to Tax or administered by a Tax authority; and (v) except as a result of an automatic extension with respect to the due date for filing a Tax Return, have not waived any statute of limitations with respect to Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency and neither the Company nor any of its Subsidiaries has received any written request by a Tax authority to execute such a waiver or extension.
- (b) The Company has delivered or made available to Parent copies of all income and other material Tax Returns for the Company and its Subsidiaries for all taxable periods since January 1, 2019.
- (c) All material Taxes required to have been collected and paid by the Company or any of its Subsidiaries on the sale of products or taxable services by the Company (whether or not denominated as sales or use Taxes) have been properly collected and paid or have been accrued for on the Company's balance sheet, and all sales tax exemption certificates or other proof of the exempt nature of sales of such products or services have, in all material respects, been properly collected and retained or, to the extent required, submitted to the appropriate Tax authority.
- (d) Neither the Company nor any of its Subsidiaries has received any written notice of any deficiency with respect to a material amount of Taxes proposed, asserted or assessed against the Company or any of its Subsidiaries. Neither the Company nor any of its Subsidiaries has received any written notice of any audits, suits, claims, examinations, investigations, or other Proceedings in respect of a material amount of Taxes of the Company or any of its Subsidiaries.

- (e) No jurisdiction in which the Company or any of its Subsidiaries does not file a Tax Return of a particular type has asserted in writing a claim that the Company or such Subsidiary is subject to Taxes of such type or required to file Tax Returns of such type in such jurisdiction.
- (f) The Company has made available to Parent correct and complete copies of any private letter rulings, requests for rulings, closing agreements, gain recognition agreements, technical advice memoranda or similar agreements or rulings with respect to Taxes of the Company and its Subsidiaries.
- (g) There are no Liens for Taxes (except Taxes not yet due and payable) on any of the assets of the Company or any of its Subsidiaries, nor to the Knowledge of the Company, is any Tax authority in the process of imposing a Lien for Taxes upon such assets (other than a Lien for Taxes not yet due and payable).
- (h) Neither the Company nor any of its Subsidiaries is a party to or is bound by any Tax sharing, allocation or indemnification agreement or arrangement (other than (i) such an agreement or arrangement exclusively between or among the Company and its Subsidiaries or (ii) any customary Tax indemnification provisions contained in commercial Contracts not primarily related to Taxes (such as financing or employment Contracts with Tax gross-up obligations or leases with Tax escalation provisions) ("Ordinary Course Commercial Agreements")).
- (i) Neither the Company nor any of its Subsidiaries (A) has been a member of an affiliated group filing a consolidated federal income Tax Return (other than a group the common parent of which was the Company) or (B) has any liability for the Taxes of any person (other than the Company or any of its Subsidiaries) under Treasury Regulations Section 1.1502-6 (or any similar provision of Law), as a transferee or successor, by Contract (other than pursuant to any Ordinary Course Commercial Agreement) or otherwise by Law.
- (j) Neither the Company nor any of its Subsidiaries has been, within the past two years or otherwise as part of a "plan (or series of related transactions)" within the meaning of Section 355(e) of the Code of which the Mergers are also a part, a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of stock intended to qualify for tax-free treatment under Section 355 of the Code.
- (k) Neither the Company nor any of its Subsidiaries has been, during any applicable period specified in Section 897(c)(1)(A)(ii) of the Code, a "United States Real Property Holding Corporation" within the meaning of Section 897(c)(2) of the Code.
- (I) Neither the Company nor any of its Subsidiaries has participated in a "listed transaction" within the meaning of Treasury Regulations Section 1.6011-4(b)(2).
- (m) The Company is, and has been since formation, properly classified for United States federal income tax purposes as a corporation. Each of the Company's Subsidiaries are, and have been since formation, treated as disregarded entities for U.S. federal income tax purposes.
- (n) No Subsidiary of the Company is characterized as a "foreign" corporation for U.S. federal income tax purposes.
- (o) Neither the Company nor any of its Subsidiaries is engaged in, or has ever been engaged in, a trade or business through a "permanent establishment" within the meaning of an applicable income Tax treaty in any country other than the United States.

- (p) The Company and its Subsidiaries have delivered to Parent and are in full compliance with all material terms and conditions of any Tax deferral, Tax exemption, Tax holiday or Tax reduction agreement or order that was applied for by the Company and its Subsidiaries. None of the Company or any of its Subsidiaries have (i) deferred payment of any payroll Taxes pursuant to Section 2302 of the CARES Act or the Payroll Tax Executive Order that remains unpaid or (ii) claimed any employee retention credit pursuant to Section 2301 of the CARES Act or Section 3134 of the Code.
- (q) Neither the Company nor any of its Subsidiaries will be required to include any material item of income in, or to exclude any material item of deduction from, taxable income in any taxable period (or portion thereof) ending after the Closing Date as a result of any closing agreement, installment sale or open transaction on or prior to the Closing Date, any accounting method change or agreement with any Tax authority made or entered into on or prior to the Closing Date, any prepaid amount received on or prior to the Closing Date outside the Ordinary Course, any intercompany transaction or excess loss account described in Section 1502 of the Code (or any corresponding provision of Tax Law) incurred on or before the Closing Date, or any election pursuant to Section 108(i) of the Code or Section 965(h) of the Code (or any similar provision of Law) made with respect to any taxable period ending on or prior to the Closing Date.
- (r) Neither the Company nor any of its Subsidiaries has taken any action, and to the Knowledge of the Company, there is no fact or circumstance, that would reasonably be expected to prevent or impede the Mergers, taken together, from qualifying for the Intended Tax Treatment. Neither the Company nor any of its Subsidiaries has taken any action, or has any plan or intention to take any action, that would reasonably be expected to cause the Mergers, taken together, to fail to qualify for the Intended Tax Treatment.
- (s) Nothing in this Section 3.16 or otherwise in this Agreement shall be construed as a representation or warranty with respect to the amount or availability of any net operating loss, capital loss, Tax credits, or other Tax asset or attribute of the Company or any of its Subsidiaries in any taxable period (or portion thereof) beginning after the Closing. No representation or warranty contained in this Section 3.16 or otherwise in this Agreement (other than those in Section 3.16(i), (j) and (q)) shall apply or be deemed to apply directly or indirectly with respect to any taxable period (or portion thereof) beginning after the Closing Date.

3.17 Intellectual Property; IT Assets; Data Privacy.

- (a) Section 3.17(a) of the Company Disclosure Letter sets forth a correct and complete list of all Registered Company IP, indicating for each item the registration or application number, the registration or application date, and the applicable filing jurisdiction.
- (b) The Company and its Subsidiaries own or have sufficient and valid written rights to use all Intellectual Property Rights used in or necessary for, the conduct of their respective businesses as currently conducted and as currently planned to be conducted by the Company and its Subsidiaries, all of which rights shall survive the consummation of the Transactions, without modification, cancellation, termination, suspension of, or acceleration of any right, obligation or payment with respect to any such Intellectual Property Rights, except for Contracts disclosed pursuant to Section 3.4(b) or that otherwise would not, individually or in the aggregate, reasonably be expected to be material to the Company and its Subsidiaries, taken as a whole.
- (c) The Company and its Subsidiaries solely and exclusively own all Company IP, free and clear of all Liens, except for Permitted Liens.

- (d) All Registered Company IP is subsisting, valid and enforceable, and no Registered Company IP is subject to any outstanding Order adversely affecting the validity or enforceability of, or the Company's or its Subsidiaries' ownership or use of, or rights in or to, any such Registered Company IP.
- (e) Neither the Company nor any of its Subsidiaries has received any claim, notice, invitation to license or similar communication (i) contesting or challenging the use, validity, enforceability or ownership of any Company IP material to the Company's or any of its Subsidiaries' respective businesses, or (ii) alleging or suggesting that the Company, any of its Subsidiaries, any of their respective products or services, or the conduct of their respective businesses infringes, misappropriates or otherwise violates the Intellectual Property Rights of any Person, whether directly or indirectly.
- (f) The conduct of the respective businesses of the Company and its Subsidiaries does not, and as currently planned to be conducted in a bona fide written operating plan will not, and the development, manufacture, use, sale, commercialization or other exploitation of any product, service or other offering currently provided by the Company or any of its Subsidiaries does not infringe, misappropriate or otherwise violate, and has not infringed, misappropriated or otherwise violated any Intellectual Property Rights of any Person.
- (g) To the Knowledge of the Company, no Person has infringed, misappropriated or otherwise violated any Company IP, whether directly or indirectly. Neither the Company nor any of its Subsidiaries has made or threatened any claim, notice, invitation to license or similar communication alleging that a Person has infringed, misappropriated or otherwise violated any Company IP.
- (h) The Company and its Subsidiaries have taken commercially reasonable measures to protect the confidentiality and value of all Trade Secrets (including source code) that are owned, used or held by the Company and its Subsidiaries, and such Trade Secrets have not been used, disclosed to or otherwise discovered by any Person, except pursuant to bona-fide valid and enforceable written nondisclosure agreements that have not been breached by such Person.
- (i) Each former and current Employee who developed for or on behalf of the Company or any of its Subsidiaries any Intellectual Property Rights material to the businesses of the Company or any of its Subsidiaries has executed a valid and enforceable written agreement containing a present assignment to the Company or its Subsidiaries, as appropriate, of all such Intellectual Property Rights. No such Employee retains any right, title or interest in or to any such Intellectual Property Rights. Section 3.17(i) of the Company Disclosure Letter sets forth a correct and complete list of any independent contractors that have contributed to the development of any Intellectual Property Rights material to the businesses of the Company or any of its Subsidiaries.
- (j) No Software that is owned or purported to be owned by the Company or any of its Subsidiaries is subject to any obligation or condition under any license identified as an open source license by the Open Source Initiative (www.opensource.org/), or any license that conditions the distribution of (or providing access to) such Software on (i) the disclosure, licensing or distribution of any source code for any portion of such Software, (ii) the granting to third parties of the right to make derivative works or other modifications to such Software, (iii) permitting reverse engineering, reverse assembly or disassembly of any portion of such Software (other than as required by Law), or (iv) distribution of such Software at no or a limited license fee. No Person (other than the Company or any of its Subsidiaries) is in possession of any source code for such Software, including any escrow agent.
- (k) The IT Assets are sufficient for the businesses of the Company and its Subsidiaries as currently conducted, except as would not, individually or in the aggregate, reasonably be

expected to be material to the Company and its Subsidiaries, taken as a whole, and there has been no unauthorized access to or unauthorized use of (i) any IT Assets, or (ii) any confidential or proprietary information (including Trade Secrets) that is in the Company's or any of its Subsidiaries' possession or control. The IT Assets operate and perform in accordance with their documentation and functional specifications and otherwise as required by the Company and its Subsidiaries in connection with their businesses as currently conducted, except as would not, individually or in the aggregate, reasonably be expected to be material to the Company and its Subsidiaries, taken as a whole. The Company and its Subsidiaries have implemented reasonable backup and disaster recovery technology consistent with reasonable industry practices. The Company and each of its Subsidiaries have taken reasonable steps and implemented reasonable safeguards, consistent with industry standard practices, designed to protect the IT Assets from unauthorized access and use, and none of the IT Assets contain or make available any material Malicious Code.

- (I) The Company and its Subsidiaries have established and implemented written policies and organizational, physical, administrative and technical measures regarding privacy, cyber security and security of Personal Information that are commercially reasonable, and consistent in all material respects with (i) applicable Laws relating to privacy, cyber security and security of Personal Information, (ii) industry standard practices, (iii) any applicable requirements of any Contracts of the Company or any of its Subsidiaries, and (iv) any publicly facing statements or policies adopted by the Company or any of its Subsidiaries (the foregoing (i) through (iv), to the extent regarding privacy, cyber security and security of Personal Information, collectively, the "Privacy and Security Obligations"). Where, and to the extent required by, applicable Privacy and Security Obligations, the Company and its Subsidiaries obligate (pursuant to written Contracts) all Persons who receive Personal Information from, or on behalf of, the Company or any of its Subsidiaries to comply with applicable Privacy and Security Obligations.
- (m) Since the Applicable Date, the Company and each of its Subsidiaries have (i) complied, in all material respects, with all of their respective applicable Privacy and Security Obligations regarding Personal Information, including with respect to the collection, use, storage, processing, transmission, transfer (including cross-border transfers), disclosure and protection of Personal Information, and (ii) used reasonable measures designed to ensure the confidentiality, privacy and security of Personal Information. No Person has gained unauthorized access to or misused any Personal Information, and neither the Company nor any of its Subsidiaries have notified, or have been obligated to notify, any Person or Governmental Entity with respect to any unauthorized access to, or misuse of, Personal Information.
- 3.18 Personal Property. Except as would not reasonably be expected to be material to the Company and its Subsidiaries (taken as a whole), the Company (a) owns, leases or licenses from third parties all tangible personal property required to conduct its business in the Ordinary Course, (b) has good and valid title to all tangible personal property owned by it, free and clear of all Liens except for Permitted Liens, and (c) immediately upon consummation of the Transactions contemplated by this Agreement, will be entitled to continue to use all material tangible personal property which is currently employed by it in the conduct of its business as presently conducted. Except as would not reasonably be expected to be material to the Company and its Subsidiaries (taken as a whole), such tangible personal property is in reasonable operating condition and reasonably fit for the particular purposes for which it is used (subject only to normal maintenance requirements and reasonable wear and tear). At the Closing, the Company will own, or have a valid and legal right to use, sufficient property, assets and other rights (whether tangible or intangible) to be able to continue to operate and conduct the business and operations of the Company in the Ordinary Course in all material respects and the Company will have good and valid title to such property, assets and other rights, free and clear of any Liens except Permitted Liens.

3.19 <u>Insurance</u>. The Company has made available to Parent true and complete copies of all material fire and casualty, general liability, business interruption, product liability and sprinkler and water damage insurance policies maintained by the Company or any of its Subsidiaries (collectively, the "<u>Insurance Policies</u>"). The Insurance Policies are each in full force and effect, subject to the Bankruptcy and Equity Exception, and all premiums due with respect to all Insurance Policies have been paid. Since the Applicable Date, neither the Company nor any of its Subsidiaries has taken any action or failed to take any action that (including with respect to the Transactions), with or without notice, lapse of time or both, would constitute or result in a breach or violation of, or default under, any of the Insurance Policies or would permit or cause the termination, non-renewal or modification thereof or acceleration or creation of any right or obligation thereunder, in each case other than such actions as would not reasonably be expected to be material to the Company and its Subsidiaries, taken as a whole.

3.20 Suppliers and Gaming Authorities.

- (a) Section 3.20(a) of the Company Disclosure Letter sets forth a complete and accurate list of the top fifteen (15) suppliers or vendors of the Company and its Subsidiaries for the twelve-month period prior to the date hereof based on payments made to each supplier during such period (each, a "Company Top Supplier"), together with the amount paid to each Company Top Supplier during such period.
- (b) Section 3.20(b) of the Company Disclosure Letter sets forth a complete and accurate list of the top fifteen (15) lottery commissions or other similar Gaming Regulatory Authorities for the twelve-month period prior to the date hereof based on revenue of the Company and its Subsidiaries in the jurisdiction of the applicable lottery commission or Gaming Regulatory Authority (each, a "Company Top Gaming Authority").
- (c) Through the date of this Agreement, (i) no Company Top Supplier, Company Top Gaming Authority or other material customer, supplier, Gaming Regulatory Authority, vendor, collaborator, distributor, licensor or contractor of the Company and its Subsidiaries has cancelled or otherwise terminated its relationship with the Company or any of its Subsidiaries or has materially altered, in a manner adverse to the Company or any of its Subsidiaries, its relationship with the Company or any of its Subsidiaries. To the Knowledge of the Company, no Company Top Supplier, Company Top Gaming Authority or other material customer, supplier, Gaming Regulatory Authority, vendor, collaborator, distributor, licensor or contractor of the Company and its Subsidiaries has any plan or intention, and has not threatened in writing, to terminate, cancel or otherwise materially modify its relationship with the Company or any of its Subsidiaries.
- 3.21 Related Party Transactions. There are no liabilities (including Indebtedness for borrowed money) of any Related Party to the Company, or of the Company to any Related Party (other than ordinary course, Employee- and director-related compensation and reimbursement liabilities). No Related Party (i) has any interest in any property (real, personal or mixed, tangible or intangible) used by the Company in the conduct of its business or (ii) has been, since the Applicable Date, a party to any Contract with the Company. In the five (5) years prior to the date hereof, all transactions between the Company and a Related Party have been on an arm's-length basis on terms no less favorable to the Company than would be available from an unaffiliated third party.
- 3.22 <u>Bank Accounts.</u> Section 3.22 of the Company Disclosure Letter sets forth a true and complete list of each bank account in the name of the Company and its Subsidiaries, including the title and number of the account, the individuals with signatory authority over such account and the financial institution at which such account is located.

- 3.23 <u>Brokers and Finders</u>. Neither the Company nor any of its Subsidiaries, or any of their respective directors, managers, members or officers, as applicable, has employed any investment banker, broker or finder or incurred or will incur any liability for any brokerage payments, investment banking fees, commissions, finders' fees or other similar payments in connection with the Transactions, except that the Company has engaged Raine Securities LLC as its financial advisor in connection with the Transactions. The Company has made available to Parent a complete and correct copy of all Contracts pursuant to which Raine Securities LLC is entitled to any fees and expenses in connection with the Transactions.
- 3.24 <u>Private Placement</u>. The Company understands and acknowledges that the offer and sale of the shares of Parent Common Stock comprising the Aggregate Stock Consideration (the "<u>Private Placement Shares</u>") to the Company Stockholders is intended to be exempt from registration, and will not be registered, under the Securities Act or any applicable state securities or "blue sky" Laws and, therefore, the Private Placement Shares will be characterized as "restricted securities" under the Securities Act and such laws and may not be sold unless they are subsequently registered under the Securities Act and qualified under state law or unless an exemption from such registration and qualification is available.
- 3.25 <u>Accredited Investors</u>. After due inquiry and reasonable investigation, including review of all completed and executed Accredited Investor Questionnaires, the Company reasonably believes that each of the Company Stockholders receiving the Accredited Investor Closing Stock Consideration is an "accredited investor" as defined in Rule 501(a) of Regulation D under the Securities Act.
- 3.26 <u>Trade Allowances</u>. Since the Applicable Date, no customer, supplier, Gaming Regulatory Authority, vendor, collaborator, distributor, licensor or contractor of the Company and its Subsidiaries is entitled to or customarily receives material discounts, allowances, rebates, credits, preferential terms or similar reductions in price or other trade terms arising from any Contracts with or concessions granted to any such Person other than in connection with Ordinary Course promotions. Since the Applicable Date, all discounts, allowances, rebates, credits, preferential terms or similar reductions in price or other trade terms, including contra transactions, to which customers, suppliers, Gaming Regulatory Authorities, vendors, collaborators, distributors, licensors or contractors of the Company and its Subsidiaries are entitled or customarily receive have been granted by the Company and its Subsidiaries in the Ordinary Course.
- 3.27 <u>Working Capital</u>. Since the Applicable Date, (a) the Company and its Subsidiaries have managed their working capital (including timing of collection of accounts receivable and of payment of accounts payable and management of inventory) and deferred revenue amounts in the Ordinary Course and in amounts that are consistent with past practice and (b) there has not been any material change in payment terms to any customer of, or from any supplier or third party to, the Company or any of its Subsidiaries.
- 3.28 Products. Since the Applicable Date, each of the products designed, produced, marketed, sold, leased, installed or made available by or on behalf of the Company or any of its Subsidiaries (each, a "Product") is, and at all times up to and including the sale thereof has been, produced in all material respects with the specifications and standards required by or contained in: (a) supplier Contracts, (b) Contracts with Gaming Regulatory Authorities, (c) all applicable requirements of Law; and (d) regulation of Governmental Entities in each jurisdiction where the Products are made available or sold. Except as set forth on Section 3.28 of the Company Disclosure Letter, since the Applicable Date, neither the Company nor any of its Subsidiaries has received any written notice that any Proceeding has been commenced or will be commenced against the Company or any of its Subsidiaries by or before any Governmental Entity alleging that any Product does not comply in all material respects with applicable Performance Standards,

except as would not reasonably be expected to be material to the Company and its Subsidiaries, taken as a whole.

- Information. The Company acknowledges and understands that, as of the date hereof, Parent may possess material information ("Undisclosed Information") regarding Parent and its Subsidiaries not known to the Company or the Company Securityholders that may be material to the value of the shares of Parent Common Stock. Notwithstanding such non-disclosure, the Company has deemed it appropriate to enter into this Agreement and to consummate the Transactions. Except in connection with the representations and warranties expressly set forth in Article IV of this Agreement, included in the representation letters contemplated by Section 5.14(i) or the information set forth in the certificates required to be delivered pursuant to Section 6.3(c), or with respect to claims of actual fraud, the Company agrees that Parent shall not have any liability to the Company or any Company Securityholders whatsoever due to or in connection with non-disclosure of Undisclosed Information in connection with the Transactions, and the Company, on behalf of itself and the Company Securityholders, hereby irrevocably waives any claim that it might have based on the failure of Parent to disclose such Undisclosed Information to the Company or the Company Securityholders.
- 3.30 Non-Reliance. The Company acknowledges and agrees that, except for the representations and warranties expressly set forth in Article IV of this Agreement, included in the representation letters contemplated by Section 5.14(i) or the information set forth in the certificates required to be delivered pursuant to Section 6.3(c), none of Parent, DK HoldCo, Merger Sub I, Merger Sub II nor any other Person has made any express or implied representation or warranty with respect to Parent or any of its Subsidiaries (including DK HoldCo, Merger Sub I and Merger Sub II) or Affiliates, in connection with this Agreement or the Transactions or with respect to the accuracy or completeness of any other information provided, or made available, to the Company or any of its Subsidiaries or Affiliates in connection with the Transactions, and the Company has not relied on any representation or warranty other than those expressly set forth in Article IV of this Agreement. Notwithstanding the foregoing, nothing in this Section 3.30 shall limit the Company's remedies with respect to claims of actual fraud.
- 3.31 No Other Representations or Warranties. Except for the representations and warranties expressly set forth in this Article III, included in the representation letters contemplated by Section 5.14(i) or the information set forth in the certificates required to be delivered pursuant to Section 6.3(c), neither the Company nor any other Person makes any other express or implied representation or warranty with respect to the Company or any of their respective Subsidiaries or Affiliates in connection with this Agreement or the Transactions, and the Company hereby disclaims any such other representations or warranties. Without limiting the generality of the foregoing, except for the representations and warranties expressly set forth in this Article III or the information set forth in the certificates required to be delivered pursuant to Section 6.2(c), neither the Company nor any other Person has made or makes any representation or warranty with respect to the respective businesses, operations, assets, liabilities, conditions (financial or otherwise) or prospects of the Company nor its Subsidiaries or Affiliates. Notwithstanding the foregoing, nothing in this Section 3.31 shall limit the DK Parties' remedies with respect to claims of actual fraud.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF PARENT, DK HOLDCO, MERGER SUB I AND MERGER SUB II

Except as set forth in (i) the Parent Reports prior to the date of this Agreement, but excluding, in each case, any disclosures set forth in any risk factor, forward-looking statement, quantitative and qualitative disclosures about market risk section or in any other section to the extent they are forward-

looking statements or cautionary, predictive or forward-looking in nature, or (ii) the corresponding sections or subsections of the Parent Disclosure Letter, Parent, DK HoldCo, Merger Sub I and Merger Sub II hereby represent and warrant to the Company as of the date hereof and as of the Closing (or in the case of representations and warranties that speak as of a specified date, as of such specified date) as follows:

- 4.1 Organization, Good Standing and Qualification. Each of Parent, DK HoldCo, Merger Sub I and Merger Sub II (a) is a legal entity duly organized, validly existing and in good standing under the Laws of its respective jurisdiction of organization, (b) has all requisite corporate or similar power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted and (c) is qualified to do business and, to the extent such concept is applicable, is in good standing as a foreign corporation or other legal entity in each jurisdiction where the ownership, leasing or operation of its assets or properties or conduct of its business requires such qualification, except in the case of clause (b) or (c) where the failure to be so qualified or in good standing or to have such power or authority, would not reasonably be expected to prevent, materially delay or materially impair the consummation of the Transactions.
- 4.2 Parent Common Stock. Prior to the Initial Effective Time, Parent will have taken all necessary action to permit it to issue the number of shares of Parent Common Stock comprising the Aggregate Stock Consideration at the Initial Effective Time. Such shares of Parent Common Stock, when issued, (a) will be duly authorized, validly issued, fully paid and nonassessable, (b) will be issued in compliance in all material respects with applicable securities laws, including the Securities Act and any applicable state securities or "blue sky" Laws, and other applicable Laws, (c) will not, at the time of issuance in accordance with the terms of this Agreement, violate any pre-emptive rights, rights of first offer, rights of first refusal or similar rights of any Person and (d) will be free of any Liens, other than any transfer restrictions imposed by federal and state securities Laws, its organizational documents or by any agreements, documents and certificates to be executed and delivered pursuant to, or in connection with, this Agreement. Such Parent Common Stock, when issued, and the offering thereof, will not be registered, and will be exempt from registration, under the Securities Act and any applicable state securities or "blue sky" Laws.
- 4.3 <u>Authority; Approval</u>. Each of Parent, DK HoldCo, Merger Sub I and Merger Sub II has all requisite corporate or similar power and authority and has taken all corporate action necessary in order to execute, deliver and perform their obligations under this Agreement and to consummate the Transactions. This Agreement has been duly executed and delivered by Parent, DK HoldCo, Merger Sub I and Merger Sub II and, assuming the due execution by the Company, constitutes a valid and binding agreement of Parent, DK HoldCo, Merger Sub I and Merger Sub II, enforceable against Parent, DK HoldCo, Merger Sub I and Merger Sub II in accordance with its terms, subject to the Bankruptcy and Equity Exception.

4.4 Governmental Filings; No Violations.

(a) Other than the expirations of waiting periods and the filings, notices, reports, consents, registrations, approvals, permits and authorizations (i) contemplated by Section 1.4, (ii) required under the HSR Act and any applicable requirements of other Antitrust Laws, and (iii) listed on Section 4.4(a) of the Parent Disclosure Letter, no expirations of waiting periods under applicable Laws are required and no notices, reports or other filings are required to be made by Parent, DK HoldCo, Merger Sub I or Merger Sub II with, nor are any consents, registrations, approvals, permits or authorizations required to be obtained by Parent, DK HoldCo, Merger Sub I or Merger Sub II from, any Governmental Entity in connection with the execution, delivery and performance of this Agreement by Parent, DK HoldCo, Merger Sub I and Merger Sub II or the consummation of the Transactions, except

those that the failure to make or obtain would not, individually or in the aggregate, reasonably be expected to prevent, materially delay or materially impair the consummation of the Transactions.

(b) The execution, delivery and performance of this Agreement by Parent, DK HoldCo, Merger Sub I and Merger Sub II do not, and the consummation of the Transactions will not, conflict with, or result in any breach or violation of, or default (with or without notice, lapse of time or both) under, or give rise to any right of termination, loss of rights, adverse modification of provisions, cancellation or acceleration of any obligations under, or result in the creation of a Lien on any of the assets of Parent, DK HoldCo, Merger Sub I or Merger Sub II under any provision of (i) the articles of incorporation, bylaws or comparable governing documents of Parent, DK HoldCo, Merger Sub I or Merger Sub II or (iii) assuming (solely with respect to performance of this Agreement and consummation of the Transactions) compliance with the matters referred to in Section 4.4(a), any Law to which Parent, DK HoldCo, Merger Sub I or Merger Sub II is subject, except, in the case of clauses (ii) and (iii) above, for any such breach, violation, default, termination, loss, adverse modification, cancellation, acceleration or creation that would not, individually or in the aggregate, reasonably be expected to prevent, materially delay or materially impair the consummation of the Transactions.

4.5 Public Filings.

- (a) Since January 1, 2023, Parent has timely filed or furnished in all material respects all Parent Reports required to be filed or furnished (as applicable) by it with the SEC pursuant to applicable Law. As of the time filed with the SEC (or, if amended or superseded by a filing, then on the date of such filing), since January 1, 2023, the Parent Reports complied with all applicable Laws in all material respects and did not contain any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading. As of the date hereof, Parent has not received from the SEC any written comments or questions with respect to any of the Parent Reports (including the financial statements included therein) that have not been resolved.
- (b) The consolidated financial statements of the Parent (including all related notes and schedules) included in the Parent Reports (1) complied as of the filing date in all material respects with the requirements of the Securities Act and the Exchange Act, as the case may be, and the applicable rules and regulations promulgated thereunder, (2) were prepared in accordance with GAAP (except, in the case of unaudited statements, as permitted by Form 10-Q of the SEC) applied on a consistent basis during the periods involved (except as may be indicated in the notes thereto), (3) fairly presented in all material respects the consolidated financial position of Parent and its consolidated Subsidiaries as of the dates thereof and the consolidated results of their operations and cash flows for the periods shown in accordance with GAAP (subject, in the case of unaudited statements, to the absence of footnote disclosure and to normal year-end audit adjustments) and (4) have been prepared from, and are in accordance with, the books and records of Parent and its consolidated Subsidiaries in all material respects.
- 4.6 <u>Legal Proceedings</u>. There are no Proceedings pending or, to the Knowledge of Parent, DK HoldCo, Merger Sub I or Merger Sub II, threatened against Parent, DK HoldCo, Merger Sub I or Merger Sub II that would reasonably be expected to prevent, materially delay or materially impair the consummation of the Transactions. None of Parent, DK HoldCo, Merger Sub I or Merger Sub II are a party to or subject to the provisions of any material Order that would, individually or in the aggregate, reasonably be expected to prevent, materially delay or materially impair the consummation of the Transactions.

- 4.7 <u>Available Funds</u>. As of the date of this Agreement, Parent has and will have at the Closing cash on hand or existing credit facilities of immediately available funds sufficient to enable it to fund the portion of the Aggregate Cash Consideration to be paid at the Closing. Parent will have sufficient funds available to satisfy its obligations to make any additional payments contemplated by this Agreement following the Closing when required to do so pursuant to the terms hereof.
- 4.8 Merger Sub I and DK HoldCo. The authorized capital stock of Merger Sub I consists solely of 1,000 shares of common stock, par value \$0.0001, all of which are validly issued. All of the issued shares of common stock of Merger Sub I are, and at the Initial Effective Time will be, owned by DK HoldCo. Merger Sub I was formed solely for the purpose of engaging in the Transactions. Merger Sub I has not conducted any business prior to the date of this Agreement and has no, and immediately prior to the Initial Effective Time will have no, assets, liabilities or obligations of any nature other than those incident to its formation and pursuant to this Agreement and the Initial Merger and the other transactions contemplated by this Agreement. All of the issued shares of capital stock of DK HoldCo are, and at the Initial Effective Time will be, owned by Parent.
- 4.9 Merger Sub II. The authorized membership interests of Merger Sub II consists solely of 1,000 limited liability company units, all of which are validly issued. All of the issued membership interests of Merger Sub II are, and at the Subsequent Effective Time will be, owned by DK HoldCo. Merger Sub II was formed solely for the purpose of engaging in the Transactions. Merger Sub II has not conducted any business prior to the date of this Agreement and has no, and immediately prior to the Subsequent Effective Time will have no, assets, liabilities or obligations of any nature other than those incident to its formation and pursuant to this Agreement and the Subsequent Merger and the other transactions contemplated by this Agreement.
- 4.10 <u>Brokers and Finders</u>. None of Parent, DK HoldCo, Merger Sub I, Merger Sub II nor any of their respective Subsidiaries, nor any of their respective directors or employees (including any officers) has employed any broker, finder or investment bank or has incurred or will incur any obligation or liability for any brokerage fees, commissions or finders fees in connection with the transactions contemplated by this Agreement, except that the DK Parties have engaged Goldman Sachs LLC as its financial advisor in connection with the Transactions, who will be entitled to receipt of a fee upon, and subject to, the consummation of the Transactions.

4.11 Tax Matters.

- (a) Neither Parent, DK HoldCo nor any of their Subsidiaries has taken any action, and to the Knowledge of Parent, there is no fact or circumstance, that would reasonably be expected to prevent or impede the Mergers, taken together, from qualifying for the Intended Tax Treatment. Neither Parent, DK HoldCo nor any of their Subsidiaries has taken any action, or has any plan or intention to take any action, that would reasonably be expected to cause the Mergers, taken together, to fail to qualify for the Intended Tax Treatment. For the avoidance of doubt, for purposes of this Section 4.11(a), in the case of any action or failure to take any action by any of the DK Parties or any of their respective Affiliates or Subsidiaries resulting in a decline in the trading price or value of Parent Common Stock, such decline shall not be, in and of itself, considered as reasonably expected to prevent or impede the Mergers, taken together, from qualifying for the Intended Tax Treatment.
- (b) DK HoldCo, since May 5, 2022 has been, and, through the Subsequent Effective Time will be, properly treated as a direct and wholly-owned corporate subsidiary of Parent for U.S. federal income tax purposes. Parent is and will be, from the date of this Agreement through the Subsequent Effective Time, in control of DK HoldCo within the meaning of Section 368(a)(2)(D) and 368(c) of the Code.

- (c) Merger Sub II is, has since the date of its formation been, and, through the Subsequent Effective Time will be, properly treated as an entity that is disregarded as separate from DK HoldCo for U.S. federal income tax purposes. Merger Sub I is, has since the date of its formation been, and through the Initial Merger will be, properly treated as a direct and wholly-owned corporate subsidiary of DK HoldCo for U.S. federal income tax purposes.
- 4.12 <u>Capitalization</u>. As of the Capitalization Time, the authorized capital stock of Parent consists of 2,100,000,000 shares, of which (a) 900,000,000 shares are designated as Parent Common Stock, of which 473,457,819 shares are issued and outstanding, (b) 900,000,000 shares are designated as Class B Common Stock, par value \$0.0001 per share ("<u>Parent Class B Stock</u>"), of which 393,013,951 shares are issued and outstanding, and (c) 300,000,000 shares are designated as Preferred Stock, par value \$0.0001 per share, of which 0 shares are outstanding. All of the issued and outstanding shares of capital stock of Parent have been duly authorized and validly issued and are fully paid, nonassessable and free of preemptive rights, with no personal liability attaching to the ownership thereof. As of the Capitalization Time, 31,650,367 restricted stock units covering shares of Parent Common Stock and 22,065,714 options to acquire shares of Parent Common Stock were outstanding pursuant to the Parent Equity Plans.
- 4.13 <u>Absence of Certain Changes</u>. From January 1, 2023 through the date of this Agreement, there has not been any change, development, event, circumstance, condition, fact, occurrence or effect that, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect with respect to Parent and its Subsidiaries.
- 4.14 Non-Reliance. Parent, DK HoldCo, Merger Sub I and Merger Sub II each acknowledges and agrees that, except for the representations and warranties expressly set forth in Article III of this Agreement, included in the representation letters contemplated by Section 5.14(i) or the information set forth in the certificates required to be delivered pursuant to Section 6.3(c), neither the Company nor any other Person has made any express or implied representation or warranty with respect to the Company or any of its Subsidiaries or Affiliates, in connection with this Agreement or the Transactions or with respect to the accuracy or completeness of any other information provided, or made available, to Parent, DK HoldCo, Merger Sub I, Merger Sub II or any of its Subsidiaries or Affiliates in connection with the Transactions, and none of Parent, DK HoldCo, Merger Sub I or Merger Sub II have relied on any representation or warranty other than those representations and warranties expressly set forth in Article III of this Agreement or the information set forth in the certificates required to be delivered pursuant to Section 6.3(c). Notwithstanding the foregoing, nothing in this Section 4.14 shall limit Parent's, DK HoldCo's, Merger Sub I's or Merger Sub II's remedies with respect to claims of actual fraud.
- 4.15 No Other Representations or Warranties. Except for the representations and warranties expressly set forth in this Article IV, included in the representation letters contemplated by Section 5.14(i) or the information set forth in the certificates required to be delivered pursuant to Section 6.3(c), none of Parent, DK HoldCo, Merger Sub I, Merger Sub II nor any other Person makes any other express or implied representation or warranty with respect to Parent, DK HoldCo, Merger Sub I, Merger Sub II or any of their respective Subsidiaries or Affiliates in connection with this Agreement or the Transactions, and Parent, DK HoldCo, Merger Sub I and Merger Sub II hereby disclaim any such other representations or warranties. Without limiting the generality of the foregoing, except for the representations and warranties expressly set forth in this Article IV, included in the representation letters contemplated by Section 5.14(i) or the information set forth in the certificates required to be delivered pursuant to Section 6.3(c), none of Parent, DK HoldCo, Merger Sub I, Merger Sub II nor any other Person has made or makes any representation or warranty with respect to the respective businesses, operations, assets, liabilities, conditions (financial or otherwise) or prospects of Parent, DK HoldCo, Merger Sub I, Merger Sub II or their respective Subsidiaries or Affiliates. Notwithstanding the foregoing, nothing in this Section 4.15 shall limit the Company's remedies with respect to claims of actual fraud.

ARTICLE V

COVENANTS

5.1 Interim Operations.

- (a) Except (i) as set forth in Section 5.1(a) of the Company Disclosure Letter, (ii) as expressly required by this Agreement, (iii) as required by applicable Law, or (iv) as otherwise consented to in advance by Parent in writing (which approval shall not be unreasonably withheld, conditioned or delayed), the Company covenants and agrees that, during the period from the date hereof until the earlier of the Closing and the termination of this Agreement pursuant to Article VII, the Company shall, and shall cause each of its Subsidiaries to, conduct its business in the Ordinary Course, including, to the extent consistent therewith, by using commercially reasonable efforts to preserve substantially intact its business organization, goodwill and assets, keep available the services of its current Employees and maintain existing relations and goodwill with Governmental Entities, customers, suppliers, licensors, licensees, vendors, distributors, creditors, lessors and others with which it has substantial business relations.
- (b) Without limiting the generality of and in furtherance of the foregoing, except (w) as set forth in the corresponding subsection of Section 5.1(b) of the Company Disclosure Letter, (x) as expressly required by this Agreement, (y) as required by applicable Law, or (z) as otherwise consented to in advance by Parent in writing (which approval shall not be unreasonably withheld, conditioned or delayed), the Company covenants and agrees that, during the period from the date hereof until the earlier of the Closing and the termination of this Agreement pursuant to Article VII, the Company shall not, and shall cause each of its Subsidiaries not to:
 - (i) amend or propose to amend its Organizational Documents;
 - (ii) merge or consolidate with any other Person or enter into any agreements or arrangements imposing material changes or restrictions on its assets, operations or businesses;
 - (iii) transfer, sell, lease, divest, cancel or otherwise dispose of, or permit or suffer to exist the creation of any Lien upon, any assets (tangible or intangible), product lines or businesses of it or any of its Subsidiaries, including capital stock of any of its Subsidiaries, except in connection with services provided in the Ordinary Course and sales of obsolete assets;
 - (iv) restructure, reorganize, dissolve or completely or partially liquidate or subject the Company or any of its Subsidiaries to any bankruptcy, receivership, insolvency or similar proceeding;
 - (v) issue, sell, pledge, dispose of, grant, transfer, encumber, or authorize the issuance, sale, pledge, disposition, grant, transfer or encumbrance of, any shares of capital stock of the Company or any of its Subsidiaries (other than the issuances of shares in respect of Company Options outstanding as of the date of this Agreement in accordance with their terms, and, as applicable, the Stock Plan as in effect on the date of this Agreement or the conversion of any Company Capital Stock entitled to conversion pursuant to the terms of the Company's Certificate of Incorporation), or securities convertible or exchangeable into or exercisable for, or valued by reference to, any shares of such capital stock, or any options, warrants or other rights of any kind to acquire any shares of such capital stock;

- (vi) reclassify, split, combine, subdivide or redeem, purchase or otherwise acquire, directly or indirectly, any of its capital stock or securities convertible or exchangeable into or exercisable for any shares of its capital stock (other than the withholding of shares to satisfy withholding Tax obligations in respect of Company Options outstanding as of the date of this Agreement in accordance with their terms and, as applicable, the Stock Plan as in effect on the date of this Agreement);
- (vii) declare, set aside, make or pay any dividend or other distribution in respect of the capital stock of the Company or repurchase, redeem or otherwise acquire any outstanding shares of Company Capital Stock or other securities of, or other ownership interests in, the Company;
- (viii) grant, extend, waive or modify any material rights in or to, or acquire, exchange, sell, assign, lease, license, transfer, let lapse, abandon or otherwise dispose of, or incur any Liens or permit any Liens to be imposed on (other than Permitted Liens), any Leased Real Property, any Company IP or other material properties, of the Company or any of its Subsidiaries, other than grants of non-exclusive licenses under Company IP in the Ordinary Course;
- (ix) make any loans, advances, guarantees or capital contributions to or investments in any Person (other than to or from the Company and any of its wholly owned Subsidiaries), in excess of \$100,000;
- (x) incur any Indebtedness for borrowed money or guarantee any such Indebtedness of another Person in excess of \$100,000, or issue or sell any debt securities or warrants or other rights to acquire any debt security of the Company or any of its Subsidiaries;
- (xi) enter into any Contract that would have been a Material Contract had it been entered into prior to the date of this Agreement or amend, modify, fail to renew, supplement, waive, terminate (partially or completely), assign, convey, encumber or otherwise transfer, in whole or in part, rights or interest pursuant to or in any Material Contract, except for terminations by the counterparty or expirations of Contracts in accordance with their terms or as required pursuant to the terms of any Benefit Plan in effect as of the date of this Agreement;
- (xii) amend, modify, cancel, or waive any debts or claims held by it or waive any material rights;
- (xiii) make any changes with respect to its accounting policies or procedures, except to the extent required by changes in Law or GAAP;
- (xiv) (A) commence any Proceeding or (B) settle any Proceeding for an amount in excess of \$100,000, net of any amount covered by insurance, indemnification or existing reserves established in accordance with GAAP, or settle any other obligation or liability of the Company or any of its Subsidiaries (A) in excess of such amount or (B) on a basis that would result in the imposition of any Order that would restrict the future activity or conduct of the Company or any of its Subsidiaries or a finding or admission of a violation of Law or violation of the rights of any Person, except, in each case, for (1) Proceedings for the routine collection of bills or (2) in such cases which it in good faith determines that the failure to commence a Proceeding would result in the material impairment of a valuable aspect of its business, so long as the Company notifies Parent in writing before the commencement of such Proceeding and keeps Parent reasonably informed of all material developments in connection therewith;

(xv) (A) make, change or rescind any material Tax election; (B) settle or compromise any claim, audit, assessment or dispute relating to any income Tax or material amount of other Tax; (C) file any material amended Tax Return; (D) make any change in any material method of Tax accounting or Tax accounting periods; (E) take any action to affirmatively surrender any claim for a refund of a material amount of Taxes; (F) enter into any closing agreement with a Tax authority relating to any material amount of Tax; or (G) consent to any extension or waiver of the limitation period applicable to any material Tax claim or assessment (other than as a result of an automatic extension with respect to the due date for filing a Tax Return);

(xvi) except as required pursuant to the terms of any Benefit Plan in effect as of the date of this Agreement, (A) increase in any manner the compensation or consulting fees, bonus, pension, welfare, fringe or other benefits, severance or termination pay of any Employee; provided, that the Company may make Ordinary Course increases to its Employees' base salaries in connection with the Company's Ordinary Course review cycle or promotions in connection with filling vacancies (with a corresponding increase to any compensation or benefits calculated by reference to base salary) where such base salary increases do not exceed (i) in the aggregate five percent (5%) of the aggregate base salaries for all Employees in respect of fiscal year 2023 and (ii) twelve percent (12%) of any single Employee's base salary for the prior year (provided, that if such Employee is being promoted to a position at the level of "director" and below in connection with such increase, such increase shall not exceed twenty percent (20%) of such Employee's base salary for the prior year), (B) except with respect to any Benefit Plan in the process of being established as of the date of this Agreement, as disclosed in Section 3.9(a) of the Company Disclosure Letter, become a party to, establish, adopt, amend, commence participation in or terminate any Benefit Plan or any arrangement that would have been a Benefit Plan had it been entered into prior to this Agreement, (C) grant any new awards, or amend or modify the terms of any outstanding awards, under any Benefit Plan, (D) take any action to accelerate the vesting or lapsing of restrictions or payment, or fund or in any other way secure the payment, of compensation or benefits under any Benefit Plan, (E) hire any Employee with an annualized salary or wage or fee in excess of \$200,000; provided, however, that (1) the aggregate annualized salaries, wages and fees of all Employees newly hired following the date hereof in accordance with this Section 5.1(b)(xvi)(E) shall not exceed \$500,000 (the "Incremental Hire Salary Cap"); provided that the Incremental Hire Salary Cap shall increase to \$1 million on the six-month anniversary of the date hereof and \$1.5 million on the twelve-month anniversary of the date hereof; and (2) the Company may make replacement hires irrespective of the Incremental Hire Salary Cap so long as the amount of any increase to the annualized salary (excluding bonus opportunities) of any replacement hire does not exceed five percent (5%) of the annual salary (excluding bonus opportunities) of the former Employee, (F) terminate the employment of any Employee, other than for cause or a legitimate business reason, with an annualized salary or wage or fee in excess of \$200,000, (G) change any actuarial or other assumptions used to calculate funding obligations with respect to any Benefit Plan that is required by applicable Law to be funded or change the manner in which contributions to such plans are made or the basis on which such contributions are determined, except as may be required by GAAP or (H) forgive any loans or issue any loans (other than routine travel advances issued in the Ordinary Course) to any Employee;

(xvii) disclose to any third party any Trade Secrets included in the Company IP in a way that results in the loss of protection for such Trade Secrets;

(xviii) (A) fail to use commercially reasonable efforts to keep in full force and effect any material Licenses, (B) modify, cancel, terminate, rescind any material License in any material respect or (C) fail to notify Parent in writing prior to applying for any material License;

(xix) cancel or terminate any material Insurance Policy or cause any of the coverage thereunder to lapse (unless simultaneously with such termination, cancellation or lapse, replacement policies providing coverage equal to or greater than the coverage under the canceled, terminated or lapsed policies, for premiums at not more than current market rates, are in full force and effect);

(xx) except to the extent expressly provided by, and consistent with the line items set forth in, the Company's capital budget set forth in Section 5.1(b)(xx) of the Company Disclosure Letter, make or authorize any payment of, or accrual or commitment for, capital expenditures in excess of \$50,000 in the aggregate or, to the extent that such payments of, or accruals or commitments for capital expenditures are in connection with operations in any states in the United States in which the Company does not operate as of the date of this Agreement, in excess of \$75,000 in the aggregate;

(xxi) enter into any line of business in (A) any geographic area outside of the United States or (B) any new geographic area within the United States, other than in the case of this clause (B), an existing line of business in a state in the United States; provided that, in the case of this clause (B), (I) the Company provides written notice to Parent prior to conducting business in any such new state, (II) entry into such new state would not, and would not reasonably be expected to, (1) restrict, restrain or otherwise prohibit the operation of the business of the Company and its Subsidiaries or (2) jeopardize any License of the Company or any Subsidiary thereof (including its Company Gaming Licenses), and (III) neither the Company nor any Subsidiary thereof incurs costs, authorizes payments or makes commitments for payment or spending in such new state in an amount in excess of \$75,000 in the aggregate;

(xxii) incur costs, authorize any payments or make any commitments for payment or spending with respect to lobbying activities in an amount in excess of \$200,000 in any calendar month:

(xxiii) acquire assets from any other Person with a fair market value or purchase price in excess of \$100,000 individually or \$200,000 in the aggregate in any transaction or series of related transactions, in each case, including any amounts or value reasonably expected to be paid in connection with a future earn-out, purchase price adjustment, release of "holdback" or similar contingent payment obligation, or that would reasonably be expected to prevent, materially delay or materially impair the ability of the Company or Parent, as applicable, to consummate the Transactions prior to the Outside Date;

(xxiv) become a party to, establish, adopt, amend, commence, participate in or terminate any collective bargaining agreement or other agreement with a labor union, works council or similar organization;

(xxv) fail to manage its working capital (including the timing of collection of accounts receivable and of the payment of accounts payable and the management of inventory) and deferred revenue amounts in the Ordinary Course;

(xxvi) change in any material respect the payment or payment terms to any Company Top Supplier outside the Ordinary Course; or

(xxvii) agree, authorize or commit to do any of the foregoing.

- (c) Nothing contained in this Agreement is intended to give Parent, directly or indirectly, the right to control or direct any of the Company's operations prior to the Closing. Prior to the Closing, the Company shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over the operations of the Company and its Subsidiaries.
- (d) Except (w) as set forth in the corresponding subsection of Section 5.1(b) of the Parent Disclosure Letter, (x) as expressly required by this Agreement, (y) as required by applicable Law, or (z) as otherwise consented to in advance by the Company in writing (which approval shall not be unreasonably withheld, conditioned or delayed), Parent covenants and agrees that, during the period from the date hereof until the earlier of the Closing and the termination of this Agreement pursuant to Article VII, Parent shall not, and shall cause each of its Subsidiaries not to:
 - (i) amend or propose to amend the Organizational Documents of Parent in a manner that would reasonably be expected to materially and adversely affect the Company Stockholders relative to the stockholders of Parent; or
 - (ii) agree, authorize or commit to do any of the foregoing.

5.2 Cooperation; Status Updates.

- (a) <u>Cooperation and Efforts</u>. Upon the terms and subject to the conditions set forth in this Agreement, each of Parent, DK HoldCo, Merger Sub I, Merger Sub II and the Company shall cooperate with each other and use (and shall cause their respective Subsidiaries and ultimate parent entities as defined under 16 C.F.R. §801.1(a)(3) ("<u>UPEs</u>") to use) their respective commercially reasonable efforts to take or cause to be taken all actions reasonably necessary, proper or advisable on their part under this Agreement and applicable Law to consummate the Transactions as promptly as reasonably practicable.
- Subject to applicable Laws and as required by any Status Updates. Governmental Entity, Parent and the Company shall each keep the other apprised of the status of matters relating to the consummation of the Transactions, including promptly furnishing the other with copies of material or substantive notices or other communications (or where no such copies are available, a reasonably detailed written description thereof) received by the other Party or any of its Subsidiaries, from any third party and/or any Governmental Entity with respect to the Transactions. Parent and the Company shall give prompt notice to one another of (i) any notice or other communication received by Parent or Company, as applicable, from any Governmental Entity or other third party alleging that the consent of such Person is or may be required in connection with the Transactions, (ii) any matter (including a breach of any representation, warranty, covenant or agreement contained in this Agreement) that would reasonably be expected to lead to the failure to satisfy any of the conditions to Closing in Section 6.1, Section 6.2, or Section 6.3, (iii) any changes, developments, events, circumstances, conditions, facts, occurrences or effects that, individually or in the aggregate, has had or is reasonably expected to have a Material Adverse Effect, or (iv) any change, occurrence, fact or condition with respect to which Parent or the Company, as applicable, may terminate this Agreement pursuant to Article VII.

5.3 Regulatory Filings and Approvals.

(a) <u>Submission of Filings and Notices</u>. Without limiting the generality of anything contained in this Section 5.3, each of Parent, DK HoldCo, Merger Sub I, Merger Sub II and the Company shall (and shall cause their respective Subsidiaries and UPEs to) use its commercially reasonable efforts to cooperate in all respects and consult with each other in connection with any filing or submission in connection with any investigation or other inquiry of any Governmental Entity, including the following:

- (i) <u>Exchanging Information</u>. Parent, DK HoldCo, Merger Sub I, Merger Sub II and the Company shall (and shall cause their respective Subsidiaries and UPEs to), upon request by the other, furnish the other with all information concerning itself, its Subsidiaries, UPEs, directors, managers, officers, members and stockholders and such other matters as may be reasonably necessary or advisable in connection with any statement, filing, notice or application made by or on behalf of Parent, the Company or any of their respective Subsidiaries or UPEs to any Governmental Entity in connection with the Transactions.
- (ii) <u>Initial Submissions</u>. Each of Parent, DK HoldCo, Merger Sub I, Merger Sub II and the Company shall (and shall cause their respective Subsidiaries and UPEs to) prepare and file as promptly as reasonably practicable all documentation to effect all necessary notices, reports and other filings and to obtain as promptly as practicable all consents, clearances, registrations approvals, permits and authorizations necessary or advisable to be obtained from any Governmental Entity in order to consummate the Transactions. Without limiting the foregoing, each of Parent and the Company shall (and shall cause their respective Subsidiaries and UPEs to) make their respective filings of the notification and report form required under the HSR Act no later than ten (10) Business Days after the date of this Agreement.
- (iii) <u>Subsequent Submissions</u>. Each of Parent, DK HoldCo, Merger Sub I, Merger Sub II and the Company shall (and shall cause their respective Subsidiaries and UPEs to) promptly provide all documents requested by any Governmental Entity to the extent reasonably necessary or advisable to obtain as promptly as practicable all consents, registrations, approvals, permits and authorizations necessary or advisable to be obtained from such Governmental Entity in order to consummate the Transactions, including to any requests for additional information or documentary material made by a Governmental Entity pursuant to 15 U.S.C. §18a(e) or any information or documentary material reasonably required to obtain the Requisite Regulatory Approvals.
- Conduct of Interactions with Government Entities. Subject to (iv) applicable Laws relating to the exchange of information, the Parties shall cooperate on all matters with any Governmental Entity in respect of any filing, investigation, inquiry or litigation relating to the Transactions or the other matters contemplated by this Section 5.3, including Section 5.3(c); provided that, in the event of disagreement between the Parties, Parent shall have final decision-making authority with respect to proceedings under the HSR Act or any other Antitrust Law; provided, further, that the Company or its outside legal counsel shall have the right to review in advance and, to the extent practicable, Parent will consider in good faith the views of the Company in connection with, all the information relating to the Company and any of its Subsidiaries that appears in any filing made with, or written materials submitted to, any Governmental Entity in connection with the Transactions. Subject to the preceding sentence and to the extent not prohibited by applicable Law, Parent, on the one hand, and the Company, on the other hand, shall, in connection with this Agreement and the Transactions, with respect to actions taken on or after the date hereof: (i) promptly notify the other Party of, and if in writing, furnish the other Party with copies of (or, in the case of oral communications, advise the other Party of) any substantive communications from or with any Governmental Entity; (ii) consult with and permit the other Party to review and discuss in advance, and consider in good faith the view of the other Party in connection with, any proposed substantive written or oral communication with any Governmental Entity; (iii) not participate in any substantive meeting or have any substantive communication with any Governmental Entity unless it has given the other Party a reasonable opportunity to consult with it in advance and, to the extent not prohibited by such Governmental Entity, gives the other Party the opportunity to attend and participate therein; (iv) promptly furnish the other Party's outside legal counsel with copies of all filings and communications between it and any such Governmental Entity with respect to this Agreement; and (v) promptly furnish

the other Party's outside legal counsel with such necessary information and reasonable assistance as the other Party's outside legal counsel may reasonably request in connection with its preparation of necessary submissions of information to any such Governmental Entity. Subject to applicable Law and not limiting the first sentence of this Section 5.3(a)(iv), Parent and the Company will consult and cooperate with one another in connection with any analyses, appearances, presentations, memoranda, briefs, arguments, opinions and proposals made or submitted on behalf of any Party hereto relating to proceedings under the HSR Act or any other Antitrust Law. None of Parent, DK HoldCo, Merger Sub I, Merger Sub II and the Company shall (and each of them shall cause their respective Subsidiaries and UPEs not to) (i) commit to or agree with any Governmental Entity to stay, toll, or extend any applicable waiting period or to not consummate the Transactions for any period of time or (ii) pull and refile, in each case, without the prior written consent of the other (which consent shall not be unreasonably withheld, conditioned or delayed); provided that Parent, DK HoldCo, Merger Sub I, Merger Sub II or any of its Affiliates or UPEs shall be entitled to pull and refile on one occasion within the first sixty (60) days following the date of this Agreement without the prior written consent of the Company.

- (v) Redactions. Materials required to be provided pursuant to this Section 5.3 may be provided on an outside legal counsel only basis or redacted (i) to remove references concerning the valuation of the Company, (ii) as necessary to comply with contractual arrangements, (iii) as necessary to address attorney-client or other privilege concerns and (iv) as necessary to remove sensitive personal, financial or confidential information of individuals that may be contained in any personal disclosure forms.
- Remedies. In furtherance and not in limitation of the covenants of the Parties contained in this Section 5.3, and in order to obtain all required approvals under the HSR Act, other applicable Antitrust Law and any other Requisite Regulatory Approval, promptly and in any event before the Outside Date, each of Parent, DK HoldCo, Merger Sub I, Merger Sub II and the Company shall, and shall cause their respective Subsidiaries and UPEs to, consent to, or offer or agree to, or otherwise take any action with respect to, any requirement, condition, limitation, Contract or Order to: (i) sell, dispose of or divest, or (ii) propose, negotiate or offer to effect, or consent or commit to, any such sale, disposal or divestiture of any assets, licenses, operations, rights, product lines, businesses or interest therein of Parent or the Company (or any of their respective Subsidiaries or UPEs) (any of the foregoing, a "Remedy Action"); provided, however, that notwithstanding the foregoing, (x) none of this Section 5.3, the "commercially reasonable" standard set forth in Section 5.2 or any other terms contained herein shall require, or be construed to require Parent or the Company or any of their respective Subsidiaries or UPEs, in order to obtain any Requisite Regulatory Approvals or any other approvals with any Governmental Entity, to take or agree to take (and, without the prior written consent of Parent, none of the Company nor any of its Subsidiaries will take or agree to take) any action (including any Remedy Action) that would or would reasonably be expected to result in or be a Burdensome Condition and (y) the Company and its Subsidiaries shall be required to become subject to, or consent or agree to or otherwise take any Remedy Action, or any action with respect to, any Order, requirement, condition, understanding or agreement of or with a Governmental Entity to take a Remedy Action, only if the applicable action or Remedy Action is conditioned on the consummation of the Transactions. "Burdensome Condition" means the executing or carrying out, consenting to or offering to or agreeing to, or otherwise taking of any action (including any Remedy Action) with respect to, any requirement, condition, limitation, understanding, agreement (including consent decrees and undertakings) (A) to sell, dispose of or divest any assets, licenses, operations, rights, product lines, businesses, properties or interest therein (1) of the UPE of Parent that are Parent Class B Stock, Parent Common Stock or other equity interests of Parent or any of its Subsidiaries held by the UPE of Parent or any Affiliates of the UPE of Parent (other than Parent and Subsidiaries of Parent) or (2) of Parent, the Company or any of their respective Affiliates (including UPEs) that bore a fair market value in excess of \$100,000 in the aggregate as of December 31, 2023 or (B) to lease, license

or otherwise encumber, hold separate or take any other behavioral remedies with respect to any assets, licenses, operations, rights, product lines, businesses, properties or interest therein of Parent, the Company or any of their respective Affiliates (including UPEs). For the avoidance of doubt, the phrase "lease, license or otherwise encumber, hold separate or take any other behavioral remedies" in this Section 5.3(b) does not refer to the sale, disposition, or divestiture of assets, licenses, operations, rights, product lines, businesses or interests.

- (c) <u>Litigation</u>. In furtherance and not in limitation of the covenants of the Parties contained in this Section 5.3, each of Parent, DK HoldCo, Merger Sub I, Merger Sub II and the Company shall (and shall cause their respective Subsidiaries and UPEs to) use their commercially reasonable efforts to contest and defend against the entry of, or to have vacated, lifted, reversed or overturned any decree, judgment, injunction or other Order, whether temporary, preliminary or permanent, that would restrain, prevent or delay the Closing on or before the Outside Date, including defending through litigation on the merits any Proceeding asserted in any court of competent jurisdiction with respect to the Transactions by the U.S. Federal Trade Commission, the U.S. Department of Justice or any other applicable Governmental Entity or any private party; <u>provided</u>, <u>that</u>, (i) Parent may, but shall not be required to, in its sole discretion determine whether to appeal any injunction or other Order issued by a federal district court and (ii) all fees, costs and expenses incurred in connection with the foregoing shall be paid by the Party incurring such fees, costs and expenses.
- (d) Interim Transactions. From the execution of this Agreement until the Closing, Parent, DK HoldCo, Merger Sub I, Merger Sub II and the Company, shall not (and shall cause their respective Subsidiaries and UPEs not to) acquire or agree to acquire, whether by merging or consolidating with, acquiring a portion of the assets of, acquiring equity in, or otherwise acquiring any portions of or interests in (or agreeing to do any of the foregoing), any Person set forth on Section 5.3(d) of the Company Disclosure Letter. Notwithstanding the foregoing, Parent's UPE shall not be precluded from acquiring, solely as a passive investment, an interest in any venture capital, private equity or other fund that does not own more than a five percent (5%) interest in any of the Persons set forth on Section 5.3(d) of the Company Disclosure Letter.

5.4 No Solicitation of Acquisition Proposals or Change in Recommendation.

- (a) <u>No Solicitation or Negotiation</u>. The Company agrees that neither it nor any of its Subsidiaries shall, and the Company shall cause its and its Subsidiaries' respective Representatives not to, directly or indirectly:
 - (i) initiate, solicit, propose, encourage or facilitate any inquiries or the making of any proposal or offer that constitutes, or could reasonably be expected to lead to, any Acquisition Proposal;
 - (ii) engage in, continue or otherwise participate in any discussions or negotiations regarding, or provide any non-public information or data to any Person relating to, any Acquisition Proposal or any inquiry, proposal or offer that could reasonably be expected to lead to an Acquisition Proposal;
 - (iii) agree to, approve, endorse, recommend or consummate any Acquisition Proposal;
 - (iv) enter into any letter of intent, memorandum of understanding, agreement in principle, acquisition agreement, merger agreement, option agreement, joint venture

agreement, partnership agreement or other agreement relating to any Acquisition Proposal (each, an "Alternative Acquisition Agreement"); or

- (v) otherwise facilitate any effort or attempt to make an Acquisition Proposal.
- (b) No Change in Recommendation. The Company Board shall not (i) withhold, withdraw, qualify or modify (or propose to withhold, withdraw, qualify or modify), in a manner adverse to Parent, the Board Recommendation; (ii) fail to include the Board Recommendation in the Information Statement or (iii) approve or recommend, or publicly declare advisable any Acquisition Proposal or publicly propose to enter into any Alternative Acquisition Agreement (any of the actions set forth in the foregoing clauses (i) through (iii), a "Change of Recommendation").
- (c) Existing Discussions. The Company agrees that it and its Subsidiaries shall, and the Company shall cause its and its Subsidiaries' respective Representatives to, (i) immediately cease and cause to be terminated any existing activities, discussions or negotiations with any Person with respect to any Acquisition Proposal or proposal that would reasonably be expected to lead to an Acquisition Proposal or any Alternative Acquisition Agreement, (ii) deliver a written notice to each such Person solely requesting the return or destruction of all Confidential Information previously provided to such Person by or on behalf of the Company or any of its Subsidiaries and (iii) immediately terminate any access by any Person (other than Parent, DK HoldCo, Merger Sub I, Merger Sub II and their respective Representatives) to any physical or electronic data room relating to an Acquisition Proposal.
- (d) Notice. The Company agrees that it will promptly (and, in any event, within twenty-four (24) hours) notify Parent if any inquiries, proposals or offers with respect to an Acquisition Proposal are received by, or any information is requested from or any such discussions or negotiation with respect to an Acquisition Proposal are sought to be initiated or continued with, it or any of its Subsidiaries or Representatives, indicating, in connection with such notice, the name of such Person and the material terms and conditions of any proposals or offers (including, if applicable, complete copies of any written requests, proposals or offers, including proposed agreements) and thereafter shall keep Parent informed, on a current basis, of the status and terms of any such proposals or offers (including any amendments thereto) and the status of any such discussions or negotiations, including any change in the Company's intentions as previously notified.
- (e) <u>Compliance by Subsidiaries and Representatives</u>. The Company agrees that any violation of this Section 5.4 by any Subsidiary of the Company or any of the Company's or its Subsidiaries' respective Representatives shall be deemed a breach of this Section 5.4 by the Company.
- Documents and applicable Laws (including Sections 228 and 262 of the DGCL), the Company shall obtain Written Consents from Company Stockholders holding a number of shares of Company Capital Stock sufficient to satisfy the Required Stockholder Approval in lieu of a meeting pursuant to the applicable provisions of the DGCL for the purposes of adopting this Agreement and acknowledging that such adoption is irrevocable and results in the waiver of any right of such Company Stockholders to exercise appraisal rights in connection with the Mergers pursuant to the applicable provisions of the DGCL no later than twenty-four (24) hours following the earlier of (i) 5:30 p.m. Eastern Standard Time on the fourth Business Day immediately following the date of this Agreement and (ii) the time that Parent disseminates a press release with respect to its entry into this Agreement (a "DK Deal Announcement"); provided that the Company shall not, and shall cause its Subsidiaries and its and their respective Representatives not to, seek to obtain from any Company Stockholders not listed on Schedule I hereto such Written Consents until the earlier of (x) 5:30 p.m. Eastern Standard Time on the fourth Business Day

immediately following the date of this Agreement and (y) the time of the DK Deal Announcement (such period, the "Initial Stockholder Consent Delivery Period"). As promptly as practicable and, in any event, prior to the expiration of the Initial Stockholder Consent Delivery Period, if Written Consents constituting the Required Stockholder Approval are duly executed and delivered to the Company, the Company shall deliver to Parent a copy thereof (including by electronic transmission), certified as correct and complete by an executive officer of the Company. The Company shall not, and shall cause its Subsidiaries and its and their respective Representatives not to, and shall instruct each Company Stockholder with any Representatives on the Company Board not to (and to cause their respective Representatives not to), publicly make, disseminate, issue or otherwise communicate any information with respect to the entry into this Agreement or the consummation of the Transactions unless and until the Company is permitted to seek Written Consents under this Section 5.5.

- Stockholder Support Agreements. After the date of this Agreement and prior to the Initial Effective Time, the Company shall use its commercially reasonable efforts to obtain Support Agreements executed and delivered by all Company Stockholders. As promptly as practicable and, in any event, prior to the Initial Effective Time, the Company shall deliver to Parent a copy of each Support Agreement (including by electronic transmission), certified as correct and complete by an executive officer of the Company. The Support Agreements will constitute each Company Stockholder's agreement and consent to (a) with respect to any Company Stockholder that did not execute a Written Consent, waive any right of such Company Stockholder to exercise appraisal rights in connection with the Mergers pursuant to the applicable provisions of the DGCL, (b) appoint the Stockholder Representative to act as its Representative, agent and attorney-in-fact pursuant to Article IX, (c) ratify the Company's Certificate of Incorporation in effect on the date hereof and waive any rights that the Company Stockholder may have in respect of the remediation and validation of the Certificate of Incorporation, (d) fully and unconditionally release any and all claims such stockholder now has, previously had or may in the future have as a Company Stockholder, including with respect to the calculation of the Per Share Merger Consideration set forth in the Funds Flow Memorandum, as a condition to receiving the Per Share Merger Consideration, in favor of Parent, DK HoldCo, Merger Sub I, Merger Sub II, Initial Surviving Company and Subsequent Surviving Company and their respective Subsidiaries and each of their directors, officers, managers, employees and Affiliates, (e) adopt this Agreement, including the provisions herein with respect to the Retention Holdback, the Purchase Price Adjustment Holdback, the Stockholder Representative Expense Amount and the indemnification obligations set forth in Article VIII and Article IX and (f) terminate all agreements, including stockholders agreements and voting agreements, between or among the Company and the Company Stockholders.
- Information Statement. Promptly following the execution and delivery of this Agreement (but in any event within ten (10) Business Days of the date hereof), the Company shall prepare and mail to each holder of Company Capital Stock an information statement regarding this Agreement, the Transactions, and related matters, which shall be in a form reasonably acceptable to Parent (as it may be amended or supplemented from time to time with the prior written consent of Parent (not to be unreasonably withheld, conditioned or delayed), the "Information Statement"). Statement shall constitute (i) an information statement for the Company's solicitation of consent (in the form of the Written Consent) of the holders of Company Capital Stock entitled to vote with respect to the adoption of this Agreement, (ii) a notice of the availability of appraisal rights under the DGCL and (iii) a solicitation of each Company Stockholder with respect to the adoption and execution of the Support Agreement. The Information Statement shall include, among other things, (a) a statement to the effect that the Company Board has unanimously recommended that the holders of Company Capital Stock entitled to vote approve the adoption of this Agreement, (b) a statement that adoption of this Agreement shall constitute, among other things, approval by the holders of Company Capital Stock of the withholding of the Retention Holdback, Purchase Price Adjustment Holdback, Stockholder Representative Expense Amount, Reserve Holdback and an agreement to the indemnification obligations

set forth in Article VIII, (c) a questionnaire in a form reasonably acceptable to Parent to be completed and delivered to Parent prior to the Closing Date stating that such Person is an Accredited Investor and agreeing to the transfer restrictions related to the shares of Parent Common Stock constituting the Accredited Investor Closing Stock Consideration contained therein (the "Accredited Investor Questionnaire") and (d) such other information as Parent and the Company reasonably agree is required or advisable under applicable Laws to be included therein. The Company shall prepare any other necessary documentation required to be provided to holders of Company Capital Stock, including any Dissenting Stockholders, pursuant to the DGCL. None of the information supplied or to be supplied by the Company for inclusion in the Information Statement or any amendment or supplement thereto will contain, as of the date of the delivery of such document, any untrue statement of a material fact, or will omit to state any material fact necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading. None of the information supplied or to be supplied by Parent for inclusion in the Information Statement, if any, or any amendment or supplement thereto will contain, as of the date of the delivery of such document, any untrue statement of a material fact, or will omit to state any material fact necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading.

- 5.8 Third-Party Consents. Upon the terms and subject to the conditions set forth in this Agreement, the Company and its Subsidiaries shall use their commercially reasonable efforts to obtain or provide, as applicable, prior to the Closing (and deliver evidence of the same to Parent at or prior to the Closing) all consents, waivers, approvals and notices of third parties as are necessary or advisable in connection with the consummation of the Transactions, including those listed in Section 5.8 of the Company Disclosure Letter. In connection therewith, neither the Company nor any of its Subsidiaries shall be required to (a) make any payment of a consent fee, "profit sharing" payment or other consideration (including increased or accelerated payments) or concede anything of value, (b) amend, supplement or otherwise modify any such Contract or (c) agree or commit to do any of the foregoing, in each case, that are not required under the express terms of any Contract with any Person for the purposes of giving, obtaining and/or effecting any third-party consents. Except as otherwise expressly set forth in Article VI, for the avoidance of doubt, each of the Parties acknowledges and agrees that the failure to obtain or provide (as applicable) any such consent, waiver, approval or notice shall not be, in and of itself, a condition to the Closing.
- Access and Reports; Retention of Books and Records. (a) Subject to applicable Law and the terms of the Confidentiality Agreement, the Company shall, and shall cause its Subsidiaries to, afford Parent's officers and other authorized Representatives reasonable access, during normal business hours and upon reasonable advanced notice throughout the period prior to the Closing, to its Employees, properties, books, Contracts and records and, during such period, the Company shall, and shall cause its Subsidiaries to, furnish promptly to Parent all information concerning its business, properties and personnel as Parent may reasonably request for the purpose of furthering the consummation of the Transactions and integration planning with respect thereto or obtaining the R&W Insurance Policy or any other reasonable business purpose in connection with the Transactions; provided that the foregoing shall not require the Company to disclose any information, that in the reasonable judgment of the Company in consultation with its outside legal counsel would (i) violate any applicable Law, (ii) result in the disclosure of any Trade Secrets or (iii) disclose any privileged information of the Company or any of its Subsidiaries; provided, further, that prior to withholding any access or information pursuant to the foregoing, the Company shall notify Parent in writing of the nature of the information being withheld and take any actions as may reasonably be requested by Parent to implement alternate arrangements in order to allow Parent such access or information to the fullest extent reasonably practicable under the circumstances without causing such result or violation provided, further, that Parent and its Representatives shall conduct any such activities in such a manner as not to interfere unreasonably with

the business or operations of the Company and/or its Subsidiaries. All information accessed by Parent pursuant to this Section 5.9 shall be governed by the terms of the Confidentiality Agreement.

Publicity. The initial press release with respect to the Transactions shall be a joint press release and thereafter no press or other public announcement, or public statement or comment in response to any inquiry relating to the Transactions shall be issued or made by the Company, the Company Securityholders or the Stockholder Representative without the prior consent of the other Parties (other than the Stockholder Representative) (which consent shall not be unreasonably withheld, conditioned or delayed), except as may be required by applicable Law; provided that, notwithstanding the foregoing, following Closing, the Stockholder Representative shall be permitted to announce that it has been engaged to serve as the Stockholder Representative in connection herewith as long as such announcement does not disclose any of the other terms hereof. Following the initial press release, Parent shall reasonably consult with the Company, and give due consideration to reasonable comments by the Company, prior to (x) issuing any press releases or otherwise making planned public statements with respect to the Transactions and (y) except as otherwise provided in Section 5.3, making any filings with any third party or any Governmental Entity (including any national securities exchange) with respect thereto, except (i) as may be required by applicable Law or by obligations pursuant to any listing agreement with or rules of any national securities exchange, including NASDAQ or (ii) any consultation that would not be reasonably practicable as a result of requirements of applicable Law. Parent may make any public statements in response to questions by the press, analysts, investors or those attending industry conferences or analyst or investor conference calls, so long as such statements are not inconsistent in tone or substance with previous statements made jointly by the Company and Parent.

5.11 Employee Benefits.

- (a) Parent agrees that all of the Employees who continue to remain employed with Parent or its Subsidiaries (including, following the Closing, the Subsequent Surviving Company) as of the Closing (the "Continuing Employees") shall be provided with, (i) during the period commencing on the Closing Date and ending on the one year anniversary of the Closing (or, if earlier, the date of the Continuing Employee's termination of employment with Parent or such Subsidiary), base salary or base wage and target annual cash bonus opportunities that are substantially comparable in the aggregate to those in effect for such Continuing Employee immediately prior to the Closing and, (ii) during the period commencing on the Closing Date and ending at the end of the year in which Closing occurs (or, if earlier, the date of the Continuing Employee's termination of employment with Parent or such Subsidiary), retirement and welfare benefits (excluding equity-based and other long-term incentive compensation) that are substantially comparable in the aggregate to those provided by the Company and its Subsidiaries to such Employees immediately prior to the Closing.
- (b) Parent shall use commercially reasonable efforts to (i) cause any pre-existing conditions or limitations and eligibility waiting periods under any group health plans of Parent or its Affiliates to be waived with respect to the Continuing Employees and their eligible dependents, (ii) give each Continuing Employee credit for the plan year in which the Closing occurs towards applicable deductibles and annual out-of-pocket limits for medical expenses incurred prior to the Closing for which payment has been made and (iii) give each Continuing Employee service credit for such Continuing Employee's employment with the Company and its Subsidiaries for purposes of vesting, benefit accrual and eligibility to participate under each applicable benefit plan of Parent, as if such service had been performed with Parent, except for benefit accrual under defined benefit pension plans, for purposes of qualifying for subsidized early retirement benefits or to the extent it would result in a duplication of benefits.

- (c) Prior to the Closing, if requested by Parent in writing, to the extent permitted by applicable Law and the terms of the applicable plan or arrangement, the Company shall cause the Company's 401(k) Plan to be terminated effective immediately prior to the Closing. In the event that Parent requests that the Company's 401(k) Plan be terminated, the Company shall provide Parent with evidence that the Company's 401(k) Plan has been terminated (the form and substance of which shall be subject to review and approval by Parent) not later than the day immediately preceding the Closing Date.
- (d) Prior to the Closing and thereafter (as applicable), the Company and Parent shall take all reasonably necessary actions as may be required, including amendments to the 401(k) Plan and/or the tax-qualified defined contribution retirement plan designated by Parent (the "Parent 401(k) Plan") to (i) permit each Continuing Employee to make rollover contributions of "eligible rollover distributions" (within the meaning of Section 401(a)(31) of the Code, including of loans) in the form of cash, Parent Common Stock, notes (in the case of loans) or a combination thereof, in an amount equal to the full account balance distributed or distributable to such Continuing Employee from the Company 401(k) Plan to the Parent 401(k) Plan, and (ii) have each Continuing Employee become a participant in the Parent 401(k) Plan on the Closing Date (giving effect to the service crediting provisions of Section 5.11(a)); it being agreed that there shall be no gap in participation in a tax-qualified defined contribution plan.
- (e) (1) At least three Business Days prior to the Closing Date, the Company shall submit for approval by its stockholders, in conformance with Section 280G of the Code and the regulations thereunder (the "280G Stockholder Vote"), any payments that could constitute a "parachute payment" pursuant to Section 280G of the Code (each a "Parachute Payment"), (2) at least five Business Days prior to the Closing Date, each of the applicable "disqualified individuals" (as defined under Section 280G of the Code and the regulations promulgated thereunder) shall have received a waiver form pursuant to which the disqualified individual may elect to irrevocably waive any applicable Parachute Payments, and (3) the Company shall have delivered to Parent true and complete copies of all disclosure and documents that comprise the waivers and stockholder approval of each Parachute Payment in sufficient time to allow Parent to comment thereon but no less than five Business Days prior to the 280G Stockholder Vote, and shall reflect all reasonable comments of Parent thereon.
- (f) Prior to making any broad-based written or broad-based oral communications to the Employees of the Company or any of its Subsidiaries pertaining to compensation or benefit matters that are directly affected by the Transactions, the Company shall provide Parent with a copy of the intended communication, Parent shall have no less than three (3) Business Days to review and comment on the communication, and the Company shall consider any such comments in good faith.
- (g) Nothing contained in this Agreement is intended to (i) be treated as an amendment of any particular Benefit Plan, (ii) prevent Parent or any of its Affiliates from amending or terminating any of their benefit plans or, after the Closing, any Benefit Plan in accordance their terms, (iii) prevent Parent, the Subsequent Surviving Company or any of their Affiliates, after the Closing, from terminating the employment of any Continuing Employee or (iv) create any third-party beneficiary rights in any Employee of the Company or any of its Subsidiaries, any beneficiary or dependent thereof, or any collective bargaining representative thereof, with respect to the compensation, terms and conditions of employment and/or benefits that may be provided to any Continuing Employee by Parent, the Subsequent Surviving Company or any of their respective Affiliates or under any benefit plan which Parent, the Subsequent Surviving Company or any of their respective Affiliates may maintain.
- (h) Parent will implement, or cause to be implemented, a Management Incentive Plan with terms consistent with those set forth on Exhibit H to provide for incentive payments for the benefit of certain Continuing Employees as soon as reasonably practicable following the Closing. Any

payments or benefits to be provided under the Management Incentive Plan shall be separate from payments to be provided pursuant to Company Options as set forth in Section 2.10.

(i) Following the date of this Agreement, the Company agrees that it shall use commercially reasonable efforts to facilitate Parent's and its Affiliates' negotiation of Offer Letters, to be effective following the Closing, with the Company's Employees.

5.12 Indemnification; Directors' and Officers' Insurance.

- (a) For a period of six (6) years after the Closing and to the fullest extent that the Company is or would be permitted to do so under applicable Law and its organizational documents, (i) Parent agrees that it will, and will cause the Subsequent Surviving Company to, indemnify and hold harmless each Indemnified Director or Officer against any costs or expenses (including reasonable and documented attorneys' fees), judgments, fines, losses, claims, damages or liabilities incurred in connection with, arising out of or otherwise related to any Proceeding in connection with, arising out of or otherwise related to matters existing or occurring at or prior to the Initial Effective Time, whether asserted or claimed prior to, at or after the Initial Effective Time, and (ii) Parent or the Subsequent Surviving Company shall also advance reasonable, documented out-of-pocket expenses as incurred in connection therewith (upon receipt from such Person or a written request therefor, accompanied by invoices or other relevant documentation); provided that any Person to whom expenses are advanced provides an undertaking to repay such advances if it is ultimately determined by final adjudication that such Person is not entitled to indemnification hereunder or under applicable Law.
- (b) Prior to the Initial Effective Time, the Company shall purchase and fully pay for an extended reporting period endorsement under the Company's existing directors' and officers' liability insurance coverage or otherwise in a form reasonably acceptable to Parent that shall provide each Indemnified Director or Officer with coverage in respect of claims arising out of or relating to any event that occurred at or prior to the Initial Effective Time (including in connection with the Transactions) for six (6) years following the Initial Effective Time with such terms as are not materially less favorable in the aggregate to the insured persons than the directors' and officers' liability insurance coverage presently maintained by the Company (the "D&O Tail Policy"). Parent shall not, and shall cause the Subsequent Surviving Company to not, take any action to eliminate such D&O Tail Policy. The cost of any D&O Tail Policy shall be considered a Company Transaction Expense for purposes of this Agreement.
- (c) Parent agrees that, subject to applicable Laws, all rights to indemnification, advancement of expenses and all limitations of liability existing in favor of each Indemnified Director or Officer as provided in (i) the Company's Organizational Documents as in effect as of the date of this Agreement and (ii) any other indemnification agreements (as applicable) as in effect as of the date of this Agreement, and disclosed in Section 5.12(c) of the Company Disclosure Letter, in each case, with respect to matters occurring on or prior to the Initial Effective Time, shall survive the Mergers and shall continue in full force and effect thereafter, without any amendment thereto for a period of six (6) years from the Initial Effective Time.
- (d) The provisions of this Section 5.12 shall survive the consummation of the Mergers and are intended to be for the benefit of, and shall be enforceable by, each Indemnified Director or Officer, each of whom is an express third-party beneficiary of this Section 5.12. The obligations of Parent and the Subsequent Surviving Company under this Section 5.12 shall not be terminated or modified in such a manner as to adversely affect the rights of any Indemnified Director or Officer under this Section 5.12 without the consent of such affected Indemnified Director or Officer.

- (e) If Parent or the Subsequent Surviving Company or any of their respective legal successors or permitted assigns shall (i) consolidate with or merge into any other Person and shall not be the continuing or surviving Person of such consolidation or merger or (ii) transfer all or substantially all of its properties and assets to any Person, then, and in each such case, proper provisions shall be made so that the legal successors and permitted assigns of Parent or the Subsequent Surviving Company, as applicable, shall assume all of the obligations set forth in this Section 5.12.
- 5.13 <u>Confidentiality</u>. The terms of the Confidentiality Agreement, entered into between the Company and DK Crown Holdings Inc. (formerly DraftKings Inc.), a Delaware corporation dated as of July 13, 2021 (as amended, supplemented or otherwise modified from time to time, including pursuant to the Clean Team Confidentiality Agreement, the "<u>Confidentiality Agreement</u>"), shall continue in full force and effect until the Closing, at which time such Confidentiality Agreement shall terminate.

5.14 Tax Matters.

- (a) Parent shall timely prepare and file, or cause to be timely prepared and filed all Tax Returns of the Company or any of its Subsidiaries that are first required to be filed by the Company or any of its Subsidiaries after the Closing Date with respect to a Pre-Closing Tax Period (including Straddle Periods) (any such Tax Return, a "Pre-Closing Tax Return"). Each Pre-Closing Tax Return shall be prepared in a manner consistent with past practice (unless otherwise required by applicable Law) and without a change of any election or any accounting method. With respect to any Pre-Closing Tax Return that relates to income Taxes, Parent agrees that, to the extent permitted by applicable Law at a "more likely than not" standard (or higher level of comfort), all Transaction Deductions will be treated as properly allocable to the Pre-Closing Tax Period of the Company or its applicable Subsidiary. Parent shall provide a copy of any Pre-Closing Tax Return to the Stockholder Representative at least thirty (30) days prior to the due date thereof for income Pre-Closing Tax Returns and ten (10) days prior to the due date for filing for any other Pre-Closing Tax Returns, Parent shall reflect any changes on the Pre-Closing Tax Return reasonably requested by the Stockholder Representative, and no Pre-Closing Tax Return shall be filed without the Stockholder Representative's consent (not to be unreasonably withheld, conditioned or delayed).
- In the case of Taxes of the Company or its Subsidiaries that are payable with respect to a taxable period that begins on or before and ends after the Closing Date (each such period, a "Straddle Period"), the portion of any such Taxes that relates to the portion of such taxable period ending on the Closing Date shall be: (1) in the case of Taxes (i) based upon, or related to, income, receipts, profits, wages, capital or net worth, (ii) imposed in connection with the sale, transfer or assignment of property (other than Transfer Taxes, which shall be allocated based on Section 5.14(d)), or (iii) required to be withheld, deemed to be the amount of Taxes equal to the amount which would be payable if the taxable year ended with the Closing Date; and (2) in the case of other Taxes (other than Transfer Taxes, which shall be allocated based on Section 5.14(d)), deemed to be the amount of such Taxes for the entire period multiplied by a fraction the numerator of which is the number of days in the period ending on the Closing Date and the denominator of which is the number of days in the entire period. To the extent any income of the Company or any of its Subsidiaries is attributable to an interest in an entity or arrangement classified as a partnership or other "flow-through" entity for Tax purposes, such entity or arrangement shall be treated for purposes of this Agreement as if its taxable year ended on the Closing Date and Taxes attributable to the income and gain of each such entity through the Closing Date shall be considered to be attributable to the portion of the taxable period ending on the Closing Date.
- (c) The Stockholder Representative and Parent shall provide each other with such cooperation and information as either of them reasonably may request of the other in filing any Tax Return pursuant to this Section 5.14 or in connection with any Tax Proceeding related to the Company or

any of its Subsidiaries. Such cooperation and information shall include providing copies of relevant Tax Returns or portions thereof, together with accompanying schedules, related work papers and documents relating to rulings or other determinations by Tax authorities, filing any Tax Returns, or entering into any agreements with a Governmental Entity to settle a Tax Proceeding.

- (d) All Transfer Taxes shall be borne equally by Parent, on the one hand, and the Company Securityholders, on the other hand. The Company Securityholders' 50% portion of any Transfer Taxes shall be included in Company Transaction Expenses and Parent's 50% portion shall be borne directly by Parent. Parent shall, at the Company Securityholders' expense, timely prepare and file any Tax Returns or other filings relating to such Transfer Taxes and promptly, after the Closing, provide a copy of such Tax Returns to the Stockholder Representative for its review and approval (not to be unreasonably withheld, conditioned or delayed) prior to filing (with any such expense incurred in connection with such filings to be deducted from the Stockholder Representative Expense Amount, to the extent funds are available therein, and to the extent the full amount of the funds are not available therein, then from (i) first, the Purchase Price Adjustment Holdback, (ii) second, the Reserve Holdback and (iii) third, the Retention Holdback). Parent, the Company and the Stockholder Representative (after the Closing) shall reasonably cooperate to establish any claim for exemption or exclusion from the application or imposition of any Transfer Taxes.
- After the Closing, the Stockholder Representative and Parent shall notify each other regarding, and within ten (10) days after, the receipt by such Party or any of its Affiliates (including the Company) of notice of any Tax Proceeding relating to the Company or any of its Subsidiaries to the extent relating to any Pre-Closing Tax Period or the Intended Tax Treatment (provided, however, that the failure of the Parent or the Stockholder Representative, as applicable, to give such notice shall not relieve the Parent or the Company Securityholders, as applicable, of any of its indemnification obligations. except to the extent that the Parent or the Company Securityholders (or the Stockholder Representative), as applicable, is actually prejudiced by such failure). If any Tax Proceeding that commences or continues after the Closing relates solely to the Reserved Matter or solely to Taxes related to a period ending on or prior to the Closing Date, the Stockholder Representative shall have the right, at its election and at the Company Securityholders' expense, to control such Tax Proceeding; provided, that with respect to any Tax Proceeding which the Stockholder Representative elects to control, (i) Parent shall have the right, at its own expense, directly or through its designated representatives, to participate fully in such Tax Proceeding, including to review in advance and comment upon submissions made in the course of such Tax Proceeding and to attend any in-person or telephonic meetings, (ii) subject to clause (iii), Parent's consent (not to be unreasonably withheld, conditioned or delayed) shall be required for any settlement by the Stockholder Representative that would affect the Tax liability of the Company, any of its Subsidiaries or Parent in any taxable period beginning after the Closing Date and (iii) with respect to the Reserved Matter, Parent's consent (which may be withheld in its sole discretion) shall be required for any settlement in respect thereof that (x) is in an amount that exceeds the Reserve Holdback Amount or is entered into following the release of the Reserve Holdback or (y) would be reasonably likely to increase the probability that the Company or any of its Subsidiaries would be required to pay sales Taxes or other Losses to any Texas Governmental Entity in respect of taxable periods following the Closing Date; provided, further, that with respect to any Tax Proceeding commencing or continuing after the Closing which the Stockholder Representative does not or is not permitted to control pursuant to this Section 5.14(e), (I) Parent shall diligently and in good faith defend such Tax Proceeding, (II) the Stockholder Representative shall have the right, at the Company Securityholders' own expense, directly or through its designated Representatives, to participate fully in such Tax Proceeding, including to review in advance and comment upon submissions made in the course of such Tax Proceeding and to attend any in-person or telephonic meetings and (III) the Stockholder Representative's consent (not to be unreasonably withheld, conditioned or delayed) shall be required for any settlement by Parent that could affect the Intended Tax Treatment or that would reasonably be expected to increase the liability for Taxes for which the Company

Securityholders are liable pursuant to this Agreement (including as a reduction to the Aggregate Purchase Price or as a result of an indemnification obligation).

- (f) Effective as of no later than the Closing, any and all Tax sharing or allocation agreements or arrangements to which the Company or any of its Subsidiaries is a party (other than Ordinary Course Commercial Agreements) shall be terminated, such that none of Parent or any of its Affiliates (including, after the Closing, the Company and its Subsidiaries) shall have any further liability thereunder. The parties to any such terminated agreement or arrangement shall, immediately prior to such termination, pay all amounts accrued and owing, if any, thereunder.
- Each of the DK Parties, the Company and their respective Affiliates and Subsidiaries shall use its commercially reasonable efforts to cause the Mergers, taken together, to qualify for the Intended Tax Treatment. None of the DK Parties, the Company, their respective Affiliates or Subsidiaries shall (and such Persons shall cause their respective Subsidiaries not to) take any action, or fail to take any action, whether before or after the Initial Effective Time, if such action or failure to act would reasonably be expected to prevent or impede the Mergers, taken together, from qualifying for the Intended Tax Treatment. Notwithstanding anything to the contrary in this Agreement, the Parties acknowledge and agree that in the case of any action or failure to take any action by any of the DK Parties or any of their respective Affiliates or Subsidiaries resulting in a decline in the trading price or value of Parent Common Stock, such decline shall not be, in and of itself, a violation of this Section 5.14(g). For the avoidance of doubt, nothing in this Section 5.14(g) shall (without limitation): (1) require (including as a condition of Closing) the DK Parties to amend this Agreement, restructure or alter the consideration (including the amount thereof) payable under this Agreement, except to the extent set forth in Section 5.14(k), (2) prevent the DK Parties from taking such post-Closing actions as are permitted by Section 368(a)(2)(C) of the Code and Treasury Regulations Section 1.368-2(k) or (3) prevent Parent from making "open market" repurchases as contemplated by Revenue Ruling 99-58.
- (h) Each Party to this Agreement shall promptly as reasonably practicable notify the other Parties in writing if, before the Closing Date, such Party knows or has reason to believe, in such Party's reasonable judgment, that the Mergers, taken together, would not reasonably be expected to qualify for the Intended Tax Treatment (and whether the terms of this Agreement could be reasonably amended in order to facilitate such qualification). Each of the DK Parties, the Company and their respective Affiliates shall (i) file all Tax Returns consistent with the Intended Tax Treatment (including attaching the statement described in Treasury Regulations Section 1.368-3(a) on or with the U.S. federal income Tax Returns of the Company and Parent for the taxable year that includes the Mergers), and (ii) take no Tax position inconsistent with the Intended Tax Treatment (whether in audits, Tax Returns or otherwise) except to the extent otherwise required by a "determination" as such term is used in Section 1313(a) of the Code.
- (i) (i) The Company shall use its commercially reasonable efforts to deliver to Cooley LLP, counsel to the Company ("Company's Counsel"), and Sullivan & Cromwell LLP, counsel to Parent ("Parent's Counsel"), a representation letter (the "Company Tax Representation Letter") in a form substantially similar to Exhibit I, dated as of the Closing Date and signed by a Company Tax Knowledge Party, containing representations of the Company as shall be reasonably necessary or appropriate to enable Company's Counsel and Parent's Counsel to render the opinions described in clauses (iii) and (iv) of this paragraph, (ii) Parent shall use its commercially reasonable efforts to deliver to Parent's Counsel and Company's Counsel a representation letter (the "Parent Tax Representation Letter") in a form substantially similar to Exhibit J, dated as of the Closing Date and signed by a Parent Tax Knowledge Party, containing representations of Parent as shall be reasonably necessary or appropriate to enable Company's Counsel and Parent's Counsel to render the opinions described in clauses (iii) and (iv) of this paragraph, (iii) the Company shall use its commercially reasonable efforts to

procure from Company's Counsel and deliver to the Company and Parent an opinion to the effect that, based on the basis of facts, representations and assumptions set forth or referred to in such opinion, the Mergers will qualify for the Intended Tax Treatment (the "Company Tax Opinion"), and (iv) Parent shall use its commercially reasonable efforts to procure from Parent's Counsel, and deliver to the Company and Parent an opinion to the effect that, based on the basis of facts, representations and assumptions set forth or referred to in such opinion, the Mergers will qualify for the Intended Tax Treatment (the "Parent Tax Opinion"); provided, that, for the avoidance of doubt, the delivery of any of the documents contemplated by this Section 5.14(i) shall not be a condition to the obligations of any Party to consummate the Transactions. In rendering the Company Tax Opinion or Parent Tax Opinion (as applicable), Company's Counsel and Parent's Counsel shall be entitled to rely upon customary assumptions, representations, warranties and covenants reasonably satisfactory to it, including the representations set forth in the representation letters contemplated by this Section 5.14(i).

- (j) After the Closing, none of the DK Parties, the Company, Subsequent Surviving Company, or any of their Affiliates shall (i) file any Tax Return of the Company or its Subsidiaries relating to any Pre-Closing Tax Period except in accordance with Section 5.14(a) or amend any Tax Return of the Company or its Subsidiaries relating to any Pre-Closing Tax Period, (ii) make or change any Tax election or change any method of accounting that has effect to any Tax Return of the Company or its Subsidiaries relating to any Pre-Closing Tax Period, (iii) initiate or file any voluntary disclosure or similar agreements with any Governmental Entity regarding Seller Indemnified Taxes or Tax Returns of the Company or its Subsidiaries relating to any Pre-Closing Tax Period or (iv) agree to extend or waive the statute of limitations with respect to Taxes of the Company or its Subsidiaries for a Pre-Closing Tax Period, in each such case, except (A) with the prior written consent of the Stockholder Representative (which will not be unreasonably withheld, conditioned or delayed), or (B) if such action would not be reasonably expected to materially increase the liability of the Company Stockholders for Seller Indemnified Taxes or otherwise for Taxes pursuant to this Agreement (including as a reduction to Aggregate Purchase Price).
- Notwithstanding anything to the contrary in this Agreement, if the value of the portion of the Aggregate Purchase Price (together with any other amounts treated as purchase price for Company Capital Stock, as defined for U.S. federal income tax purposes), and, for the avoidance of doubt, not taking into account the Parent Common Stock described in clause (b) of the definition of Merger Stock Consideration, payable or issuable to Company Stockholders in respect of their Company Capital Stock (the "Total Stockholder Consideration") that is comprised of shares of Parent Common Stock (the "Stock Portion") is not at least 40% of the aggregate value of the Total Stockholder Consideration as of the Closing Date (the "Reorg Threshold"), determined using the Closing Date Parent Trading Price, then the portion of the Total Stockholder Consideration that is comprised of cash (the "Cash Portion") shall be reduced and the Stock Portion shall be increased to the minimum extent necessary, determined using the Closing Date Parent Trading Price, to achieve the Reorg Threshold; provided that in the event an increase in the Stock Portion pursuant to this Section 5.14(k) would require Parent to issue any shares of Parent Common Stock (including shares that would be deliverable pursuant to converted equity awards pursuant to Section 2.10) such that after giving effect to such issuance the aggregate number of shares of Parent Common Stock that would be issued by Parent pursuant to this Agreement would exceed 19.99% of the shares of Parent Common Stock that are issued and outstanding immediately prior to such issuance (the "Parent Share Cap"), then, at the discretion of the Company, either (A) the Stock Portion shall be comprised of a number of shares of Parent Common Stock equal to the Parent Share Cap and the Cash Portion shall be increased by an amount equal to (i) the number of shares of Parent Common Stock that would have been issued but for the Parent Share Cap, multiplied by (ii) the Closing Date Parent Trading Price or (B) the Company Stockholders shall receive the Total Stockholder Consideration that such Company Stockholders would have received if this Section 5.14(k) were not in the Agreement; provided, further, that if, by disregarding the Total Stockholder Consideration

payable in respect of Company Capital Stock acquired upon exercise of an In-the-Money Option after the date hereof, the Reorg Threshold would not be achieved, this Section 5.14(k) shall be applied (and, to the minimum extent necessary to satisfy the Reorg Threshold, the Stock Portion shall be increased and the Cash Portion shall be reduced) without taking into account such Total Stockholder Consideration. For the avoidance of doubt, in no event shall Parent be required to issue any shares of Parent Common Stock pursuant to this Agreement to the extent that after giving effect to such issuance, the aggregate number of shares of Parent Common Stock that would be issued pursuant to this Agreement would exceed the Parent Share Cap. For the avoidance of doubt, for each additional share of Parent Common Stock that is added to the Stock Portion, the Cash Portion shall be reduced by the Closing Date Parent Trading Price. For purposes of calculating the Total Stockholder Consideration under this Section 5.14(k) at the Closing, (i) it shall be assumed that the Stockholder Representative Expense Amount, the Purchase Price Adjustment Holdback, the Retention Holdback Amount and the Reserve Holdback Amount are paid in full to the Company Securityholders in cash at the Closing Date and that any Aggregate Purchase Price that would otherwise be paid in respect of a Dissenting Share is paid in cash at the Closing Date, and (ii) any Post-Closing Adjustment payments pursuant to Section 2.7(a) (other than a release of the Purchase Price Adjustment Holdback) shall not be taken into account at Closing. Notwithstanding anything to the contrary in this Agreement, once the Post-Closing Adjustment is determined pursuant to Section 2.7(a), and to the extent any other adjustments are made to the purchase price (as defined for U.S. federal income tax purposes) paid or payable to Company Stockholders, appropriate adjustments to the mix of cash and Parent Common Stock payable to the Company Stockholders pursuant to Section 2.7(a) or any other section of this Agreement shall be made in a manner consistent with this Section 5.14(k), if necessary, to ensure that the Stock Portion of the Total Stockholder Consideration (as modified after giving effect to any Post-Closing Adjustment payments pursuant to Section 2.7(a) or any other adjustment to the purchase price) is not less than the Reorg Threshold (as adjusted to take into account all such adjustments after the Closing Date, and calculating the Parent Common Stock payable at such time for purposes of confirming the Reorg Threshold (including for purposes of calculating the additional number of shares Parent Common Stock payable in lieu of cash and the corresponding reduction in cash as contemplated by this Section 5.14(k)) consistent with the definition of "Closing Date Parent Trading Price" but utilizing the trading day ending on the date such adjustment to the purchase price (as defined for U.S. federal income tax purposes) becomes due and payable in lieu of the Closing Date). "Closing Date Parent Trading Price" means the average of the highest and lowest price per share of Parent Common Stock on the NASDAQ (as reported in the New York City edition of The Wall Street Journal or, if not reported thereby, as reported in another authoritative source mutually agreed by the Company and Parent) for the trading day ending on the Closing Date (or the trading day immediately following the Closing Date if the Closing Date is not a trading day). Prior to making any payment pursuant to this Agreement to any Company Stockholder, Parent and the Company (prior to the Closing) and the Stockholder Representative (from and after the Closing) will reasonably and in good faith cooperate to make any determinations or adjustments required by this Section 5.14(k), including prior to making any Post-Closing Adjustment payments.

- (I) Prior to the Closing, Parent and the Company shall use commercially reasonable efforts to cooperate with the Payments Administrator to enable the Company Stockholders to designate, in the Letter of Transmittal or a separate attachment thereto, shares of Company Capital Stock (on a per share or per lot basis) that will be exchanged for the cash consideration and stock consideration payable pursuant to this Agreement and the proportions thereof for purposes of Treasury Regulation Section 1.356-1(b).
- (m) For all applicable Tax purposes, the Parties agree that the entirety of the Aggregate Purchase Price (including, for the avoidance of doubt, the Purchase Price Adjustment Holdback, Retention Holdback Amount, Reserve Holdback Amount and Stockholder Representative Expense Amount), except for amounts payable in respect of Company Options, will be treated as

consideration for the Company Capital Stock, and none of such Aggregate Purchase Price will be allocated to the confidentiality, non-competition and non-solicitation agreements entered into in connection with this Agreement (including the Support Agreements) or otherwise treated as compensation for services, and no Party will take any action or filing position inconsistent with the foregoing.

5.15 <u>Takeover Statutes</u>. If any Takeover Statute is or may become applicable to the Mergers or the other Transactions contemplated by this Agreement, the Company and the Company Board shall grant such approvals and take such actions as are necessary so that such Transactions may be consummated as promptly as practicable on the terms contemplated by this Agreement and otherwise act to eliminate or minimize the effects of such statute or regulation on such Transactions.

5.16 R&W Insurance Policy.

- (a) Parent shall use commercially reasonable efforts to bind the R&W Insurance Policy and to execute and enter into the R&W Insurance Policy at or prior to the Closing, which such policy shall, for the avoidance of doubt, include terms to the effect that the R&W Insurance Policy insurers waive their rights to bring any claim against the Company, the Stockholder Representative or the Company Stockholders by way of subrogation, claim for contribution or otherwise, except in the event of actual fraud. Parent shall not to modify or amend any provision of R&W Insurance Policy that relates or pertains to the rights to subrogation against the Company, the Stockholder Representative or the Company Stockholders in a manner adverse to the Company, the Stockholder Representative or the Company Stockholders, without the Company's prior written consent prior to Closing or the Stockholder Representative's prior written consent at or following the Closing. For the avoidance of doubt, Parent acknowledges and agrees that the obtaining of R&W Insurance Policy is not a condition to the Closing.
- (b) The Company shall, and shall cause its Subsidiaries to, cooperate with Parent, the R&W Insurance Policy insurers and their respective Representatives in connection with binding R&W Insurance Policy, including by (i) cooperating with the R&W Insurance Policy insurers' due diligence investigation of the Company and its Subsidiaries and (ii) using commercially reasonable efforts to answer any due diligence questions or document requests, including by providing as promptly as reasonably practicable responses and underlying documents reasonably responsive to such questions or requests, as applicable.

5.17 Transfer of Private Placement Shares; Registration Statement.

As promptly as practicable following the Closing (and in any event no later than fifteen (15) Business Days after the Closing Date, Parent shall prepare and file with the SEC a registration statement of Parent registering the resale by each Accredited Investor who has returned a completed Support Agreement and a completed Selling Stockholder Questionnaire in the form attached hereto as Exhibit K (a "Selling Stockholder Questionnaire") of the shares of Parent Common Stock to be issued to the Accredited Investors hereunder (such shares, the "Registrable Shares," and such registration statement, the "Resale Registration Statement") prior to the Closing Date provided that any such securities shall cease to be Registrable Shares on the earliest to occur of when (i) such Registrable Shares have been disposed of in accordance with the Resale Registration Statement, (ii) such Registrable Shares shall have been sold in accordance with Rule 144 (or any similar provision then in effect), (iii) with respect to a holder, all such securities still held by such holder are eligible for resale by such holder, together with its Affiliates, pursuant to Rule 144 under the Securities Act (or other exemption from registration under the Securities Act) without any volume, manner of sale or other limitations (including current public information requirements under Rule 144) and provided that any restrictive legends on all Registrable Shares have been removed (or been caused to be removed) by Parent or (iv) such Registrable Shares have ceased to be outstanding. The Company shall use its commercially reasonable efforts to cause to be delivered to Parent completed, executed and delivered by the Company Stockholders who are Accredited Investors the Selling Stockholder Questionnaires. Each Accredited Investor that has returned a completed Support Agreement and a properly completed Selling Stockholder Questionnaire prior to the Closing Date is referred to herein as a "Selling Stockholder."

- If Parent is eligible to file a Resale Registration Statement on Form S-3 pursuant to Rule 462(e) under the Securities Act (an "Automatic Resale Registration Statement"), the Resale Registration Statement shall be an Automatic Resale Registration Statement. If Parent is not eligible to use an Automatic Resale Registration Statement, the Resale Registration Statement shall be on Form S-3, or if Form S-3 is not available to Parent, another appropriate form. If the Resale Registration Statement is not an Automatic Resale Registration Statement, Parent shall use its commercially reasonable efforts to have the Resale Registration Statement declared effective under the Securities Act as promptly as practicable after such Resale Registration Statement is filed. Parent will advise the Selling Stockholders promptly after Parent receives any written request by the SEC for amendment of the Resale Registration Statement or any written SEC comments thereon. Once the Resale Registration Statement is declared effective, Parent shall notify each of the Selling Stockholders of such declaration, and thereafter, subject to the other applicable provisions of this Agreement, shall use commercially reasonable efforts to cause the Resale Registration Statement to be continuously effective and usable until the date that is the first anniversary of the Closing Date, or such earlier time when no Registrable Shares remain (the "Registration Period"). Parent shall use commercially reasonable efforts to cause the Resale Registration Statement (including the documents incorporated therein by reference) to comply as to form in all material respects with all applicable requirements of the Securities Act and the Exchange Act.
- The Resale Registration Statement (or any prospectus or prospectus supplement forming a part of such Resale Registration Statement), as initially filed, shall include the Registrable Shares of all Selling Stockholders for whom Parent has received completed Selling Stockholder Questionnaires and completed Support Agreements on or before the Closing Date and, subject to any applicable lock-up arrangements, allow for distributions by such Selling Stockholders (A) to an Affiliate of such Selling Stockholder, (B) if the Selling Stockholder is an individual, to an immediate family member or trust for the benefit of such Selling Stockholder or one or more of such Selling Stockholder's immediate family members, (C) pursuant to the laws of testamentary or intestate succession or otherwise involuntarily transferred by operation of law or (D) if the Selling Stockholder is a partnership, corporation, or limited liability company, to any one or more partners, stockholders or members thereof; provided, however, that such Selling Stockholder shall give Parent written notice prior to the time of such transfer stating the name and address of the transferee and identifying the Registrable Shares being transferred to the transferee and such transferee shall first execute and deliver to Parent a joinder agreement in form and substance reasonably satisfactory to Parent agreeing to be bound by the terms of this Agreement applicable to Selling Stockholders. Any such transfer of such Registrable Shares pursuant to this Section 5.17(c) is referred to as a "Permitted Transfer," and any such transferee of Registrable Shares pursuant to this Section 5.17(c) is referred to herein as a "Permitted Transferee." On a date requested by the Stockholder Representative in writing (so long as such date is at least fifteen (15) Business Days after such request), Parent shall use commercially reasonable efforts to file an amendment or supplement, as appropriate, to the Resale Registration Statement (and any prospectus or prospectus supplement forming a part of such Resale Registration Statement) to include the Registrable Shares of any Permitted Transferees to the extent such filing is required in order to permit the Permitted Transferees to offer and sell pursuant to the Resale Registration Statement the Registrable Shares received by the Permitted Transferees in the Permitted Transfer. Parent further agrees to provide in the Resale Registration Statement (and in any prospectus or prospectus supplement forming a part of such Resale Registration Statement) that all Permitted Transferees shall, by virtue of receiving Registrable Shares in a Permitted Transfer, be deemed to be selling stockholders under the Resale Registration Statement (or any such prospectus or prospectus supplement) with respect to the Registrable Shares received by such

Permitted Transferees in such Permitted Transfers. Parent shall only be required to file two (2) such amendments or supplements. Parent shall include the plan of distribution in the Resale Registration Statement (and any prospectus or prospectus supplement forming a part of such Resale Registration Statement).

- Parent shall notify each of the Selling Stockholders as promptly as reasonably practicable upon discovery that, or upon the discovery of the happening of any event as a result of which, the Resale Registration Statement (or any prospectus or prospectus supplement forming a part of such Resale Registration Statement) contains an untrue statement of a material fact or omits any fact necessary to make the statements therein not misleading in the light of the circumstances under which they were made, and use commercially reasonable efforts to supplement or amend such prospectus as promptly as reasonably practicable so that such prospectus will not contain an untrue statement of a material fact or omit to state any material fact necessary in order to make the statements made therein not misleading in the light of the circumstances under which they were made. After the Resale Registration Statement becomes effective, Parent shall notify each of the Selling Stockholders, of any request by the SEC that Parent amend or supplement such Resale Registration Statement or prospectus, and Parent shall use commercially reasonable efforts to prepare and file with the SEC such amendments and supplements to the Resale Registration Statement and the prospectus used in connection therewith as may be reasonably necessary to keep the Resale Registration Statement effective and to comply with the provisions of the Securities Act with respect to the disposition of all Registrable Shares covered by the Resale Registration Statement. Parent shall furnish to the Selling Stockholders such numbers of copies of a prospectus, including a preliminary prospectus, and any supplement to any prospectus, as required by the Securities Act. Parent shall reasonably cooperate with the Selling Stockholders with respect to the removal of any restrictive legends as the Selling Stockholders may reasonably request in order to facilitate the disposition of their Registrable Shares, subject to each Selling Stockholder promptly providing any information reasonably requested by Parent to facilitate such action (including any customary representation letters or other information reasonably requested by Parent, its transfer agent or counsel). If a Selling Stockholder holds Registrable Shares that are eligible to be sold (i) without restriction under Rule 144 (other than the restrictions set forth under Rule 144(i)) or (ii) pursuant to an effective Resale Registration Statement, then, at such Selling Stockholder's written request, accompanied by such customary representations and other documents as Parent or its transfer agent shall reasonably request, Parent shall, reasonably promptly, use commercially reasonable efforts to cause Parent's transfer agent to remove any restrictive legend set forth on the Registrable Shares held by such Selling Stockholder (including, if required by Parent's transfer agent, by delivering to Parent's transfer agent a direction letter and opinion of counsel in form reasonably acceptable to Parent's transfer agent).
- (e) Parent shall indemnify and hold harmless each Selling Stockholder with Registrable Shares included in the Resale Registration Statement against any and all Losses to which such Selling Stockholder may become subject by reason of (a) any untrue statement or alleged untrue statement of a material fact contained in or incorporated by reference into the Resale Registration Statement or any amendments or supplements thereto, (b) the omission or alleged omission to state therein a material fact required to be stated therein, or necessary to make the statements therein not misleading or (c) any violation or alleged violation by Parent (or any of its Affiliates) of the Securities Act, the Exchange Act or any state securities Law in connection with the Resale Registration Statement or the offer or sale of Registrable Shares thereunder, in each case, solely to the extent such Losses directly arise out of or result from any claim or cause of action made against such Selling Stockholder by an unaffiliated third party (excluding, for the avoidance of doubt, any Permitted Transferee of such Selling Stockholder) who purchased such Registrable Shares from such Selling Stockholder; provided, that Parent shall not be liable for any such Losses to the extent such Losses arise out of or are based upon information furnished to Parent by or on behalf of such Selling Stockholder expressly for use in the Resale Registration Statement; provided, further, that the indemnity obligations set forth in this Section

5.17(e) shall not apply (i) to amounts paid in settlement of any such Losses if such settlement is effected without the prior written consent of Parent or (ii) if Parent is not given reasonably prompt notice of the claim. Parent shall have the right to assume the defense and settlement of any claim or suit for which Parent may be responsible for indemnification under this Section 5.17(e).

Upon receipt of written notice from Parent that the Resale Registration Statement (or any prospectus or prospectus supplement forming a part of such Resale Registration Statement) contains or would contain an untrue statement of a material fact or omits any fact necessary to make the statements therein not misleading in the light of the circumstances under which they were made, each of the Selling Stockholders shall forthwith discontinue disposition of Registrable Shares pursuant to the Resale Registration Statement until he, she or it has received copies of a prospectus that does not contain an untrue statement of a material fact or omit to state any material fact necessary in order to make the statements made therein not misleading in the light of the circumstances under which they were made (it being understood that Parent hereby covenants to prepare and file such supplement or amendment as soon as reasonably practicable after the time of such notice as set forth in Section 5.17(d) above), or until he, she or it is advised in writing by Parent that the use of the prospectus may be resumed (any such period, a "Suspension Period"). If the filing, initial effectiveness or continued use of the Resale Registration Statement at any time would require Parent to make a public disclosure that in the good faith judgment of Parent, after consultation with outside counsel, (i) would be required to be made in the Resale Registration Statement (or any prospectus or prospectus supplement forming a part of such Resale Registration Statement) in order for the Resale Registration Statement or supplement or prospectus not to contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein (in the case of any prospectus and any preliminary prospectus, in the light of the circumstances under which they were made) not misleading, (ii) would reasonably be likely to require premature disclosure of information that (x) if disclosed at such time, could be harmful to the interests of Parent and its stockholders, (y) would not be required to be made at such time if the Resale Registration Statement were not being filed or (z) otherwise requires the public disclosure of material non-public information, or (iii) Parent has a bona fide business purpose for not making such information public, including if such disclosure would materially interfere with a business, acquisition or divestiture or financing transaction involving Parent or any of its Subsidiaries or would require the inclusion in the Resale Registration Statement of financial statements that are unavailable to Parent for reasons beyond Parent's control, Parent may, upon giving prompt written notice of such action to the Selling Stockholders, delay the filing or initial effectiveness of, or suspend use of, the Resale Registration Statement for the shortest period of time, but in no event more than forty-five (45) days, determined in good faith by Parent to be necessary or appropriate for such purpose (any such period, a "Blackout Period"). In the event Parent exercises its rights under this Section 5.17(f), the Selling Stockholders shall suspend, immediately upon their receipt of the notice referred to above, their use of the prospectus relating to the Resale Registration Statement in connection with any sale or offer to sell Registrable Shares, Parent shall promptly notify the Selling Stockholders of the expiration of any period during which it exercised its rights under this Section 5.17(f). Notwithstanding anything to the contrary in this Section 5.17, in no event shall the duration of all Suspension Periods and Blackout Periods continue for more than ninety (90) days in the aggregate during any 365-day period; provided, that during any such Suspension Period or Blackout Period, (x) Parent may only register any securities for sale by Parent or for resale by any other securityholders during such Suspension Period or Blackout Period if such securities are also subject to such Suspension Period or Blackout Period, (y) Parent shall work in good faith to limit the duration of any such Suspension Period or Blackout Period and (z) Parent shall suspend the usage of all other then-effective registration statements and related prospectuses during such Suspension Period or Blackout Period other than registration statements and related prospectuses for the sale of securities to or by employees of Parent or a Subsidiary of Parent pursuant to equity incentive plans. In the event of the postponement of the filing of any amendment or supplement, or of the suspension of the use of, or trading under, the Resale Registration Statement pursuant to Section 5.17, the Registration Period shall be extended by that number of days equal to the number of days in the Suspension Period and/or Blackout Period. Promptly following the cessation or discontinuance of the facts and circumstances forming the basis for any such postponement or suspension, Parent shall amend the Resale Registration Statement and/or amend or supplement any related prospectus or prospectus supplement to the extent necessary, and take all other actions reasonably necessary, to allow the offer and sale or other disposition of Registrable Shares to recommence as promptly as reasonably practicable, and shall promptly notify each of the Selling Stockholders in writing when such offers and sales or other dispositions of Registrable Shares under the Resale Registration Statement may recommence.

- (g) From the date of this Agreement until the earlier of the date this Agreement is terminated in accordance with its terms and the end of the Registration Period, Parent shall use its commercially reasonable efforts to make and keep public information available, as those terms are defined in Rule 144 under the Securities Act, and file with the SEC in a timely manner all reports and other documents required to be filed by Parent under the Securities Act and the Exchange Act.
- (h) All of the expenses incurred in connection with any registration of Registrable Shares pursuant to this Agreement, including all SEC fees, blue sky registration and filing fees, listing notices and filing fees, printing fees and expenses, transfer agents' and registrars' fees and expenses and all fees and expenses of Parent's outside counsel and independent accountants of Parent shall be paid by Parent. Parent shall not be responsible for any selling expenses of any Selling Stockholder (including any broker's fees or commissions) or fees or expenses of outside counsel or independent accountants of any Selling Stockholder or, to the extent incurred prior to the Closing, the Company in connection with the Resale Registration Statement.
- (i) Parent shall use commercially reasonable efforts to cause the shares of Parent Common Stock being issued in the Transactions to be approved for listing (subject to notice of issuance) on NASDAQ effective as of the Closing.
- 5.18 <u>Director and Officer Resignations</u>. The Company shall deliver to Parent prior to the Closing a resignation letter (which may be conditioned upon the occurrence of the Closing) from each then-current member of the Company Board and all then-current officers of the Company, unless Parent requests that any such resignations not be delivered no later than three Business Days prior to Closing.
- 5.19 <u>Delivery of Data Room</u>. At the Closing the Company shall deliver or cause to be delivered to each of Parent and the R&W Insurance Policy insurer two copies of a CD, DVD-ROM or USB drive containing a true and complete copy of the Data Room as of the date of Closing Date.
- 5.20 <u>Company Indebtedness</u>. The Company shall use its reasonable efforts to, on a timely basis and as reasonably requested by Parent, (i) deliver (or cause to be delivered) at the Closing, notices of the payoff, discharge and termination of any outstanding Indebtedness and other obligations of the Company under the MVB Loan Documents or any other Indebtedness described in clauses (a) or (b) of the definition of "Indebtedness" ("<u>Borrowed Money Indebtedness</u>") and other obligations required to be paid off, discharged or terminated in accordance with and within the time periods required by the MVB Loan Documents or other Contracts governing such Borrowed Money Indebtedness (which notices may be conditioned upon the Closing to the extent permitted under the MVB Loan Documents and such applicable Contracts governing such Borrowed Money Indebtedness) (collectively, the "<u>Debt Payoff Amount</u>"), (ii) take all other actions required or advisable prior to the Closing to facilitate the repayment of the obligations with respect to and termination of the commitments under such Borrowed Money Indebtedness and the release of any encumbrances and termination of all guarantees granted in connection therewith, and (iii) obtain customary executed payoff letters or other similar evidence in form and substance reasonably acceptable to Parent in respect of such Borrowed Money Indebtedness (the "<u>Debt</u>

<u>Payoff Letters</u>") at least three Business Days prior to Closing. At its option and upon reasonable notice to the Company, Parent shall be permitted to deliver to the Company in respect of any Borrowed Money Indebtedness prior to or at the Closing an amount in immediately available funds equal to the amount set forth in the Debt Payoff Letters (which amount shall be deemed to have been paid directly to the applicable lender(s) party to such Debt Payoff Letters for purposes of this Agreement), and the Company shall use such funds to effect the repayment or prepayment (as the case may be) at the Closing of such Borrowed Money Indebtedness set forth in such Debt Payoff Letters.

- 5.21 <u>Lobbying</u>. Prior to undertaking or engaging in any new initiative to lobby for legislative or regulatory changes in any jurisdiction in respect of any existing line or business or future line of business of the Company or any of its Subsidiaries, the Company shall notify Parent in writing (with such notice specifying the jurisdiction and substance of such initiative in reasonable detail) reasonably in advance of commencing such lobbying initiative.
- 5.22 <u>Continued Licensure</u>. The Company shall use commercially reasonable efforts to ensure that all Company Gaming Licenses remain unaffected by the departure of any Person from the Company.
- 5.23 <u>Further Assurances</u>. The Parties shall execute and deliver, or shall cause to be executed and delivered, such documents and other instruments and shall take, or shall cause to be taken, such further actions as may be reasonably required to carry out the provisions of this Agreement and give effect to the Transactions.

ARTICLE VI

CONDITIONS

- 6.1 <u>Conditions to Each Party's Obligation to Consummate the Transactions</u>. The respective obligations of each Party to consummate the Transactions is subject to the satisfaction or waiver at or prior to the Closing of each of the following conditions:
- (a) <u>Stockholder Approval</u>. This Agreement shall have been duly adopted by holders of shares of Company Capital Stock constituting the Required Stockholder Approval.
- (b) Regulatory Approvals. (i) The waiting period (and any extension thereof) applicable to the consummation of the Transactions under the HSR Act shall have expired or been earlier terminated, (ii) all other authorizations, consents, orders, approvals, filings and declarations, and all expirations of waiting periods required from, any Governmental Entity required for the consummation of the Transactions, in each case for purposes of this clause (ii) as set forth on Section 6.1(b) of the Company Disclosure Letter and Section 6.1(b) of the Parent Disclosure Letter shall have been filed, occurred or been obtained (all such authorizations, consents, orders, approvals, filings and declarations and the lapse of all such waiting periods, including under the HSR Act, being the "Requisite Regulatory Approvals") and (iii) all such Requisite Regulatory Approvals shall be in full force and effect.
- (c) <u>Laws or Orders</u>. No Governmental Entity of competent jurisdiction shall have enacted, enforced, entered, issued or promulgated any Order or Law that is in effect and has the effect of (i) making the Transactions illegal or otherwise prohibiting consummation of the Transactions or (ii) causing the Transactions to be rescinded following their consummation.
- 6.2 <u>Conditions to Obligation of Parent, DK HoldCo, Merger Sub and Merger Sub I</u>. The respective obligations of Parent, DK HoldCo, Merger Sub I and Merger Sub II to consummate the

Transactions is also subject to the satisfaction or waiver by Parent at or prior to the Closing of the following conditions:

- (a) Representations and Warranties of the Company.
- (i) Each of the representations and warranties set forth in Section 3.1 (Organization, Good Standing and Qualification), Section 3.3 (Authority; Approval), Section 3.6(d) (Absence of Certain Changes) and Section 3.12 (Takeover Statutes) shall be true and correct in all respects as of the date of this Agreement and as of the Closing as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct in all respects as of such earlier date);
- (ii) Each of the representations and warranties set forth in Section 3.2(a) Section 3.2(b), Section 3.2(e), Section 3.2(f), Section 3.2(g), and Section 3.2(i) (Capital Structure) shall be true and correct (without giving effect to any qualifications or limitations as to de minimis) in all but de minimis respects as of the date of this Agreement and as of the Closing as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct in all but de minimis respects as of such earlier date (without giving effect to any qualifications or limitations as to de minimis));
- (iii) Each of the representations and warranties set forth in Section 3.2(c), Section 3.2(d), Section 3.2(h) and Section 3.2(j) (Capital Structure), Section 3.4(b)(i) (No Violations of Organizational Documents) and Section 3.23 (Brokers and Finders) shall be true and correct in all material respects as of the date of this Agreement and as of the Closing as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct in all material respects as of such earlier date); and
- (iv) Each of the representations and warranties of the Company set forth in Article III (other than those set forth in the foregoing clauses (i), (ii) and (iii) of this Section 6.2(a)) and any certificates delivered by the Company or the Stockholder Representative pursuant to this Agreement shall be true and correct (without giving effect to any materiality qualifiers, including "Material Adverse Effect," contained therein) in all respects as of the date hereof and as of the Closing as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct in all respects as of such earlier date) except where the failure of any such representation and warranty to be so true and correct (without giving effect to any qualifications or limitations as to de minimis, materiality or Material Adverse Effect therein) has not or would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect with respect to the Company and its Subsidiaries.
- (b) <u>Performance of Obligations of the Company</u>. The Company shall have performed and complied in all material respects with all covenants required to be performed by it under this Agreement at or prior to the Closing.
- (c) <u>Company Closing Certificate</u>. Parent, DK HoldCo, Merger Sub I and Merger Sub II shall have received a certificate signed on behalf of the Company by an executive officer of the Company (solely in his or her capacity as such and not in his or her personal capacity, and without

personal liability) certifying that the conditions set forth in Section 6.2(a), Section 6.2(b) and Section 6.2(d) have been satisfied.

- (d) <u>No Company Material Adverse Effect</u>. Since the date of this Agreement, there shall not have occurred a Material Adverse Effect with respect to the Company and its Subsidiaries that is continuing as of immediately prior to the Initial Effective Time.
- (e) <u>Continued Employment</u>. (x) At least 80% of the Key Employees, including the Key Employees specified on Section 6.2(f) of the Company Disclosure Letter shall be employed by the Company (except to the extent such employment is terminated by death or Disability of a Key Employee) and shall not have served any resignations or be subject to pending termination, and (y) the Designated Stockholder shall have signed an Offer Letter with Parent or a Subsidiary thereof and shall not have taken any action to rescind such Offer Letter.
- (f) <u>Payments Administrator Agreement</u>. Parent shall have received the Payments Administrator Agreement duly executed by the Stockholder Representative and the Payments Administrator.
- (g) <u>Escrow Agreement</u>. Parent shall have received the Escrow Agreement duly executed by the Escrow Agent and the Stockholder Representative.
- (h) <u>Stockholders Agreements</u>. All agreements, including stockholders, voting agreements and any side letters, between or among the Company and the Company Stockholders shall have been terminated and be of no further force or effect as of the Initial Effective Time.
- (i) <u>Continued Operations</u>. None of the Company or any of its Subsidiaries shall have received any written notice or other written correspondence from any Governmental Entity of competent jurisdiction that remains in effect providing that the Company or any of its Subsidiaries is not entitled or permitted to engage in lottery courier services, and none of the Company or any of its Subsidiaries shall have ceased to operate its lottery courier business, in (a) any jurisdiction listed on Section 6.2(i)(a) of the Company Disclosure Letter and (b) more than 25% of the number of jurisdictions listed on Section 6.2(i)(b) of the Company Disclosure Letter.
- (j) <u>Support Agreement</u>. Support Agreements shall have been obtained from Company Stockholders comprising, in the aggregate, no less than ninety percent (90%) of the shares of Company Capital Stock outstanding immediately prior to the Initial Effective Time.
- (k) <u>Written Consents</u>. Written Consents from Company Stockholders holding a number of shares of Company Capital Stock sufficient to satisfy the Required Stockholder Approval shall have been obtained and delivered to Parent.
- (I) Actions Under Agreements. (i) The Company shall have (A) delivered to Parent the written consent of each party to the contracts set forth on Section 6.2(I)(i)(A) of the Company Disclosure Letter to the consummation of the Transactions or (B) satisfied the conditions set forth in Section 6.2(I)(i)(B) of the Company Disclosure Letter with respect to any such consents that are not obtained and (ii) the Company shall have delivered to Parent amendments or waivers to the contracts set forth on Section 6.2(I)(ii) of the Company Disclosure Letter executed by each party thereto reflecting the modifications or the waivers of rights described on Section 6.2(I)(ii) of the Company Disclosure Letter.

- 6.3 <u>Conditions to Obligation of the Company</u>. The obligation of the Company to consummate the Transactions is also subject to the satisfaction or waiver by the Company at or prior to the Closing of the following conditions:
 - (a) Representations and Warranties.
 - (i) Each of the representations and warranties of Parent, DK HoldCo, Merger Sub I and Merger Sub II set forth in Section 4.1 (Organization, Good Standing and Qualification), Section 4.3 (Authority; Approval) and Section 4.13 (Absence of Certain Changes) shall be true and correct in all respects as of the date of this Agreement and as of the Closing as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct in all respects as of such earlier date);
 - (ii) Each of the representations and warranties of Parent, DK HoldCo, Merger Sub I and Merger Sub II set forth in Section 4.12 (Capitalization) shall be true and correct in all but de minimis respects as of the date of this Agreement and as of the Closing as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct in all material respects as of such earlier date);
 - (iii) Each of the representations and warranties of Parent, DK HoldCo, Merger Sub I and Merger Sub II set forth in Section 4.2 (Parent Common Stock), Section 4.4(b)(i) (No Violations of Organizational Documents), Section 4.8 (Merger Sub I) and Section 4.9 (Merger Sub II) shall be true and correct in all material respects as of the date of this Agreement and as of the Closing as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct in all material respects as of such earlier date); and
 - (iv) Each of the representations and warranties of Parent, DK HoldCo, Merger Sub I and Merger Sub II set forth in Article IV (other than those set forth in the foregoing clauses (i), (ii) and (iii) of this Section 6.3(a)) and any certificates delivered by Parent, DK HoldCo, Merger Sub I or Merger Sub II pursuant to this Agreement shall be true and correct (without giving effect to any materiality qualifiers, including "Material Adverse Effect," contained therein) as of the date hereof and as of the Closing as though made on and as of such date and time (except to the extent that any such representation and warranty speaks as of an earlier date, in which case such representation and warranty shall be true and correct as of such earlier date), except where the failure of any such representation and warranty to be so true and correct has not or would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect with respect to Parent and its Subsidiaries.
- (b) <u>Performance of Obligations of Parent, DK HoldCo, Merger Sub I and Merger Sub II</u>. Each of Parent, DK HoldCo, Merger Sub I and Merger Sub II shall have performed and complied in all material respects with all covenants required to be performed by it under this Agreement at or prior to the Closing.
- (c) <u>Closing Certificate</u>. The Company shall have received a certificate signed on behalf of Parent, DK HoldCo, Merger Sub I and Merger Sub II by an executive officer of Parent (solely in his or her capacity as such and not in his or her personal capacity, and without personal liability) certifying that the conditions set forth in Section 6.3(a), Section 6.3(b) and Section 6.3(f) have been satisfied.

- (d) <u>Payments Administrator Agreement</u>. The Company shall have received the Payments Administrator Agreement duly executed by Parent and the Payments Administrator.
- (e) <u>Escrow Agreement</u>. The Company shall have received the Escrow Agreement duly executed by the Escrow Agent, Parent and the Stockholder Representative.
- (f) No Parent Material Adverse Effect. Since the date of this Agreement, there shall not have occurred a Material Adverse Effect with respect to Parent and its Subsidiaries that is continuing as of immediately prior to the Initial Effective Time.

ARTICLE VII

TERMINATION

- 7.1 <u>Termination by Mutual Written Consent</u>. This Agreement may be terminated and the Transactions may be abandoned at any time prior to the Closing by mutual written agreement of the Company and Parent.
- 7.2 <u>Termination by Either Parent or the Company</u>. This Agreement may be terminated and the Transactions may be abandoned at any time prior to the Closing by either Parent or the Company, by giving written notice of such termination to the other Party, if:
- the Closing shall not have occurred on or prior to November 11, 2024 (the "Outside Date"): provided, however, that if as of the Outside Date any of the conditions to Closing set forth in Section 6.1(b) (solely as it relates to any applicable Antitrust Laws) or Section 6.1(c) (solely as it relates to any applicable Antitrust Laws) have not been satisfied or, to the extent permitted by applicable Law, waived but all other conditions to Closing set forth in Article VI have been satisfied or, to the extent permitted by applicable Law, waived (except for those conditions that by their nature are to be satisfied at the Closing), then the Outside Date shall be automatically extended without further action by the Parties to February 11, 2025 (the "First Extended Outside Date") (and in the case of such extension, any reference to the Outside Date in this Agreement shall be a reference to the First Extended Outside Date); provided, further, that if as of the First Extended Outside Date any of the conditions to Closing set forth in Section 6.1(b) (solely as it relates to any applicable Antitrust Laws) or Section 6.1(c) (solely as it relates to any applicable Antitrust Laws) have not been satisfied or, to the extent permitted by applicable Law, waived but all other conditions to Closing set forth in Article VI have been satisfied or, to the extent permitted by applicable Law, waived (except for those conditions that by their nature are to be satisfied at the Closing), then the First Extended Outside Date shall be automatically extended without further action by the Parties to May 11, 2025 (the "Second Extended Outside Date") (and in the case of such extension, any reference to the Outside Date in this Agreement shall be a reference to the Second Extended Outside Date); provided, further, that the right to terminate this Agreement pursuant to this Section 7.2(a) shall not be available to any Party that has breached in any material respect its obligations under this Agreement in any manner that shall have proximately contributed to the occurrence of the failure of a condition to the consummation of the Transactions; or
- (b) any Law or Order permanently restraining, enjoining or otherwise prohibiting the consummation of the Transactions shall have become final and non-appealable; provided that the right to terminate this Agreement pursuant to this Section 7.2(b) shall not be available to any Party that has breached in any material respect its obligations under this Agreement in any manner that shall have proximately contributed to the occurrence of the failure of a condition to the consummation of the Transactions.

- 7.3 <u>Termination by Parent</u>. This Agreement may be terminated and the Transactions may be abandoned by Parent by giving written notice of such termination to the Company, if:
- (a) if at any time prior to the Closing, there has been a breach by the Company of any covenant or agreement set forth in this Agreement, or if any representation or warranty of the Company shall have become untrue, in either case, such that the conditions in Section 6.2(a) or Section 6.2(b) would not be satisfied (and such breach or failure to be true and correct is not curable prior to the Outside Date, or if curable prior to the Outside Date, has not been cured within the earlier of (i) thirty (30) days after the giving of notice thereof by Parent to the Company or (ii) the Outside Date); provided, however, that the right to terminate this Agreement pursuant to this Section 7.3(a) shall not be available to Parent if it has breached in any material respect its obligations under this Agreement in any manner that shall have proximately contributed to the occurrence of the failure of a condition to the consummation of the Transactions;
- (b) if Written Consents from Company Stockholders holding a number of shares of Company Capital Stock sufficient to satisfy the Required Stockholder Approval shall not have been obtained and delivered to Parent prior to the end of the Initial Stockholder Consent Delivery Period; provided, however, that this termination right shall no longer apply and shall expire if not invoked by Parent prior to actual delivery of such Written Consents; or
- (c) if at any time prior to the time the Required Stockholder Approval is obtained the Company Board shall have made a Change of Recommendation.
- 7.4 Termination by the Company. This Agreement may be terminated and the Transactions may be abandoned by the Company by giving written notice of such termination to Parent if at any time prior to the Closing, there has been a breach by Parent, DK HoldCo, Merger Sub I or Merger Sub II of any covenant or agreement set forth in this Agreement, or if any representation or warranty of Parent, DK HoldCo, Merger Sub I or Merger Sub II shall have become untrue, in either case, such that the conditions in Section 6.3(a) or Section 6.3(b) would not be satisfied (and such breach or failure to be true and correct is not curable prior to the Outside Date, or if curable prior to the Outside Date, has not been cured within the earlier of (i) thirty (30) days after the giving of notice thereof by the Company to Parent or (ii) the Outside Date); provided, however, that the right to terminate this Agreement pursuant to this Section 7.4 shall not be available to the Company if it has breached in any material respect its obligations under this Agreement in any manner that shall have proximately contributed to the occurrence of the failure of a condition to the consummation of the Transactions.
- 7.5 <u>Effect of Termination and Abandonment</u>. In the event of termination of this Agreement and the abandonment of the Transactions pursuant to this Article VII, this Agreement shall become void and of no effect with no liability to any Person on the part of any Party (or any of its Representatives or Affiliates); <u>provided</u>, <u>however</u>, notwithstanding anything in this Agreement to the contrary, (a) no such termination shall relieve any Party of any liability or damages to any other Party resulting from any Willful Breach of this Agreement and (b) the provisions set forth in this Section 7.5, Section 7.6, Section 7.7, Section 9.3, Article X and the Confidentiality Agreement shall survive the termination of this Agreement.

7.6 Bridge Loan.

(a) Notwithstanding anything to the contrary in this Agreement, if this Agreement is validly terminated:

- (i) by either Parent or the Company pursuant to Section 7.2(a), and (A) any of the conditions set forth in Section 6.1(b)(i), Section 6.1(b)(iii), or Section 6.1(c) (in each case, solely with respect to Antitrust Laws) have not been satisfied or waived by Parent as of the date of such termination and (B) all other conditions set forth in Section 6.1 and Section 6.2 have been satisfied or waived (or in the case of conditions that by their nature are to be satisfied at the Closing, are capable of being satisfied if the Closing were to occur on the date of such termination); or
- (ii) by either Parent or the Company pursuant to Section 7.2(b) solely with respect to a Law that is an Antitrust Law or an Order issued solely with respect to an Antitrust Law; or
- (iii) by the Company pursuant to Section 7.4 solely as a result of a breach by Parent, DK HoldCo, Merger Sub I or Merger Sub II of any covenant or agreement set forth in Section 5.3;

then, provided that there has not been any Willful Breach by the Company of any of its obligations under Section 5.3 that shall have proximately contributed to the occurrence of the failure of a condition to the consummation of the Transactions, then, in any case of the foregoing clauses (i), (ii) or (iii) (any of the events described in the foregoing clauses (i), (ii) or (iii), a "Regulatory Termination Event"), Parent and the Company shall use commercially reasonable efforts to negotiate in good faith entry into an unsecured bridge credit facility with a term of not less than eighteen (18) months (the "Bridge Loan Credit Facility", and the loan in respect of such credit facility, the "Bridge Loan") pursuant to which Parent, or an Affiliate thereof, is the lender and the Company is the borrower and which shall have a principal amount equal to all reasonable, documented and out-of-pocket fees and expenses incurred by the Company and its Subsidiaries to the extent incurred in connection with the Company's efforts to obtain all required approvals under the HSR Act or any other applicable Antitrust Law pursuant to this Agreement, including in connection with any so-called "second request" or any litigation on the merits of any claim asserted in any court with respect to the Transactions by the U.S. Federal Trade Commission, the U.S. Department of Justice or any other applicable Governmental Entity or any private party (including the reasonable and documented out-of-pocket fees and expenses of all attorneys, consultants, economists and other experts retained by the Company and all reasonable documented out-of-pocket travel and related expenses) (the "Bridge Loan Amount"); provided that the Bridge Loan Amount (x) shall not include any such fees or expenses that in the aggregate exceed \$10,000,000 (for clarity, without taking into account the Extension Payment Amount) that were incurred without the prior written consent of Parent (not to be unreasonably withheld, conditioned or delayed), (y) shall bear interest at a rate or rates to be mutually agreed between Parent and the Company, each acting in good faith, and (z) shall contain such other terms and conditions to be mutually agreed between Parent and the Company, each acting in good faith, in the case of each of clauses (y) and (z), based on the then prevailing market rates and terms applicable to a bridge credit facility in an amount equal to the sum of the Bridge Loan Amount and the Extension Payment Amount for a company in the same industry and at the same stage of development as the Company. In the event the Parties enter into the Bridge Loan Credit Facility, the Company agrees that it shall be required to repay the Extension Payment Amount to Parent pursuant to the terms and conditions of the Bridge Loan Credit Facility and that the terms and conditions applicable to the Extension Payment Amount set forth in the Bridge Loan Credit Facility shall be the same as those that apply to the Bridge Loan Amount.

(b) Each Party acknowledges that the agreements in this Section 7.6 are an integral part of this Agreement and that, without these agreements, the Parties would not have entered into this Agreement. The Parties further acknowledge that the Bridge Loan Credit Facility contemplated by the foregoing Section 7.6(a), and the funding of the foregoing pursuant to the foregoing Section 7.6(a), shall not constitute a penalty but rather liquidated damages for losses and damages in a reasonable amount that will compensate the Company in the circumstances in which the amount pursuant to the foregoing

Section 7.6(a) is to be funded for the efforts and resources expended and opportunities foregone while negotiating this Agreement and in reliance on this Agreement and on the expectation of the consummation of the Transactions, which amount would otherwise be impossible to calculate with precision. In the event that this Agreement is terminated as a result of a Regulatory Termination Event, except in the case of a Willful Breach by Parent, DK HoldCo, Merger Sub I or Merger Sub II of any covenant or agreement set forth in Section 5.3, (i) the lending of the Bridge Loan Amount by DK HoldCo or any of its Affiliates to the Company pursuant to Section 7.6(a) shall be the sole and exclusive remedy of the Company, its Subsidiaries and its and their respective Representatives against Parent, DK HoldCo, Merger Sub I, Merger Sub II and their respective Subsidiaries and Representatives for damages or other relief pursuant to this Agreement and (ii) in no event will the Company, its Subsidiaries or its or their respective Representatives or any other person seek to recover any damages or seek any other remedy based on a claim in law or equity with respect to (A) any loss suffered, directly or indirectly, as a result of the failure of the Transactions to be consummated, (B) the termination of this Agreement, (C) any liabilities or obligations arising under this Agreement or (D) any claims or actions arising out of or relating to any breach, termination or failure of or under this Agreement.

7.7 Treatment of Extension Payment.

- (a) In the event this Agreement is terminated pursuant to Article VII and (i) the Transactions are abandoned by Parent pursuant to Section 7.3 or (ii) a Material Adverse Effect with respect to the Company and its Subsidiaries has occurred and is continuing as of the date of the termination of this Agreement (each of (i) and (ii), a "Company Payment Trigger"), then the Company shall pay to Parent an amount in cash equal to the Extension Payment Amount (the "Company Termination Fee") by wire transfer of immediately available funds within five (5) Business Days following the date of such termination and abandonment. In the event this Agreement is terminated pursuant to Article VII in any circumstance not set forth in the immediately preceding sentence, the Extension Payment Amount shall be retained by the Company without any further obligation to Parent in respect of the Extension Payment Amount.
- (b) Each Party acknowledges that the agreements in this Section 7.7 are an integral part of this Agreement and that, without these agreements, the Parties would not have entered into this Agreement. In the event that the Company is permitted to retain the Extension Payment Amount, any Losses in respect of claims or actions asserted by the Company, any of its Affiliates or any of its or their respective Representatives against any DK Party, its Affiliates or any of its or their respective Representatives shall first be offset against the Extension Payment Amount.

ARTICLE VIII

SURVIVAL; INDEMNIFICATION

8.1 Survival.

- (a) The representations and warranties of Parent, DK HoldCo, Merger Sub I and Merger Sub II contained in or made pursuant to Article IV shall not survive the Closing; <u>provided</u> that the representations and warranties set forth in Section 4.15 (No Other Representations or Warranties) shall survive the Closing in full force and effect indefinitely.
- (b) The representations and warranties of the Company set forth in Article III shall survive the Closing in full force and effect until the date that is eighteen (18) months after the Closing Date; provided, however, that:

- (i) the representations and warranties set forth in Section 3.31 (No Other Representations or Warranties) shall survive the Closing in full force and effect indefinitely;
- (ii) the Company Fundamental Representations shall survive the Closing in full force and effect until the date that is six (6) years after the Closing Date; and
- (iii) the representations and warranties contained in or made pursuant to Section 3.16 (Taxes) shall survive the Closing in full force and effect until the 60th calendar day following the expiration of the statute of limitations otherwise applicable to claims for breach of the federal and state Laws governing the liabilities, actions and other matters referred to in such representations and warranties, giving effect to any waivers, tolling periods or extensions thereof (each such survival period set forth in this Section 8.1, a "Representations Survival Period").
- (c) (i) The covenants and agreements of the Parties contained in this Agreement that require performance prior to the Closing will terminate one (1) year following the Closing and (ii) the covenants and agreements of the Parties contained in this Agreement that by their terms are required to be performed after the Closing shall survive the Closing in full force and effect until the date on which such covenants and agreements have been fully performed or otherwise satisfied in accordance herewith.
- Survival Period of an item is shorter or longer than the statute of limitations that would otherwise have been applicable to such item, then, by Contract, the applicable statute of limitations with respect to such item shall be reduced or increased to the applicable Representations Survival Period contemplated hereby. Notwithstanding the foregoing, if written notice of a claim has been given pursuant to and in accordance with this Article VIII prior to the expiration of the applicable Representations Survival Period by the Person seeking indemnification for such claim, then the representations, warranties, covenants or agreements with respect to which such claim alleges a breach shall survive to the extent until the Final Determination thereof. No Party shall have any liability to any Person with respect to any provision of this Agreement or the subject matter thereof following the applicable Representations Survival Period specified in Section 8.1, which supersedes any statute of limitations that would otherwise apply, and no Party shall thereafter assert any claim, cause of action, right or remedy, or any Proceeding, with respect to such provision or the subject matter thereof. No provision of this Article VIII shall apply to or limit any claim that a Party committed actual fraud in making any representation or warranty, which may be brought at any time.

8.2 <u>Indemnification by Company Stockholders.</u>

- (a) Following the Closing until the applicable survival dates provided in Section 8.1, Parent, DK HoldCo, Merger Sub I, Merger Sub II and their respective Affiliates, Representatives, successors and permitted assigns, in their capacity as such (collectively, the "Parent Indemnified Parties") shall be entitled to be indemnified, defended, held harmless and reimbursed for, from and against all Losses to the extent imposed on, incurred or suffered by or asserted against any Parent Indemnified Party in connection with or arising out of:
 - (i) any breach by the Company of (A) any Company Fundamental Representations or (B) any Non-Fundamental Company Representations; it being understood that for purposes of this Section 8.2(a)(i) any qualifications relating to materiality (such as the terms "material", "Material Adverse Effect" and "de minimis") contained in such representation or warranty (except (i) with respect to the representations and warranties set forth in Section 3.6(d) and (ii) for the term "Material Contract") shall be disregarded for purposes of determining whether such representation or warranty was breached or not true and correct and the quantity of Losses;

- (ii) any breach by the Company of any of the covenants or agreements of the Company set forth in this Agreement required by this Agreement to be performed or complied with by the Company;
- (iii) any claims or threatened claims by or purportedly on behalf of any holder or former securityholder of the Company or rights to acquire capital stock of the Company in such holder's or former holder's capacity as such (other than claims to enforce such Person's rights under this Agreement or any or any document or instrument delivered in connection with the Transactions) relating to matters arising at or before the Closing, including claims related to distributions, dividends, stock splits, stock dividends, conversions or preemptive rights;
- (iv) any claim, cause of action, right or remedy, or any Proceeding, asserted at any time by any actual or alleged securityholder of the Company (other than claims to enforce such Person's rights under this Agreement or any or any document or instrument delivered in connection with the Transactions) relating to the allocation or entitlement to a portion of the Aggregate Purchase Price or other amount paid or to be paid in connection with the Transactions, including any assertion of contractual, employment or other rights and any assertion of rights to own or acquire any security;
- (v) any amounts required to be paid to any Company Stockholder that are in excess of what such Company Stockholder has or would have received hereunder in respect of such Company Stockholder's Company Capital Stock, including in respect of such Company Stockholder's Dissenting Shares (and any interest required to be paid thereon) that are in excess of what such Company Stockholder is allocated to receive pursuant to the Funds Flow Memorandum or would have received hereunder in respect of such Dissenting Shares had such Company Stockholder not been a holder of Dissenting Shares, and any other Losses incurred in connection with the process of attending to such Company Stockholder's Company Capital Stock or Dissenting Shares; or
 - (vi) any Seller Indemnified Taxes.
- (b) <u>Limitations</u>. Recovery by the Parent Indemnified Parties of their aggregate Losses will be subject to the following limitations:
 - (i) The Parent Indemnified Parties shall not be entitled to be indemnified, defended, held harmless or reimbursed pursuant to this Article VIII in respect of any Losses of the type described in Section 8.2(a)(i)(B) (A) unless and until the aggregate amount of such Losses exceeds \$1,875,000 (the "Deductible"), in which case the Parent Indemnified Parties shall be entitled to be indemnified, defended, held harmless and reimbursed for, from and against all such Losses that are in excess of the Deductible and (B) in excess of \$1,875,000 (the "Cap"); provided, however, that the Cap shall be reduced to \$1,500,000 on the twelve (12) month anniversary of the Closing Date; provided, further, that notwithstanding anything in the foregoing to the contrary, the limitations set forth in this Section 8.2(b)(i) shall not apply with respect to Losses of the type described in Section 8.2(a) (other than Section 8.2(a)(i)(B)) or Losses resulting from actual fraud, in each case, with respect to which all Losses shall be recoverable from the first dollar.
 - (ii) With respect to Losses claimed pursuant to this Article VIII (other than any Losses of the type described in Section 8.2(a)(i)(B)), Losses shall be recovered (A) first from the Retention Holdback, and (B) thereafter, to the extent such Losses exceed the amount claimed from the Retention Holdback in all unresolved or unsatisfied Claim Notices, directly from each Company Stockholder according to its Pro Rata Share of such Losses up to the aggregate Per Share Merger Consideration actually received by such Company Stockholder (based on the Closing Parent Trading Price); provided that (1) in respect of the Reserved Matter, recovery for the Losses shall be recovered

(I) first from the Reserve Holdback, (II) second, to the extent such Losses exceed the amount claimed from the Reserve Holdback, from the Retention Holdback and (III) thereafter, to the extent such Losses exceed the amount claimed from the Reserve Holdback and Retention Holdback in all unresolved or unsatisfied Claim Notices, directly from each Company Stockholder according to its Pro Rata Share of such Losses up to the aggregate Per Share Merger Consideration actually received by such Company Stockholder (based on the Closing Parent Trading Price); and (2) the Parent Indemnified Parties shall be entitled to recover directly from each Company Stockholder according to its Pro Rata Share of any Losses incurred by the Parent Indemnified Parties pursuant to Section 8.2(a)(i)(B) up to the amount of Losses recovered by the Parent Indemnified Parties pursuant to this Article VIII (other than any Losses of the type described in Section 8.2(a)(i)(B)) from the Retention Holdback. In no event will the aggregate liability of any Company Stockholder under this Agreement, the Transactions or any document or instrument delivered in connection herewith or therewith exceed the aggregate Per Share Merger Consideration actually received by such Company Stockholder (based on the Closing Parent Trading Price), except in the case of actual fraud by such Company Stockholder. Notwithstanding anything to the contrary, this Section 8.2(b) shall not apply to Article IX, which shall be binding upon, and enforceable by the Stockholder Representative in its entirety against, the Company Securityholders.

8.3 Claim Procedures.

- (a) Claim Notice and Claim Objection Notice.
- Except as set forth in Section 5.14 with respect to Taxes, in order for any Parent Indemnified Party to duly make a valid claim under Section 8.2, the Parent Indemnified Party must provide written notice (a "Claim Notice") to the Stockholder Representative promptly, and in any event within twenty (20) Business Days of the first date on which such Parent Indemnified Party has knowledge of facts, matters or circumstances from which it is reasonably apparent that such an occurrence is likely to have occurred; provided that no delay on the part of the Parent Indemnified Party in so notifying the Stockholder Representative shall relieve the Company Stockholders from any obligation hereunder unless the Company Stockholders are materially prejudiced thereby in its defense thereof. The Claim Notice shall set forth a description in reasonable detail of the occurrence(s) specified in Section 8.2 which the Parent Indemnified Party alleges to have occurred, a description of the facts and circumstances giving rise to such occurrences, the estimated amount of Losses imposed, incurred, suffered or asserted in connection therewith or arising therefrom (to the extent then ascertainable) and the method of computation of such Losses. The Stockholder Representative and the Parent Indemnified Party shall use commercially reasonable efforts to avoid production of confidential information (consistent with applicable Law) to third parties and to cause all communications among employees, counsel and others representing any party to a Third Party Claim to be made so as to preserve any applicable attorney-client or work product privileges.
- (ii) In the event that the Stockholder Representative objects in writing within thirty (30) days after receipt of a Claim Notice by delivery of a written notice of objection containing a reasonably detailed description of the facts and circumstances supporting an objection to the applicable indemnification claim (an "Claim Objection Notice") to Parent, the Stockholder Representative and Parent shall attempt in good faith to agree upon the rights of the respective parties with respect to each of such claims. If the Stockholder Representative and Parent should so agree, a memorandum setting forth such agreement shall be prepared and signed by both parties. In the event that there is a dispute relating to any Claim Objection Notice that cannot be settled in accordance with Section 8.3(a)(ii), each of Parent and Stockholder Representative may file suit with respect to such dispute in any court having jurisdiction in accordance with Section 10.4. Notwithstanding the

foregoing, the Company and the Stockholder Representative agree that this Section 8.3(a)(ii) shall not prejudice or otherwise limit in any manner the rights of the Parent Indemnified Parties in Article VIII.

- (b) In the event the Claim Notice results from any written notice of a claim, or a Proceeding asserted or threatened against the Parent Indemnified Party by a third party (a "Third Party Claim"):
 - (i) The Parent Indemnified Parties shall provide a Claim Notice to the Stockholder Representative with respect to any Third Party Claim that might give rise to any Loss by the Parent Indemnified Parties promptly, and in any event within than twenty (20) Business Days of the first date on which such Parent Indemnified Party has knowledge of the Third Party Claim; provided that no delay on the part of the Parent Indemnified Party in so notifying the Stockholder Representative shall relieve the Company Stockholders from any obligation hereunder unless the Company Stockholders are materially prejudiced thereby in its defense thereof. Such notice shall be accompanied by copies of any written demand, summons, complaint or other pleading with respect to such Third Party Claim that may have been served, or any notice or other any documentation submitted by the third party making such Third Party Claim.
 - From and after receipt of notice of a Third Party Claim pursuant to Section 8.3(b)(i), Stockholder Representative shall have the right to assume and conduct, at the Company Stockholders' expense, the defense against any Third Party Claim. If the Stockholder Representative has not elected to proceed with the defense of such Third Party Claim on its own by delivery of a written notice to Parent within 30 days of delivery of the Claim Notice, or fails to diligently contest such Third Party Claim after such election, Parent may elect (by written notice delivered to the Stockholder Representative) to control the defense of such Third Party Claim or to prosecute such Third Party Claim to conclusion or settlement. The Party that is controlling the defense of such Third Party Claim (the "Controlling Party") shall (A) keep the other Party (the "Non-Controlling Party") reasonably and promptly apprised of all significant factual and procedural developments (including additional information which may come under the Controlling Party's control) in connection therewith, (B) consider in good faith recommendations made by the Non-Controlling Party with respect to the defense of such Third Party Claim and (C) not enter into any settlement, compromise or consent to judgment with respect to such Third Party Claim without the prior written consent of the Non-Controlling Party (such consent not to be unreasonably withheld, conditioned or delayed and shall be deemed to have been given unless the Non-Controlling Party shall have objected within thirty (30) days after written request for such consent). The Non-Controlling Party shall use commercially reasonable efforts to make available information in its possession that is not privileged, commercially sensitive or a Trade Secret and that is reasonably available and necessary for the defense of the Third Party Claim as the Controlling Party may reasonably request and shall reasonably cooperate with the Controlling Party in such defense.
 - (iii) The Stockholder Representative shall be the sole party entitled to exercise any rights of the Company Stockholders with respect to Third Party Claims under this Section 8.3(b).

8.4 Recoveries.

(a) <u>Insurance</u>. In calculating the amount of any Loss, the proceeds actually received by the Parent Indemnified Party or any of its Affiliates under any insurance policy, including under the R&W Insurance Policy, or pursuant to any claim, recovery, settlement or payment by or against any other Person, in each case relating to the matters described in the Claim Notice shall be deducted, except to the extent that the adjustment itself would excuse, exclude or limit the coverage of all or part of such Loss.

The Parent Indemnified Party shall use commercially reasonable efforts to seek to recover any such insurance, including under the R&W Insurance Policy, or other proceeds from third parties.

- (b) <u>Reimbursement</u>. If a Parent Indemnified Party recovers an amount from a third party in respect of a Loss after all or a portion of such Loss has been paid by the Company Stockholders pursuant to this Article VIII, the Parent Indemnified Party shall promptly remit to the Payments Administrator the excess (if any) of (i) the amount paid by the Company Stockholders in respect of such Loss, plus the amount received from the third party in respect thereof, <u>less</u> (ii) the full amount of the Loss;
- (c) <u>No Double Recovery</u>. No Parent Indemnified Party shall be entitled to recover more than once in respect of the same Loss (notwithstanding that such Loss may result from more than one of the occurrences specified in Section 8.2).
- (d) Aggregate Cash Consideration. No party will be entitled to be indemnified for an amount pertaining to any Loss to the extent that such amount pertaining to such Loss was included or taken into account in the calculation of the Aggregate Cash Consideration as finally determined pursuant to Section 2.6.
- (e) <u>Mitigation</u>. Following the Closing, each Parent Indemnified Party shall use commercially reasonable efforts to mitigate all Losses to the extent required by applicable Law; <u>provided</u> that the costs to mitigate shall be deemed to be indemnifiable Losses.

8.5 Indemnity Payments.

- If the entire amount then in the Retention Holdback or the Reserve Holdback, as applicable, as contemplated by Section 8.2(b) is sufficient to cover the aggregate indemnification obligations of the Company Stockholders pursuant to this Article VIII, then no later than five (5) Business Days following any Final Determination of the claims set forth in the applicable Claim Notice, Parent and the Stockholder Representative shall deliver a joint written instruction, duly executed by each of Parent and the Stockholder Representative to the Escrow Agent directing the Escrow Agent to transfer and release to Parent an amount from the Retention Holdback or Reserve Holdback, as applicable, equal to the amount of the Losses for which the Company Stockholders are liable, as finally determined in accordance with this Agreement. If the entire amount then in the Retention Holdback and the Reserve Holdback, as applicable, is insufficient to cover such Losses, then, subject to the limitations set forth in this Article VIII: (A) then no later than five (5) Business Days following any Final Determination of the claims set forth in the applicable Claim Notice, Parent and the Stockholder Representative shall deliver a joint written instruction, duly executed by each of Parent and the Stockholder Representative to the Escrow Agent directing the Escrow Agent to transfer and release to Parent the remainder of the Retention Holdback and Reserve Holdback, as applicable; and (B) Parent will notify and deliver to the Stockholder Representative the applicable Notice Schedule and, within twenty (20) Business Days of such notice, each Company Stockholder will pay to Parent in cash (on behalf of itself and any other Parent Indemnified Party) its Pro Rata Share of any such shortfall. Following its receipt of any such notice from Parent described in the foregoing clause (B) regarding amounts owed directly by the Company Stockholders to Parent, the Stockholder Representative shall, subject to its prior receipt of a Notice Schedule, promptly notify each Company Stockholder of any obligation with respect to any such shortfall; provided, that the Stockholder Representative's obligation to notify each Company Stockholder of any obligation to pay amounts owed pursuant to Section 8.5(a) shall be satisfied upon transmission of such email notice to the email addresses included in the Notice Schedule.
- (b) Notwithstanding the foregoing Section 8.5(a), if the Parent Indemnified Parties are entitled to recover directly from each Company Stockholder pursuant to Section 8.2(b)(ii)(B)(2), then

no later than five (5) Business Days following any Final Determination of the claims set forth in the applicable Claim Notice, Parent will notify and deliver to the Stockholder Representative the applicable Notice Schedule and, within twenty (20) Business Days of such notice, each Company Stockholder will pay to Parent in cash (on behalf of itself and any other Parent Indemnified Party) its Pro Rata Share of the amount of Loss for which it is liable hereunder. Following its receipt of any such notice from Parent described in the foregoing sentence regarding amounts owed directly by the Company Stockholders to Parent, the Stockholder Representative shall, subject to its prior receipt of a Notice Schedule, promptly notify each Company Stockholder of any obligation with respect to any such shortfall; provided, that the Stockholder Representative's obligation to notify each Company Stockholder of any obligation to pay amounts owed pursuant to Section 8.5(b) shall be satisfied upon transmission of such email notice to the email addresses included in the Notice Schedule.

- (c) All payments made by the Company Stockholders to a Parent Indemnified Party in respect of any claim pursuant to Section 8.2 shall be treated as adjustments to the consideration paid pursuant to the Transactions for Tax purposes. For the purposes of this Article VIII, a Company Stockholder's Pro Rata Share shall be deemed to exclude Dissenting Shares from clause (b) of the definition of Pro Rata Share.
- 8.6 Exclusive Remedies. Following the Closing, no Party shall assert against any other Party any claim, cause of action, right or remedy, or any Proceeding, relating to this Agreement, the Transactions or any document or instrument delivered in connection herewith or therewith, other than (i) as set forth in Article II, (ii) claims pursuant to this Article VIII, (iii) claims arising from actual fraud, Willful Breach or criminal activity on the part of a Party hereto in connection with the Transactions, or (iv) claims under the R&W Insurance Policy and (v) claims pursuant to Section 10.4(d). Following the Closing, the claims and remedies specified in clauses (i) through (v) of the previous sentence shall constitute the Parties' sole and exclusive rights and remedies available to the Parties for any and all Losses or other claims relating to or arising out of this Agreement, the Transactions and any document or instrument delivered in connection herewith or therewith, and shall supersede all other rights and remedies available at Law or in equity (including any right of rescission). All claims for indemnification, compensation or reimbursement under this Article VIII will be asserted by Parent on behalf of the applicable Parent Indemnified Parties. Notwithstanding anything to the contrary, this Section 8.6 shall not apply to Article IX, which shall be binding upon, and enforceable by the Stockholder Representative in its entirety against, the Company Securityholders.
- 8.7 No Effect on R&W Insurance Policy. Notwithstanding anything to the contrary contained herein, no limitations (including any survival limitations and other limitations set forth in this Article VIII), qualifications or procedures in this Agreement shall be deemed to limit or modify the ability of Parent to make claims under or recover under the R&W Insurance Policy.

ARTICLE IX

STOCKHOLDER REPRESENTATIVE

9.1 <u>Acknowledgement.</u> By voting in favor of, or consenting to, the adoption of this Agreement, the consummation of the Transactions and receiving the benefits thereof, including the right to receive the consideration payable in connection with the Transactions, each Company Securityholder hereby irrevocably constitutes and appoints the Stockholder Representative as of the Closing, and the Stockholder Representative hereby accepts such appointment as of the Closing, to act on behalf of such Company Securityholder as its representative, agent and attorney-in-fact, with full power of substitution, to act in its name, place and stead for all purposes in connection with this Agreement, any agreements ancillary hereto, and the Transactions. The power of attorney granted in this Section 9.1 by each

Company Securityholder is coupled with an interest, is irrevocable (except as provided by Section 9.5), may be delegated by the Stockholder Representative and shall survive the death, dissolution or incapacity of any Company Securityholder and the consummation of the Transactions. Parent, DK HoldCo, Merger Sub I and Merger Sub II shall be entitled to rely on the actions of the Stockholder Representative taken on behalf of the Company Securityholders in connection with the Transactions. Without limiting the generality of the foregoing, by signing this Agreement, each Company Securityholder shall be deemed to have authorized and empowered the Stockholder Representative as of the Closing, on its behalf and in its name, to:

- (a) act for and on behalf of each Company Securityholder in any and all capacities and to do and perform every act and thing required or permitted to be done, in the reasonable judgment of the Stockholder Representative, in connection with the Transactions;
- (b) give and receive all notices or documents given or to be given to or by the Company Securityholders pursuant hereto or in connection herewith and to give and receive and accept service of legal process on behalf of Company Securityholders in connection with any Proceeding relating to or arising under this Agreement, the Escrow Agreement or the Payments Administrator Agreement;
- (c) engage counsel and such accountants and other advisors for the Stockholder Representative;
- (d) incur, set aside reserves for, and pay such expenses on behalf of the Company Securityholders and Stockholder Representative, in connection with the Transactions as the Stockholder Representative may deem appropriate;
 - (e) expend the Stockholder Representative Expense Amount;
- (f) as contemplated by Section 9.2, require any of the Company Securityholders to advance or pay funds to the Stockholder Representative to cover existing or expected future expenses of the Stockholder Representative in such amounts as the Stockholder Representative, in his sole discretion, specifies, which amounts shall be contributed by the Company Securityholders in accordance with their respective Pro Rata Shares; provided, however, that in the event that any Company Securityholder does not fully fund such Company Securityholder's Pro Rata Share, then notwithstanding any other provision of this Agreement, such Company Securityholder shall not, thereafter, share in any further distributions under this Agreement until all fully funding Company Securityholders have thereafter received aggregate distributions pursuant to this Agreement equaling the dollar amount of their full Pro Rata Shares so funded;
- (g) take such action, including commencing or defending any Proceeding, on behalf of the Company Securityholders as the Stockholder Representative may deem appropriate in respect of the following:
 - (i) any matters related to the calculation, payment or distribution of purchase price or any component thereof contemplated by Article II (including any dispute relating thereto);
 - (ii) any claims (including settlement thereof) made by any Parent Indemnified Party for indemnification pursuant to Article VIII; and
 - (iii) any amendment, waiver or consent with respect to, or termination of, this Agreement, the Escrow Agreement or the Payments Administrator Agreement;

- (h) take such other action as the Stockholder Representative is authorized to take under this Agreement, the Escrow Agreement or the Payments Administrator Agreement;
- (i) make all determinations required or permitted under this Agreement, the Escrow Agreement or the Payments Administrator Agreement;
- (j) represent each Company Securityholder or all or certain Company Securityholders as a group in any litigation and negotiate or enter into settlements and compromises relating to any disputes arising in connection with the Transactions; and
- (k) take all relevant action in all such other matters as the Stockholder Representative may deem necessary or appropriate to consummate the Transactions.
- Stockholder Representative Expense Amount. The Stockholder Representative Expense 9.2 Amount will be used for any fees and expenses incurred by the Stockholder Representative in connection with its duties under this Agreement. In the event that the Stockholder Representative Expense Amount is insufficient to cover the fees and expenses incurred by the Stockholder Representative in connection with its duties under this Agreement, each of the Company Securityholders shall be obligated to pay an amount equal to its Pro Rata Share of any such deficiency. Without limiting the foregoing, the Stockholder Representative shall be entitled to deduct any such fees and expenses in excess of the then-remaining portion of the Stockholder Representative Expense Amount from any amounts payable to the Company Securityholders in accordance with the terms of this Agreement, the Escrow Agreement or the Payments Administrator Agreement. The Company Securityholders will not receive any interest or earnings on the Stockholder Representative Expense Amount and irrevocably transfer and assign to the Stockholder Representative any ownership right that they may otherwise have had in any such interest or earnings. The Stockholder Representative will hold these funds separate from its corporate funds and will not voluntarily make these funds available to its creditors in the event of bankruptcy. As soon as practicable following the completion of the Stockholder Representative's responsibilities, the Stockholder Representative will deliver any portion of the then-remaining Stockholder Representative Expense Amount to the Payments Administrator for further distribution to the Company Securityholders in proportion to each Company Securityholder's Pro Rata Share. For tax purposes, the Stockholder Representative Expense Amount will be treated as having been received and voluntarily set aside by the Company Securityholders at the time of Closing.
- Exculpation and Indemnification of Stockholder Representative. 9.3 The Stockholder Representative will incur no liability in connection with its services pursuant to this Agreement, any agreements related hereto, and the Transactions except to the extent resulting from its gross negligence, bad faith, or willful misconduct. The Stockholder Representative shall not be liable for any action or omission pursuant to the advice of counsel. The Company Securityholders shall indemnify the Stockholder Representative against any reasonable, documented, and out-of-pocket losses, liabilities and expenses ("Representative Losses") arising out of or in connection with this Agreement, any agreements related hereto, and the Transactions, in each case as such Representative Loss is suffered or incurred; provided, that in the event that any such Representative Loss is finally adjudicated to have been caused by the gross negligence, bad faith, or willful misconduct of the Stockholder Representative, the Stockholder Representative will reimburse the Company Securityholders the amount of such indemnified Representative Loss to the extent attributable to such gross negligence, bad faith, or willful misconduct. Representative Losses may be recovered by the Stockholder Representative from (i) the Stockholder Representative Expense Amount and (ii) any other funds that become payable to the Company Securityholders under this Agreement at such time as such amounts would otherwise be distributable to the Company Securityholders; provided, that while the Stockholder Representative may be paid from the aforementioned sources of funds, this does not relieve the Company Securityholders from their obligation

to promptly pay such Representative Losses as they are suffered or incurred. In no event will the Stockholder Representative be required to advance its own funds on behalf of the Company Securityholders or otherwise. Notwithstanding anything in this Agreement to the contrary, any restrictions or limitations on liability or indemnification obligations of, or provisions limiting the recourse against non-parties otherwise applicable to, the Company Securityholders set forth elsewhere in this Agreement are not intended to be applicable to the indemnities provided to the Stockholder Representative hereunder. The foregoing indemnities will survive the Closing, the resignation or removal of the Stockholder Representative or the termination of this Agreement.

- 9.4 <u>Reliance</u>. In the performance of its duties hereunder, the Stockholder Representative shall be entitled to (a) rely upon any document or instrument reasonably believed to be genuine, accurate as to content and signed by any Company Securityholder or any Party hereunder and (b) assume that any Person purporting to give any notice in accordance with the provisions hereof has been duly authorized to do so. Following the Closing, a decision, act, consent or instruction of the Stockholder Representative, including an amendment, extension or waiver of this Agreement, shall constitute a decision of the Company Securityholders and shall be final, binding and conclusive upon the Company Securityholders. Parent shall be entitled to rely conclusively upon any statement or action within the scope of Section 9.1 made or taken by the Stockholder Representative as being the action of the Company Securityholders.
- Replacement of the Stockholder Representative. The Stockholder Representative may resign at any time upon ten (10) Business Days prior written notice to Parent and the Company, provided that if the Stockholder Representative resigns, the Company Securityholders shall designate a replacement shareholder representative within ten (10) Business days following such resignation in accordance with this Section 9.5. The Stockholder Representative may be replaced at any time by the affirmative vote or written consent of the Company Stockholders representing at least fifty percent (50%) of the Fully Diluted Non-Option Shares outstanding immediately prior to the Initial Effective Time. The Company Stockholders exercising such replacement right shall cause (a) written notice of such resignation or replacement (along with the replacement Stockholder Representative's notice details and evidence of the requisite Company Stockholder consent having been obtained) to be promptly given to Parent and each other Company Stockholder and (b) the replacement Stockholder Representative to accede to this Agreement as the Stockholder Representative, it being understood that no replacement of the Stockholder Representative pursuant to this Section 9.5 shall be effective until the notice contemplated by clause (a) has been delivered and the accession contemplated by clause (b) shall have occurred.

ARTICLE X

MISCELLANEOUS AND GENERAL

- 10.1 <u>Modification or Amendment; Waiver</u>. Subject to Section 9.5, any provision of this Agreement may be amended, modified, supplemented or waived only by an instrument in writing duly executed by each Party. Any such amendment, modification, supplement or waiver shall be for such period and subject to such conditions as shall be specified in the instrument effecting the same and shall be binding upon the Parties, and any such waiver shall be effective only in the specific instance and for the purposes for which given. The rights and remedies herein provided shall be cumulative and not exclusive of any rights or remedies provided by Law except as provided in Article VIII.
- 10.2 <u>Expenses</u>. Except as otherwise provided in this Agreement and whether or not the Transactions are consummated, all costs and expenses (including fees and expenses of counsel and financial advisors, if any) incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the Party incurring such costs and expenses.

- 10.3 <u>Counterparts.</u> This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, and all of which shall constitute one and the same Agreement. Facsimile signatures or signatures received as an attachment to electronic mail (including .pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000 (e.g. www.docusign.com)) shall be treated as original signatures for all purposes of this Agreement.
- 10.4 <u>Governing Law and Venue; Submission to Jurisdiction; Selection of Forum; Waiver of Trial by Jury; Specific Performance.</u>
- (a) This Agreement, and all Proceedings (whether in contract, tort or statute) that may be based upon, arise out of or relate to this Agreement, or the negotiation, execution or performance of this Agreement (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement), shall be governed by, and enforced in accordance with, the Laws of the State of Delaware, including its statutes of limitations, without giving effect to applicable principles of conflicts of law to the extent that the application of the laws of another jurisdiction (whether of the State of Delaware or any other jurisdiction) would be required thereby.
- (b) Each Party agrees that it shall bring any Proceeding in respect of any claim based upon, arising out of or otherwise relating to this Agreement or the transactions contemplated by this Agreement exclusively in the Court of Chancery of the State of Delaware, or if such court finds it lacks subject matter jurisdiction, the Superior Court of the State of Delaware (Complex Commercial Division); provided that if subject matter jurisdiction over the matter that is the subject of the applicable Proceeding is vested exclusively in the U.S. federal courts, such Proceeding shall be heard in the U.S. District Court for the District of Delaware (the "Chosen Courts") and solely in connection with such Proceeding (i) irrevocably submits to the exclusive jurisdiction of the Chosen Courts, (ii) waives any objection to the laying of venue in any such Proceeding in the Chosen Courts, (iii) waives any objection that the Chosen Courts are an inconvenient forum or do not have jurisdiction over any Party hereto and (iv) agrees that mailing of process or other papers in connection with any such Proceeding in the manner provided in Section 9.1(b) or Section 10.5 (or in such other manner as may be permitted by Law shall be valid and sufficient service thereof).
- (c) EACH PARTY HERETO IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY LAW ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY WITH RESPECT TO ANY PROCEEDING BASED UPON, ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY. EACH PARTY HEREBY ACKNOWLEDGES AND CERTIFIES (I) THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF THE OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF ANY PROCEEDING, SEEK TO ENFORCE THE FOREGOING WAIVER, (II) IT UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (III) IT MAKES THIS WAIVER VOLUNTARILY AND (IV) IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS CONTAINED IN THIS SECTION 10.4(c).
- (d) The Parties agree that if any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached, irreparable damage would occur and money damages will not be a sufficient remedy. Accordingly, each Party shall be entitled to specific performance of the terms of this Agreement and immediate injunctive relief, without the necessity of proving the inadequacy of money damages as a remedy, in addition to any other remedy to which such Party may be entitled at Law or in equity.

Notices. All notices and other communications to be given or made hereunder shall be in writing and shall be deemed to have been duly given or made on the date of delivery to the recipient thereof if received prior to 5:00 P.M. Eastern Time in the place of delivery and such day is a Business Day (or otherwise on the next succeeding Business Day) if (a) served by personal delivery or by an internationally recognized overnight courier to the Person or entity for whom it is intended, (b) delivered by registered or certified mail, return receipt requested, or (c) sent by email, as provided in this Section 10.5; provided that the email is confirmed in writing by the recipient thereof (excluding out of office replies or other automatically generated responses) or is followed up within one Business Day after email by dispatch pursuant to one of the other methods described herein:

To Parent, DK HoldCo, Merger Sub I or Merger Sub II:

DraftKings Inc. 222 Berkeley Street, 5th Floor Boston, MA 02116

Email: sdodge@draftkings.com

Attn: Stanton Dodge, Chief Legal Officer & Secretary

With a copy to:

Sullivan & Cromwell LLP 125 Broad Street New York, NY 10004

Email: millersc@sullcrom.com; parnesl@sullcrom.com

Attn: Scott D. Miller and Lee C. Parnes

To the Company:

Jackpocket Inc. 145 West 45th Street, Floor 12 New York, NY 10036 Email: [Redacted] Attn: Carter Vance, General Counsel

With a copy to:

Cooley LLP 55 Hudson Yards

New York, NY 10001

Email: kcooper@cooley.com; jlaretto@cooley.com Attn: Kevin Cooper and Jeff Laretto

To Stockholder Representative:

Shareholder Representative Services LLC 950 17th Street, Suite 1400 Denver, CO 80202 Telephone: (303) 648-4085

Email: deals@srsacquiom.com Attn: Managing Director

With a copy to:

Cooley LLP 55 Hudson Yards

New York, NY 10001

Email: kcooper@cooley.com; jlaretto@cooley.com

Attn: Kevin Cooper and Jeff Laretto

or to such other Person or addressees as may be designated in writing by the Party to receive such notice as provided above; <u>provided</u>, <u>however</u>, that copies shall be provided to outside counsel for convenience only and that such copies shall not, in and of themselves, constitute notice, and the failure to provide any such copy shall not alter the effectiveness of any notice or other communication otherwise duly made or given.

- 10.6 <u>Entire Agreement</u>. This Agreement (including any exhibits or schedules hereto, including the Company Disclosure Letter and the Parent Disclosure Letter) and the Confidentiality Agreement constitute the entire agreement and supersede all other prior agreements, understandings, representations and warranties both written and oral, among the Parties, with respect to the subject matter hereof.
- 10.7 <u>No Third Party Beneficiaries</u>. Except as provided in Section 5.12 (Indemnification; *Directors' and Officers' Insurance*), Article VIII (Indemnification) and Article IX (Stockholder Representative) only, there shall be no third-party beneficiaries of this Agreement, or any exhibit, annex or schedule hereto or thereto, and none of them shall confer on any Person other than the parties hereto and thereto any claim, cause of action, right or remedy.
- 10.8 Obligations of Parent and of the Company. Whenever this Agreement requires a Subsidiary of Parent to take any action, such requirement shall be deemed to include an undertaking on the part of Parent to cause such Subsidiary to take such action. Whenever this Agreement requires a Subsidiary of the Company to take any action, such requirement shall be deemed to include an undertaking on the part of the Company to cause such Subsidiary to take such action.
- 10.9 <u>Severability</u>. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction or other authority of competent jurisdiction to be invalid, void or unenforceable, or the application of such provision, covenant or restriction to any Person or any circumstance, is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision and (b) the remainder of this Agreement and the application of such provision, covenant or restriction to other Persons or circumstances shall not be affected by such invalidity or unenforceability, nor shall such invalidity or unenforceability affect the validity or enforceability of such provision, or the application of such provision, in any other jurisdiction and the remainder of the terms, provisions, covenants and restrictions

of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

10.10 Interpretation; Construction.

- (a) The table of contents and headings herein are for convenience of reference only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions hereof. Where a reference in this Agreement is made to an Annex, Exhibit, Section or Schedule, such reference shall be to an Annex, Exhibit, Section or Schedule to this Agreement unless otherwise indicated.
- If a term is defined as one part of speech (such as a noun), it shall have a (b) corresponding meaning when used as another part of speech (such as a verb). The terms defined in the singular have a comparable meaning when used in the plural and vice versa. The rule known as the ejusdem generis rule shall not apply, and accordingly, general words introduced by the word "other" shall not be given a restrictive meaning by reason of the fact that they are preceded by words indicating a particular class of acts, matters or things. Unless the context of this Agreement clearly requires otherwise, words importing the masculine, female or neutral genders shall include the other genders. Whenever the words "include", "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation." The words "hereof," "herein" and "hereunder" and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement. The word "or" shall not be exclusive. The word "extent" in the phrase "to the extent" means the degree to which a subject or thing extends, and such phrase shall not mean simply "if". Currency amounts referenced herein are in U.S. Dollars. Any capitalized term used in any Schedule or Exhibit but not otherwise defined therein shall have the meaning given to them as set forth in this Agreement. All accounting terms used herein and not expressly defined herein shall have the meanings given to them under GAAP. References to "written" or "in writing" include documents in electronic form or transmission by email. References to "actual fraud" shall be deemed to exclude forward looking statements (as defined in 15 U.S.C. §78u-5) made in good faith.
- (c) Except as otherwise specifically provided herein, all references in this Agreement to any Law include the rules and regulations promulgated thereunder, in each case as amended, reenacted, consolidated or replaced from time to time and in the case of any such amendment, re-enactment, consolidation or replacement, reference herein to a particular provision shall be read as referring to such amended, re-enacted, consolidated or replaced provision and shall also include, unless the context otherwise requires, all applicable guidelines, bulletins or policies made in connection therewith; provided that for purposes of any representations and warranties contained in this Agreement that are made as of a specific date, references to any Law shall be deemed to refer to such Law as amended as of such date. Any agreement or instrument referred to herein means such agreement or instrument as from time to time amended, modified or supplemented, including by waiver or consent and all attachments thereto and instruments incorporated therein.
- (d) Whenever this Agreement refers to a number of days, such number shall refer to calendar days unless Business Days are specified. Whenever any action must be taken hereunder on or by a day that is not a Business Day, then such action may be validly taken on or by the next day that is a Business Day.
- (e) Each representation, warranty, covenant and condition herein shall be given full, separate and independent effect. The provisions hereof are cumulative. A more specific provision shall limit the applicability of any other, more general, provision.

- (f) The Parties drafted this Agreement jointly through the exchange of drafts hereof, so no presumption or burden of proof favoring or disfavoring any Party by virtue of the authorship of any provision of this Agreement.
- (g) Neither the specification of any dollar amount in any representation or warranty contained in this Agreement nor the inclusion of any specific item in any Schedule is intended to imply that such amount, or higher or lower amounts, or the item so included or other items, are or are not material, and no Party shall use the fact of setting forth of any such amount or the inclusion of any such item in any dispute or controversy between the Parties as to whether any obligation, item or matter not described herein or included in any Schedule is or is not material for purposes of this Agreement. Neither the specification of any item or matter in any representation or warranty contained in this Agreement nor the inclusion of any specific item in any Schedule is intended to imply that such item or matter, or other items or matters, are or are not in the Ordinary Course, and no Party shall use the fact of the setting forth or the inclusion of any specific item or matter in any dispute or controversy between the Parties as to whether any obligation, item or matter not described herein or included in any Schedule is or is not in the Ordinary Course for purposes of this Agreement.
- (h) The Company has set forth information in the Company Disclosure Letter in a Section of such Company Disclosure Letter that corresponds to the Section of this Agreement to which it relates. Parent has set forth information in the Parent Disclosure Letter in a section of such Parent Disclosure Letter that corresponds to the section of this Agreement to which it relates. The fact that any item of information is disclosed in any Section or subsection of the Company Disclosure Letter or the Parent Disclosure Letter shall be deemed disclosure with respect to any other Section or subsection of the Company Disclosure Letter and the Parent Disclosure Letter, respectively, to which the relevance of such item is apparent based on a plain reading of such disclosure.
- (i) All calculations involving the determination of dollar amounts pursuant to or in connection with this Agreement, the Escrow Agreement or the Payments Administrator Agreement shall be calculated to the nearest penny (two decimal places).
- 10.11 <u>Successors and Assigns</u>. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors, legal representatives and permitted assigns. No Party to this Agreement may assign any of its rights or delegate any of its obligations under this Agreement, by operation of Law or otherwise, without the prior written consent of the other Party; <u>provided</u> that each of Merger Sub I and Merger Sub II may assign any and all of its rights or obligations under this Agreement to one or more wholly owned Subsidiaries of Parent. Any purported assignment in violation of this Agreement is void.
- 10.12 <u>Fulfillment of Obligations</u>. Any obligation of any Party to any other Party under this Agreement, the Escrow Agreement or the Payments Administrator Agreement which obligation is performed, satisfied or fulfilled completely by an Affiliate of such Party, shall be deemed to have been performed, satisfied or fulfilled by such Party.
- 10.13 Waiver of Conflicts; Privilege. Parent, DK HoldCo, Merger Sub I, Merger Sub II and the Company agree that, as to all communications among Company's Counsel and the Stockholder Representative and the Company Stockholders and their respective Affiliates (individually and collectively, the "Seller Group") that relate to Company's Counsel's representation of the Company in the Transactions (but not representations prior to engagement for or separate from such transactions), the attorney-client privilege and the expectation of client confidence belongs solely to the Seller Group and may be controlled only by the Seller Group and will not pass to or be claimed, or deemed waived, by Parent, DK HoldCo, Merger Sub II, Merger Sub II, the Initial Surviving Company or the Subsequent

Surviving Company. This right to the attorney client privilege will exist even if such communications may exist on the Company's computer system or in documents in Company's possession. Notwithstanding the foregoing, in the event that a dispute arises between Parent and its Subsidiaries, and a Person other than the Stockholder Representative or a Company Stockholder after the Closing, Parent and its Subsidiaries, may assert (but not waive) the attorney-client privilege. If the Stockholder Representative so desires, acting on behalf of the Company Stockholders and without the need for any consent or waiver by the Company or Parent, Company's Counsel shall be permitted to represent the Stockholder Representative or any other member of the Seller Group after the Closing in connection with any matter, including without limitation, anything related to the Transactions, any other agreements referenced herein or any disagreement or dispute relating thereto. Without limiting the generality of the foregoing, after the Closing, Company's Counsel shall be permitted to represent the Stockholder Representative or any other member of the Seller Group, in connection with any negotiation, transaction or dispute (including any litigation, arbitration or other adversary Proceeding) with Parent, the Company or any of their agents or Affiliates under or relating to this Agreement, the Transactions, and any related matter, such as claims or disputes arising under other agreements entered into in connection with this Agreement. Upon and after the Closing, the Company shall cease to have any attorney-client relationship with Company's Counsel, unless and to the extent Company's Counsel is specifically engaged in writing by the Company to represent the Company after the Closing and either such engagement involves no conflict of interest with respect to the Company Stockholders or the Stockholder Representative consents in writing at the time to such engagement. Any such representation of the Company by Company's Counsel after the Closing shall not affect the foregoing provisions hereof.

[Signature Page Follows]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed as of the date first written above.

DRAFTKINGS INC.

By: /s/ Jason Robins

Name: Jason Robins

Title: Chief Executive Officer and Chairman

DRAFTKINGS HOLDINGS INC.

By: /s/ Paul Liberman

Name: Paul Liberman

Title: President and Chief Executive Officer

FORTUNE MERGER SUB INC.

By: /s/ Paul Liberman

Name: Paul Liberman

Title: President and Chief Executive Officer

FORTUNE MERGER SUB LLC

By: /s/ Paul Liberman

Name: Paul Liberman

Title: President and Chief Executive Officer

JACKPOCKET INC.

By: /s/ Peter J. Sullivan III

Name: Peter J. Sullivan III
Title: Chief Executive Officer

SHAREHOLDER REPRESENTATIVE SERVICES

LLC, as Stockholder Representative

By: /s/ Corey Quinlan

Name: Corey Quinlan Title: Director, Deal Intake

EXHIBIT A

DEFINITIONS

As used in this Agreement, the following terms have the meanings specified in this Exhibit A.

- "280G Stockholder Vote" has the meaning set forth in Section 5.11(e).
- "401(k) Plan" means the Jackpocket Inc. Retirement Plan.
- "Accountant" has the meaning set forth in Section 2.6(c)(ii).
- "Accredited Investor" means a Company Stockholder that (a)(i) is an "accredited investor" as defined in Rule 501(a) of Regulation D under the Securities Act; and (ii) delivers a duly completed and executed Accredited Investor Questionnaire or (b) is determined by Parent and the Company prior to the Closing in their reasonable discretion to be an "accredited investor" as defined in Rule 501(a) of Regulation D under the Securities Act.
 - "Accredited Investor Closing Cash Consideration" has the meaning set forth in Section 2.1(b).
 - "Accredited Investor Closing Stock Consideration" has the meaning set forth in Section 2.1(b).
- "Accredited Investor Company Capital Stock Closing Consideration" has the meaning set forth in Section 2.1(b).
 - "Accredited Investor Questionnaire" has the meaning set forth in Section 5.7.
- "Acquisition Proposal" means (i) any proposal or offer with respect to a merger, joint venture, partnership, consolidation, dissolution, liquidation, tender offer, recapitalization, reorganization, share exchange, business combination or similar transaction involving the Company pursuant to which the Company Stockholders immediately preceding such transaction would hold, directly or indirectly, less than eighty percent (80%) of the equity interests in the surviving or resulting entity of such transaction and (ii) any acquisition by any Person resulting in, or proposal or offer, which if consummated would result in, any Person becoming the beneficial owner of, directly or indirectly, in one or a series of related transactions, twenty percent (20%) or more of the total voting power or of any class of equity securities of the Company, or twenty percent (20%) or more of the consolidated total assets of the Company, in each case other than the transactions contemplated by this Agreement.
- "Affiliate" means, with respect to any Person, any other Person that directly or indirectly controls, is controlled by, or is under common control with, such Person. As used in this definition, the term "controls" (including the terms "controlled by" and "under common control with") means possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through ownership of voting securities, by Contract or otherwise. For the avoidance of doubt, (i) prior to the Closing, neither the Company nor any of its Subsidiaries shall be an "Affiliate" of Parent, and (ii) following the Closing, neither the Company nor any of its Subsidiaries shall be an "Affiliate" of any Company Stockholder.
 - "Affiliate Contract" has the meaning set forth in Section 3.13(a)(xviii).
- "Aggregate Cash Consideration" means an amount equal to: (i) \$412,500,000 plus (ii) the Aggregate Company Option Exercise Price plus (iii) Cash on Hand minus (iv) Closing Indebtedness plus

- (v) the Net Working Capital Difference minus (vi) Company Transaction Expenses minus (vii) the Extension Payment Amount.
- "Aggregate Company Option Exercise Price" means the aggregate exercise price payable by all holders of Company Options with respect to the Company Options outstanding immediately prior to the Initial Effective Time, assuming the cancellation of all Company Options cancelled as of immediately prior to the Initial Effective Time pursuant to this Agreement or action of the Company Board.
- "Aggregate Purchase Price" means, collectively, the Aggregate Cash Consideration and the Aggregate Stock Consideration.

"Aggregate Stock Consideration" means the following:

- (A) if the Closing Parent Trading Price is greater than \$42.8582, then the Aggregate Stock Consideration shall be 7,874,805.754791 shares of Parent Common Stock;
- (B) if the Closing Parent Trading Price is greater than or equal to \$31.6778, but less than or equal to \$42.8582, then the Aggregate Stock Consideration shall be a number of shares of Parent Common Stock equal to the quotient obtained by dividing (x) \$337,500,000 by (y) the Closing Parent Trading Price;
- (C) if the Closing Parent Trading Price is less than \$31.6778, then the Aggregate Stock Consideration shall be 10,654,148.962365 shares of Parent Common Stock.
- "Agreement" has the meaning set forth in the Preamble.
- "Alternative Acquisition Agreement" has the meaning set forth in Section 5.4(a)(iv).
- "Antitrust Law" means any Law designed or intended to prohibit, restrict, or regulate actions having the purpose or effect of monopolization or restraint of trade or lessening competition through merger or acquisition, including the Sherman Act of 1890, as amended; the Clayton Act of 1914, as amended; the HSR Act; the Federal Trade Commission Act of 1914, as amended; and similar international, multilateral, multinational, national, federal or state competition Laws.
 - "Applicable Date" has the meaning set forth in Section 3.5(e).
- "Applicable Gaming Law" means all applicable Laws (including any statute, regulation, rule, requirement, standard, guidance, Order, announcement or notice of any Gaming Regulatory Authority) which are applicable to the Company, its Subsidiaries and/or their respective businesses.
 - "Bankruptcy and Equity Exception" has the meaning set forth in Section 3.3(a).
- "Benefit Plan" means any benefit or compensation plan, program, policy, practice, agreement, Contract, arrangement or other obligation, whether or not in writing and whether or not funded, in each case, that is sponsored or maintained by, or required to be contributed to, or with respect to which any potential liability is borne by the Company or any of its Subsidiaries. Benefit Plans include, but are not limited to, "employee benefit plans" within the meaning of Section 3(3) of ERISA ("ERISA Plans"), employment, non-compete and/or non-solicit, consulting, retirement, severance, termination or change in control agreements, deferred compensation, equity-based, incentive, bonus, supplemental retirement, profit sharing, insurance, medical, welfare, material fringe or other benefits or remuneration of any kind.

- "Board Recommendation" has the meaning set forth in Section 3.3(b).
- "Burdensome Condition" has the meaning set forth in Section 5.3(b).
- "Business Day" means any day other than (a) a Saturday or a Sunday or (b) a day on which banking and savings and loan institutions are authorized or required by Law to be closed in New York, New York.
 - "Capitalization Time" has the meaning set forth in Section 3.2(a).
- "CARES Act" means The Coronavirus AID, Relief, and Economic Security Act, Pub. L. 116-136 (03/27/2020).
- "Cash on Hand" means all cash and cash equivalents held by the Company and its Subsidiaries as of 12:01 A.M. Eastern Time on the Closing Date calculated in accordance with GAAP (including any checks or pending electronic debits received by the Company, its Subsidiaries or their respective banks prior to such time, whether or not cleared, and any incoming wire transfers to the extent the respective amounts of such checks, electronic debits and wire transfers are not included in the calculation of Net Working Capital); provided that Cash on Hand shall (i) be calculated net of outgoing wire transfers and drafts written or issued by the Company or any of its Subsidiaries, (ii) include (a) security deposits in respect of Leased Real Property (up to an aggregate amount of \$500,000) and (b) cash and cash equivalents held at MVB Bank, Inc. in connection with the MVB Loan Documents and amounts deposited in the Company's interactive gaming FBO account at MVB Bank Inc., in each case, to the extent that all Indebtedness under the MVB Loan Documents is repaid and all Liens in connection therewith are released, in each case, as of or promptly following the Closing, and such cash and cash equivalents are otherwise freely usable immediately following the Closing, (iii) subject to the foregoing clause (ii), exclude any cash, cash equivalents or marketable securities that are not freely usable, distributable or transferable (including because it is subject to restrictions, limitations, penalties or Taxes on use, distribution or transfer by Law, Contract or otherwise), restricted balances (including security deposits, bond guarantees, collateral reserve amounts, amounts held in escrow for the benefit of third parties (including user funds held by the Company or any of its Subsidiaries) and escheat, unclaimed property, (iv) be reduced by the amount of any insurance proceeds received, including in connection with the damage or destruction of any asset, except to the extent such insurance proceeds have been used to replace any such asset, and (v) be reduced by the amount of any cash to be distributed pursuant to any dividend or share repurchase declared, but not yet paid, prior to 12:01 A.M. Eastern Time on the Closing Date.
 - "Certificate" has the meaning set forth in Section 2.1(b).
 - "Certificate of Incorporation" has the meaning set forth in Section 3.1.
 - "Certificate of Initial Merger" has the meaning set forth in Section 1.4.
 - "Certificate of Subsequent Merger" has the meaning set forth in Section 1.4.
 - "Change of Recommendation" has the meaning set forth in Section 5.4(b).
 - "Chosen Courts" has the meaning set forth in Section 10.4(b).
 - "Claim Notice" has the meaning set forth in Section 8.3(a)(i).

- "Class A Stock" has the meaning set forth in the Recitals.
- "Class B Stock" has the meaning set forth in the Recitals.
- "Class B-1 Stock" has the meaning set forth in the Recitals.
- "Class C-1 Stock" has the meaning set forth in the Recitals.
- "Class C-2 Stock" has the meaning set forth in the Recitals.
- "Class C-3 Stock" has the meaning set forth in the Recitals.
- "Class D-1 Stock" has the meaning set forth in the Recitals.
- "Class D-2 Stock" has the meaning set forth in the Recitals.
- "Class D-3 Stock" has the meaning set forth in the Recitals.
- "Clean Team Confidentiality Agreement" means that certain Clean Team Confidentiality Agreement, dated as of December 13, 2023, by and between the Company and DK HoldCo.
 - "Closing" has the meaning set forth in Section 1.3.
 - "Closing Date" has the meaning set forth in Section 1.3
- "Closing Indebtedness" means the Indebtedness of the Company as of immediately prior to the Closing, which includes, for the avoidance of doubt, Indebtedness under the MVB Loan Documents. For the avoidance of doubt, Closing Indebtedness shall not include any Company Transaction Expenses or any amounts included in Net Working Capital.
- "Closing Parent Trading Price" means the volume-weighted average of the closing prices per share of Parent Common Stock on the NASDAQ (as reported in the New York City edition of The Wall Street Journal or, if not reported thereby, as reported in another authoritative source mutually agreed by the Company and Parent) for the thirty (30) consecutive full trading days ending on second trading day immediately preceding the Closing Date. For all purposes of this Agreement, the Closing Parent Trading Price shall be calculated to the nearest one-hundredth of one cent.
 - "Closing Payments Fund" has the meaning set forth in Section 2.5(b).
 - "Code" has the meaning set forth in the Recitals.
 - "Common Stock" has the meaning set forth in the Recitals.
 - "Company" has the meaning set forth in the Preamble.
 - "Company Board" has the meaning set forth in the Recitals.
 - "Company Capital Stock" has the meaning set forth in the Recitals.
- "Company Capital Stock Closing Consideration" has the meaning set forth in Section 2.1(b)(ii).

- "Company's Counsel" has the meaning set forth in Section 5.14(i).
- "Company Disclosure Letter" means the disclosure letter, dated as of the date hereof, that has been prepared by the Company and delivered to Parent, DK HoldCo, Merger Sub I and Merger Sub II with this Agreement.
- "Company Fundamental Representations" means the representations and warranties of the Company set forth in Section 3.1 (Organization, Good Standing and Qualification), Section 3.2 (Capital Structure), Section 3.3 (Authority; Approval), Section 3.12 (Takeover Statutes), Section 3.21 (Related Party Transactions) and Section 3.23 (Brokers and Finders).
- "Company Gaming Licenses" means any License issued or granted to the Company or any of its Subsidiaries by a Gaming Regulatory Authority.
- "Company IP" means all Intellectual Property Rights owned or purported to be owned by the Company or any of its Subsidiaries.
 - "Company Option" has the meaning set forth in Section 2.10(a).
 - "Company Payment Trigger" has the meaning set forth in Section 7.7(a).
- "Company Performance Option" means the Company Options listed on Section 2.10(b) of the Company Disclosure Letter.
- "Company Preferred Stock" means, collectively, the Seed Stock, Class A Stock, Class B Stock, Class B-1 Stock, Class C-1 Stock, Class C-2 Stock, Class C-3 Stock, Class D-1 Stock, Class D-2 Stock and the Class D-3 Stock."
- "Company Securityholders" means the Company Stockholders and the holders of In-The-Money Options.
 - "Company Stockholders" has the meaning set forth in the Recitals.
- "Company Tax Knowledge Party" means Peter Sullivan or Sean Siuda or, in the event that none of the foregoing individuals are reasonably available, any other authorized officer of the Company.
 - "Company Termination Fee" has the meaning set forth in Section 7.7(a).
 - "Company Top Gaming Authority" has the meaning set forth in Section 3.20(b).
 - "Company Top Supplier" has the meaning set forth in Section 3.20(a).
- "Company Transaction Expenses" means, without duplication, to the extent not paid or cause to be paid by the Company immediately prior to the Closing (including those which become due and payable after the Closing Date pursuant to Contracts entered into prior to the Closing Date), the aggregate amount of (a) all fees and expenses payable by the Company or any of its Subsidiaries to any attorneys, accountants, financial advisors, consultants or other advisors in connection with the Transactions or any sales process relating to a strategic transaction involving the Company or its Subsidiaries, (b) all bonuses, change-of-control, success, severance, retention, sale, stay or similar type of bonus, compensation or incentive payments or other similar payments pursuant to arrangements made by the Company or its Subsidiaries prior to the Closing that become due or payable to any current or former employees,

directors, officers or other service providers of the Company or any of its Subsidiaries in connection with the Transactions (and the employer portion of any payroll, employment or similar Taxes related thereto (other than Excluded Payroll Taxes)), including all payments in respect of the "Specified Options" listed on Section 3.2 of the Company Disclosure Letter; (c) any fees, costs, expenses or other liabilities incurred (or that would be incurred or made) as a result of the settlement of any Contract or transaction with a Related Party, (d) the cost of the D&O Tail Policy; and (e) 50% of all Transfer Taxes.

"Company Transaction Expenses Paid by Parent" means the Company Transaction Expenses that are to be paid by Parent on behalf of the Company as set forth on the Funds Flow Memorandum (which, for the avoidance of doubt, constitute Company Transaction Expenses).

"Company Warrants" means all outstanding warrants to purchase shares of Company Capital Stock.

"Confidential Information" means confidential information in any form regarding the Company or any of its Subsidiaries or disclosed in connection with the Transactions. Confidential Information shall not include any information that (a) at the time of disclosure or thereafter is published or otherwise generally available to the industry that the disclosing party primarily participates or to the public, in each case, other than by breach of this Agreement or the Confidentiality Agreement; (b) is rightfully received by either party or their respective Representatives from a third party to the receiving party's knowledge after reasonable inquiry without confidentiality limitations, prior to disclosure by the disclosing party; (c) was or is independently developed by the receiving party or its respective Representatives without the use of any Confidential Information provided by the disclosing party; (d) is approved for public release by written authorization of disclosing party; or (e) the Parties agree in writing to release from the terms of the Confidentiality Agreement.

"Confidentiality Agreement" has the meaning set forth in Section 5.13.

"Continuing Employees" has the meaning set forth in Section 5.11(a).

"Contract" means any agreement, undertaking, lease, license, contract, note, mortgage, indenture, arrangement or other similar obligation, whether written or oral.

"Controlling Party" has the meaning set forth in Section 8.3(b)(ii).

"COVID-19" means the COVID-19 or SARS-CoV-2 virus (or any mutation or variation thereof or related health condition).

"COVID-19 Measures" means any quarantine, "shelter in place," "stay at home," social distancing, shut down, closure or sequester order, guideline, recommendation or Law, or any other applicable Laws, guidelines or recommendations by any Governmental Entity in connection with or in response to COVID-19.

"D&O Tail Policy" has the meaning set forth in Section 5.12.

"<u>Data Room</u>" means the Company's "Project Fortune" electronic data room hosted by Intralinks and to which Parent was provided access at all times subject to the terms of the Clean Team Agreement prior to the date hereof in connection with the Transactions.

"Debt Payoff Amount" has the meaning set forth in Section 5.20.

- "Debt Payoff Letters" has the meaning set forth in Section 5.20.
- "Deductible" has the meaning set forth in Section 8.2(b).
- "Designated Optionholder" means Lior Shemesh.
- "Designated Stockholder" means Peter Sullivan.
- "DGCL" has the meaning set forth in the Recitals.
- "<u>Disability</u>" means a physical or mental impairment which totally and permanently prevents the individual from performing his or her customary and usual duties for Parent (or any Subsidiary thereof).
 - "Disputed Items" has the meaning set forth in Section 2.6(c)(ii).
 - "Dissenting Shares" has the meaning set forth in the definition of "Dissenting Stockholders."
- "Dissenting Stockholders" means the holders of Company Capital Stock who have duly demanded appraisal pursuant to Section 262 of the DGCL and have not effectively withdrawn or otherwise waived or lost such right to appraisal under Section 262 of the DGCL (such Company Capital Stock for which appraisal has been so duly demanded and the right thereto under Section 262 of the DGCL not effectively withdrawn or otherwise waived or lost, the "Dissenting Shares").
 - "DK Deal Announcement" has the meaning set forth in Section 5.5.
 - "DK Parties" means Parent, DK HoldCo, Merger Sub I and Merger Sub II.
 - "DLLCA" has the meaning set forth in the Recitals.
- "Employee" means any current or former employee (whether full- or part-time), director, officer or independent contractor (who is a natural person) of the Company or any of its Subsidiaries.
- "Environmental Law" means any Law relating to: (a) the protection, investigation or restoration of the environment, health, safety or natural resources, (b) the handling, use, presence, disposal, release or threatened release of, or exposure to, any Hazardous Substance or (c) noise, odor, indoor air, employee exposure, wetlands, pollution, contamination or any injury or threat of injury to Persons or property relating to any Hazardous Substance.
- "Equity Award Conversion Amount" means (x) Non-Accredited Investor Company Capital Stock Closing Consideration divided by (y) the Closing Parent Trading Price.
 - "ERISA" means the Employee Retirement Income Security Act of 1974.
- "ERISA Affiliate" means all employers (whether or not incorporated) that would be treated together with the Company or any of its Subsidiaries as a "single employer" within the meaning of Section 414 of the Code.
 - "ERISA Plans" has the meaning set forth in the definition of "Benefit Plan."
 - "Escrow Agent" has the meaning set forth in Section 2.4(a).

"Escrow Agreement" means an agreement between the Escrow Agent, Parent, and the Stockholder Representative in form and substance reasonably acceptable to each of Parent and the Stockholder Representative pursuant to which the Escrow Agent will hold, safeguard and release the Purchase Price Adjustment Holdback, the Retention Holdback and the Reserve Holdback in accordance with the terms of this Agreement.

"Estimated Closing Statement" has the meaning set forth in Section 2.3(a)(i).

"<u>Estimated Merger Cash Consideration</u>" means an amount in cash equal to the Merger Cash Consideration, as estimated in the Estimated Closing Statement.

"Exchange Act" means the Securities Exchange Act of 1934.

"Excluded Payroll Taxes" means (i) the employer portion of any employment or payroll Taxes with respect to any Company Transaction Expenses or any other compensatory payments made in connection with the Mergers and the transactions contemplated by this Agreement, in each case, to the extent such payments are not paid prior to, at or substantially contemporaneously with the Closing and (ii) any such employment or payroll Taxes incurred as a result of the release of the Purchase Price Adjustment Holdback and the Retention Holdback or any retention payments payable or equity awards granted by Parent or any of its Affiliates pursuant to the terms of any post-Closing employment arrangement with Parent or any of its Affiliates.

"Excluded Shares" means (i) shares of Company Capital Stock owned by Parent, DK HoldCo, Merger Sub I, Merger Sub II or any other direct or indirect wholly-owned Subsidiary of Parent and shares of Company Capital Stock owned by the Company or any direct or indirect wholly-owned Subsidiary of the Company, and in each case not held on behalf of third parties, and (ii) Dissenting Shares.

"Extension Payment Amount" means the payment of \$5,000,000 by or on behalf of Parent to the Company pursuant to that certain letter agreement, dated as of February 6, 2024, by and between Parent and the Company.

"FCPA" means the United States Foreign Corrupt Practices Act of 1977.

"<u>Final Aggregate Cash Consideration</u>" means the Aggregate Cash Consideration as set forth on the Final Closing Statement.

"<u>Final Closing Statement</u>" means the Post-Closing Statement that (a) is deemed final in accordance with Section 2.6(c)(i) or (b) results from the determinations made by the Accountant in accordance with Section 2.6(c)(iv), as applicable.

"<u>Final Determination</u>" means, with respect to a dispute, an occurrence where (a) the parties to the dispute have reached an agreement in writing, (b) a court of competent jurisdiction shall have entered a final and non-appealable Order or judgment with respect to a claim or (c) an arbitration or like panel shall have rendered a final non-appealable determination with respect to disputes the parties have agreed to submit thereto.

"<u>Final Merger Cash Consideration</u>" means the Merger Cash Consideration as set forth on the Final Closing Statement.

"Financial Statements" has the meaning set forth in Section 3.5(a).

"First Release Date" has the meaning set forth in Section 2.4(b).

"Fully Diluted Non-Option Shares" means the aggregate number of shares of Company Capital Stock (other than Excluded Shares but including Dissenting Shares) issued and outstanding as of immediately prior to the Initial Effective Time, assuming the conversion and exercise of all outstanding securities convertible into and exercisable for, or other rights to acquire, Company Capital Stock (in each case, whether or not immediately exercisable, and determined on an as-converted to Common Stock basis), in each case, excluding the aggregate number of shares of Company Capital Stock issuable upon the exercise of the Company Options outstanding (whether or not immediately exercisable) immediately prior to the Initial Effective Time.

"Fully Diluted Pro Rata Share" means, with respect to a Company Securityholder, the percentage equal to the quotient of (a) the sum of (i) total number of shares of Company Capital Stock held by such Company Securityholder immediately prior to the Initial Effective Time (as described opposite such Company Securityholder's name in the Funds Flow Memorandum) and (ii) the total number of shares of Company Capital Stock issuable to such Company Securityholder upon exercise of Company Options held by such Company Securityholder immediately prior to the Initial Effective Time (as described opposite such Company Securityholder's name in the Funds Flow Memorandum) divided by (b) the total number of Fully Diluted Shares.

"<u>Fully Diluted Shares</u>" means the aggregate number of shares of Company Capital Stock equal to the aggregate number of shares of Company Capital Stock (other than Excluded Shares but including Dissenting Shares) issued and outstanding as of immediately prior to the Initial Effective Time, assuming the conversion and exercise of all outstanding securities convertible into and exercisable for, or other rights to acquire, Company Capital Stock (in each case, whether or not immediately exercisable, and determined on an as-converted to Common Stock basis) and the cancellation of all such securities cancelled as of immediately prior to the Initial Effective Time pursuant to this Agreement or action of the Company Board.

"Funds Flow Memorandum" means a memorandum, which shall be certified as complete and correct by the Chief Executive Officer of the Company on behalf of the Company (solely in his capacity as such and not in his personal capacity, and without personal liability), and which shall be in a form to be reasonably agreed upon by Parent and the Company, which shall include the following information:

- (a) a complete and correct list as of immediately prior to the Initial Effective Time of (i) the names of each of Company Stockholder, (ii) whether such Company Stockholder is an Accredited Investor or Non-Accredited Investor, (iii) the number and class or series of shares of Company Capital Stock held by such Company Stockholder and the certificate number of such Certificate held by such Company Stockholder, (iv) the number of Company Options held by the holders thereof, (v) the Pro Rata Share of each Company Stockholder and (vi) the Fully Diluted Pro Rata Share for each Company Securityholder;
- (b) a true and correct calculation of the respective amounts of the Aggregate Purchase Price to be paid in accordance with this Agreement to each Company Securityholder;
- (c) the institutions and account information with respect to the payments set forth in Section 2.4 and Section 2.5(a); and
- (d) each vendor and payee (with wire instructions or payment information) for Company Transaction Expenses Paid by Parent.

- "GAAP" means United States generally accepted accounting principles.
- "Gaming Regulatory Authority" means any Governmental Entity in any jurisdiction with regulatory control, authority or jurisdiction over gambling, online gambling, betting, lottery, sweepstakes, bingo, fantasy contests, lottery courier and gaming activities (if any), including, for the avoidance of doubt, the Governmental Entities issuing the Company Gaming Licenses.
- "Governmental Entity" means any United States, non-United States, supranational or transnational governmental (including public international organizations), quasi-governmental, regulatory or self-regulatory authority, agency, commission, body, department or instrumentality or any court, tribunal or arbitrator or other entity or subdivision thereof or other legislative, executive or judicial entity or subdivision thereof, in each case, of competent jurisdiction (including, for the avoidance of doubt, the Gaming Regulatory Authorities).
- "<u>Hazardous Substance</u>" means (a) any substance that is listed, classified or regulated pursuant to any Environmental Law, (b) any petroleum product or by-product, asbestos-containing material, lead-containing paint or plumbing, polychlorinated biphenyls, mold, radioactive material or radon and (c) any other substance that may be the subject of regulatory action by any Governmental Entity in connection with any Environmental Law.
 - "HSR Act" means the Hart-Scott-Rodino Antitrust Improvement Act of 1976.
 - "In-the-Money Option" has the meaning set forth in Section 2.10(a).
 - "Incremental Hire Salary Cap" has the meaning set forth in Section 5.1(b)(xvi)(E).
 - "Indebtedness" means the following, without duplication:
 - (A) borrowed money of the Company and its Subsidiaries, including borrowings from any bank, financial institution or other entity, including loans and bank overdrafts;
 - (B) liabilities under any loan note, loan stock, debenture, letters of credit or other similar instrument or security including, for the avoidance of doubt, the MVB Loan Documents;
 - (C) obligations for the reimbursement of any obligor on any drawn-upon letter of credit (to the extent drawn), banker's acceptance or similar transaction;
 - (D) liabilities under a finance or capitalized lease or hire purchase agreements or any analogous agreement or arrangement creating obligations with respect to the deferred purchase price of property, but excluding any liabilities under operating leases;
 - (E) liabilities under any currency or interest swap or other interest or currency protection, hedging or financial futures transaction or arrangement;
 - (F) deferred purchase price of assets, businesses, properties or services, including all seller notes and "earn out" payments (which, in each case, shall be calculated assuming the maximum achievement of performance targets);
 - (G) self-insurance accruals;

- (H) the Unpaid Pre-Closing Income Tax Amount and all unpaid Taxes which would otherwise have been required to be paid in a Pre-Closing Tax Period but the payment of which was deferred pursuant to the CARES Act or any similar provision of state, local or non-U.S. Tax law,
- (I) all accrued and unpaid deferred compensation, severance and bonus obligations (prorated by the number of days that have elapsed through the Closing Date for the year in which the Closing occurs) as of the Closing (including the employer portion of any payroll, employment and similar Taxes associated with such obligations (other than Excluded Payroll Taxes));
- (J) all unfunded or underfunded liabilities and/or payment obligations with respect to any defined benefit pension plans that Employees of the Company or any of its Subsidiaries participate in;
- (K) all liabilities for accrued or unpaid time off with respect to Employees (including the employer portion of any payroll, employment and similar Taxes associated with such obligations (other than Excluded Payroll Taxes));
- (L) all obligations of a type referred to in clauses (A) through (K) above which the Company or any of its Subsidiaries (1) has guaranteed, (2) is responsible or liable, directly or indirectly, as obligor or guarantor or (3) has agreed (contingently or otherwise) to purchase or otherwise acquire or in respect of which it has otherwise assured a credit against loss;
- (M) accrued interest, principal, charges, costs, prepayment penalty, fees, breakage costs or expenses (including arising as a result of the discharge of such amount at Closing or the consummation of the Transactions), to the extent due or owing in respect of those items listed in clauses (A) through (L) above, whether resulting from their payment or discharge or otherwise.

For the avoidance of doubt, "Indebtedness" shall not include any amount included in calculating Net Working Capital or Company Transaction Expenses.

"Indemnified Director or Officer" means each present and former (determined as of the Initial Effective Time) director and officer of the Company or any of its Subsidiaries, in each case, when acting in such capacity or in serving as a director, officer, member, trustee or fiduciary of another entity or enterprise, at the request or benefit of the Company or any of its Subsidiaries.

"Information Statement" has the meaning set forth in Section 5.7.

"Initial Effective Time" has the meaning set forth in Section 1.4.

"Initial Merger" has the meaning set forth in the Recitals.

"Initial Stockholder Consent Delivery Period" has the meaning set forth in Section 5.5.

"Initial Surviving Company" has the meaning set forth in the Recitals.

"Insurance Policies" has the meaning set forth in Section 3.19.

"Intellectual Property Rights" means all rights anywhere in the world in or to: (a) trademarks, service marks, brand names, certification marks, collective marks, d/b/a's, logos, symbols, trade dress, trade names, and other indicia of origin, all applications and registrations for the foregoing, and all

goodwill associated therewith and symbolized thereby, including all renewals of the same; (b) patents, patent applications, registrations and invention disclosures, including divisionals, revisions, supplementary protection certificates, continuations, continuations-in-part, renewals, extensions, substitutes, re-issues and re-examinations; (c) confidential or proprietary information, trade secrets, inventions, discoveries, ideas, improvements, information, know-how, technical data and technical databases, algorithms, techniques, processes, schematics, business and other methods, formulae, drawings, specifications, source code, prototypes, models, flow-charts, testing procedures and results, designs, customer lists and supplier lists (collectively, "Trade Secrets"); (d) published and unpublished works of authorship, whether copyrightable or not (including Software, website and mobile content, data, databases and other compilations of technical information), copyrights therein and thereto, and registrations and applications therefor, and all renewals, extensions, restorations and reversions thereof; rights to use any Person's name, image, likeness or other unequivocal identifiers, and related publicity rights, (e) Internet domain names and URLs; and (f) all other intellectual property, industrial or proprietary rights.

"Intended Tax Treatment" has the meaning set forth in the Recitals.

"IRS" means the United States Internal Revenue Service.

"IT Assets" means hardware, technology devices, computers, mobile devices, Software, servers, networks, workstations, routers, hubs, circuits, switches, data communications lines, and all other information technology equipment that are owned or used (including through cloud-based or other third-party service providers) by the Company or any of its Subsidiaries in the operation of the business as currently conducted, and all information or data stored therein or processed thereby, and all associated documentation.

"Key Employee" means each of the Employees set forth on Section 1.1 of the Company Disclosure Letter.

"Knowledge" means (i) with respect to the Company, any Key Employee who is actually aware of such fact or other matter or should have become aware of such fact or other matter after reasonable due inquiry, including reasonable due inquiry of his or her direct reports, and (ii) with respect to Parent, Jason Robins, Jason Park, or Malcolm Turner who is actually aware of such fact or other matter or should have become aware of such fact or other matter after reasonable due inquiry, including reasonable due inquiry of his or her direct reports.

"<u>Law</u>" or "<u>Laws</u>" means any federal, state, local, foreign, international or transnational law, statute, ordinance, common law, rule, regulation, standard, judgment, determination, Order, writ, injunction, decree, arbitration award, treaty, agency requirement, authorization, license or permit of any Governmental Entity.

"Lease" or "Leases" has the meaning set forth in Section 3.14(b).

"Leased Real Property" has the meaning set forth in Section 3.14(b).

"Letter of Transmittal" has the meaning set forth in Section 2.5(d).

"<u>Licenses</u>" means permits, licenses, memorandums of understanding, authorization to operate, certifications, approvals, registrations, consents, authorizations, franchises, variances, contractual authority, exemptions and orders issued or granted by a Governmental Entity.

"<u>Lien</u>" means any lien, charge, pledge, mortgage, easement, hypothecation, security interest, claim or other encumbrance.

"Loss" means any and all damages, losses, charges, liabilities, claims, demands, actions, suits, proceedings, payments, judgments, Orders, settlements, assessments, deficiencies, Taxes, interest, penalties and costs and expenses (including reasonable and documented out of pocket legal fees, costs and expenses).

"Malicious Code" means disabling codes or instructions, spyware, Trojan horses, worms, viruses or other software routines that facilitate or cause unauthorized access to, or disruption, impairment, disablement, encryption, unavailability or destruction of, Software, data or other materials.

"Material Adverse Effect" means, with respect to any Person, any change, development, event, circumstance, condition, fact, occurrence or effect that, individually or taken together with any and all other changes, developments, events, circumstances, conditions, facts, occurrences or effects, (i) would reasonably be expected to prevent, materially delay or materially impair the ability of such Person consummate the Transactions or (ii) has, or would reasonably be expected to have, a material adverse effect on the financial condition, business operations or results of operations of such Person and its Subsidiaries (taken as a whole); provided, however, that, in the case of the foregoing clause (ii) only, none of the following, either alone or in combination, shall be taken into account in determining whether a Material Adverse Effect is occurring, has occurred or would reasonably be expected to occur:

- (A) changes in the economy, capital, securities or financial markets or political, legislative, regulatory or business conditions generally;
- (B) to the extent resulting from the public announcement (in and of itself) of entry into this Agreement or public announcement (in and of itself) of the Transactions;
- (C) changes in GAAP or in any Law of general applicability following the date of this Agreement;
- (D) any failure in and of itself by such Person to meet any projections or forecasts or estimates of revenues or earnings; <u>provided</u> that the exception in this clause (D) shall not prevent or otherwise affect a determination that any effect underlying such failure has resulted in, or contributed to, or would reasonably be expected to result in, or contribute to, a Material Adverse Effect;
- (E) any change, development or effect resulting from acts of war (whether or not declared), any outbreak of hostilities, sabotage, terrorism, military actions or the escalation of any of the foregoing, cyberattacks impacting such Person, any weather or natural disaster, or any outbreak of illness, pandemic or epidemic (including the existence and impact of the COVID-19 Measures occurring prior to the date of this Agreement) or any other force majeure event;
- (F) any action or failure to take action which action or failure to act is expressly required by, or prohibited to be taken by, this Agreement (except any action expressly required to be taken or not taken pursuant to Section 5.1(a)); or
- (G) changes generally impacting conditions in the industries in which such Person and its Subsidiaries conduct business.

<u>provided</u>, <u>further</u>, that, with respect to clauses (A), (C), (E) and (G) of this definition, any such changes, developments, events, circumstances, conditions, facts, occurrences or effects shall be taken into account in determining whether a "Material Adverse Effect" has occurred, is occurring or would reasonably be expected to occur to the extent it disproportionately and adversely affects such Person or any of its Subsidiaries relative to other similarly situated companies operating in the industry or geographic markets in which such Person or any of its Subsidiaries operates.

- "Material Contract" has the meaning set forth in Section 3.13(a).
- "Mergers" has the meaning set forth in the Recitals.
- "Merger Cash Consideration" means (a) the Aggregate Cash Consideration, minus (b) the Retention Holdback, minus (c) Purchase Price Adjustment Holdback, minus (d) the Stockholder Representative Expense Amount, minus (e) the Reserve Holdback.
- "<u>Merger Stock Consideration</u>" means (a) the Aggregate Stock Consideration, minus (b) the aggregate number of shares of Parent Common Stock underlying the In-the-Money Options pursuant to Section 2.10.
 - "Merger Sub I" has the meaning set forth in the Preamble.
 - "Merger Sub II" has the meaning set forth in the Preamble.
 - "Most Recent Balance Sheet" has the meaning set forth in Section 3.5(a).
- "MVB Loan Documents" means each of (a) the Loan Agreement, dated as of April 28th, 2023, by and between MVB Bank, Inc. and the Company and (b) the Promissory Note, dated April 28, 2023, by and between the Company and MVB Bank, Inc.
 - "NASDAO" means The Nasdag Stock Market LLC.
- "Net Working Capital" means an amount equal to, as of 12:01 A.M. Eastern Time on the Closing Date, (a) the aggregate amount of the consolidated current assets of the Company and its Subsidiaries, minus (b) the aggregate amount of the consolidated current liabilities of the Company and its Subsidiaries; provided, however, that current assets and current liabilities shall exclude any (A) Closing Indebtedness, (B) Cash on Hand, (C) Company Transaction Expenses, (D) reserve in respect of the Reserved Matter and (E) assets or liabilities for income Taxes and any deferred Tax assets or deferred Tax liabilities; provided, further, that Net Working Capital shall be determined in accordance with (x) GAAP as applied by the Company in the Financial Statements to the extent such Financial Statements are prepared in accordance with GAAP on a consistent basis and (y) the principles specifically set forth in the illustrative calculation of Net Working Capital set forth in Exhibit F (provided that if there is a conflict between the foregoing and the principles specifically set forth in the illustrative calculation on Exhibit F, the latter shall control with respect to such conflict to the extent the applicable principle set forth in the illustrative calculation on Exhibit F is expressly identified as not being prepared in accordance with GAAP) and any Tax items included in Net Working Capital shall be determined in accordance with clauses (a), (b), (d), (e), and (f) of the definition of Unpaid Pre-Closing Income Tax Amount.
- "Net Working Capital Difference" means an amount equal to (i) Net Working Capital minus (ii) the Net Working Capital Target Amount.
 - "Net Working Capital Target Amount" means negative \$5,500,000.

- "Non-Accredited Investor" means each Company Stockholder who is not an Accredited Investor.
- "Non-Accredited Investor Closing Cash Consideration" has the meaning set forth in Section 2.1(b)(ii).
- "Non-Accredited Investor Closing Stock Consideration" has the meaning set forth in Section 2.1(b)(ii).
 - "Non-Controlling Party" has the meaning set forth in Section 8.3(b)(ii).
- "Non-Fundamental Company Representations" means any representation made by the Company in Article III that is not a Company Fundamental Representation.
 - "Notice of Objection" has the meaning set forth in Section 2.6(c)(i).
- "Notice Schedule" means a schedule, to be delivered to the Stockholder Representative by Parent or the Subsequent Surviving Company following the Closing, including the email address of each Company Stockholder as it may be updated from time to time.
 - "Offer Letter" has the meaning set forth in the recitals.
- "Order" means any administrative decision or award, decree, injunction, judgment, order, directive, quasi-judicial decision or award, ruling or writ of any arbitrator, mediator or Governmental Entity.
- "Ordinary Course" means the conduct of the business of the Company and its Subsidiaries in the ordinary course of business consistent with past practice.
 - "Organizational Documents" has the meaning set forth in Section 3.1.
- "Other Anti-Bribery Laws" means, other than the FCPA, all applicable anti-bribery, anti-corruption, anti-money-laundering and similar Laws in jurisdictions in which the Company or any of its Subsidiaries do business, have done business, in which any agent thereof is conducting or has conducted business involving the Company or any of its Subsidiaries or the Company or any of its Subsidiaries are otherwise subject.
 - "Option Acknowledgment Agreement" has the meaning set forth in the recitals.
 - "Outside Date" has the meaning set forth in Section 7.2(a).
 - "Parachute Payment" has the meaning set forth in Section 5.12(d).
 - "Parent 401(k) Plan" has the meaning set forth in Section 5.11(d).
 - "Parent" has the meaning set forth in the Preamble.
 - "Parent Class B Stock" has the meaning set forth in Section 4.12.
 - "Parent Common Stock" means Class A common stock, par value \$0.0001 per share, of Parent.

- "<u>Parent Disclosure Letter</u>" means the disclosure letter, dated as of the date hereof, that has been prepared by Parent, DK HoldCo, Merger Sub I and Merger Sub II and delivered to the Company with this Agreement.
- "Parent Equity Plans" means the DraftKings Inc. 2020 Incentive Award Plan, the DraftKings Inc. 2017 Equity Incentive Plan, the DraftKings Inc. 2012 Stock Option & Restricted Stock Incentive Plan and the SBTech (Global) Limited 2011 Global Share Option Plan.
- "<u>Parent Gaming Licenses</u>" means any License issued or granted to Parent or any of its Subsidiaries by a Gaming Regulatory Authority.
 - "Parent Indemnified Parties" has the meaning set forth in Section 8.2(a).
- "Parent Reports" mean any reports, forms, proxy statements, registration statements and other statements, certifications and documents filed with or furnished to the SEC (including notes, exhibits and schedules thereto and all other information incorporated by reference and any amendments and supplements thereto) by Parent or DraftKings Holdings Inc. (formerly DraftKings Inc.), a Nevada corporation and the predecessor registrant to Parent, in each case, pursuant to the Exchange Act or the Securities Act.
- <u>"Parent Tax Knowledge Party"</u> means any of Jason Robins, Jason Park or Malcom Turner or, in the event that none of the foregoing individuals are reasonably available, any other authorized officer of Parent.
 - "Parent's Counsel" has the meaning set forth in Section 5.14(i).
 - "Parties" or "Party" has the meaning set forth in the Preamble.
 - "Payments Administrator" has the meaning set forth in Section 2.5(a).
 - "Payments Administrator Agreement" has the meaning set forth in Section 2.5(a).
- "Payroll Tax Executive Order" means the Presidential Memorandum on Deferring Payroll Tax Obligations in Light of the Ongoing COVID-19 Disaster, as issued on August 8, 2020 and including any administrative or other guidance published with respect thereto by any Governmental Entity.
- "<u>Performance Standards</u>" means (a) all applicable published safety standards and certifications, (b) all applicable contractual performance standards or specifications and (c) all applicable regulatory requirements for a product.
- "Permitted Liens" means the following Liens: (i) Liens for current Taxes, assessments or other governmental charges not yet due and payable, or which may be hereafter paid without penalty or that the taxpayer is contesting in good faith through appropriate Proceedings, and for which adequate reserves have been established in the accounting books and records of the Company and its Subsidiaries prior to the date hereof; (ii) mechanics', materialmens', carriers', workmen's, repairmen's or other like common law, statutory or consensual Liens arising or incurred in the Ordinary Course and which are not, in the aggregate, material to the Company and its Subsidiaries, taken as a whole; (iii) immaterial inchoate statutory liens to secure obligations to landlords, lessors or renters under leases or rental agreements; (iv) any minor and immaterial imperfection of title or similar liens, charges or encumbrances which individually or in the aggregate with other such liens, charges and encumbrances does not impair the value of or the ability to use or transfer the property subject to such lien, charge or encumbrance or the

use of such property in the conduct of the business of the Company or any of its Subsidiaries; and (v) non-exclusive licenses of Intellectual Property Rights granted by the Company or its Subsidiaries in the Ordinary Course.

- "Permitted Transfer" has the meaning set forth in Section 5.17(c).
- "Permitted Transferee" has the meaning set forth in Section 5.17(c).
- "Per Share Closing Consideration" has the meaning set forth in Section 2.1(b)(ii).
- "Per Share Merger Consideration" means the right of one share of Company Capital Stock to receive (i) the Per Share Closing Consideration, (ii) the Per Share Post-Closing Adjustment Amount, if any, (iii) the Per Share Retention Holdback Consideration, if any, (iv) the Per Share Reserve Holdback Consideration, if any and (v) the Fully Diluted Pro Rata Share Amount, if any, disbursed to Company Stockholders from the Stockholder Representative Expense Amount in accordance with Section 9.2.
- "<u>Per Share Portion</u>" means a fraction (a) the numerator of which is one, and (b) the denominator of which is the total number of Fully Diluted Non-Option Shares.
- "<u>Per Share Post-Closing Adjustment Amount</u>" means an amount in cash equal to the quotient of (a) the sum of (i) the Post-Closing Adjustment and (ii) the Purchase Price Adjustment Holdback, divided by (b) the number of Fully Diluted Shares; <u>provided</u> that if the value calculated pursuant to this definition is a negative number, the Per Share Post-Closing Adjustment Amount shall be zero.
- "<u>Per Share Reserve Holdback Consideration</u>" means an amount in cash equal to the quotient of (i) any amount released to the Company Securityholders from the Reserve Holdback as provided for in Section 2.4(c) divided by (ii) the total number of Fully Diluted Shares.
- "Per Share Retention Holdback Consideration" means an amount in cash equal to the quotient of (i) any amount released (whether in one or more releases) to the Company Securityholders from the Retention Holdback as provided for in Section 2.4(b) divided by (ii) the total number of Fully Diluted Shares.
- "Person" means any individual, corporation (including not-for-profit), general or limited partnership, limited liability company, joint venture, estate, trust, association, organization, Governmental Entity or other entity of any kind or nature.
- "Personal Information" means any information that identifies, relates to, describes, is reasonably capable of being associated with or could reasonably be used to identify or be linked, directly or indirectly, with, a particular individual, household, browser or device, and any information that constitutes "personal data," "personal information," or similar term as defined by applicable Laws or the Company's privacy policies.
- "<u>Post-Closing Adjustment</u>" means an amount equal to (i) the Final Merger Cash Consideration minus (ii) the Estimated Merger Cash Consideration.
- "Post-Closing Statement" means the written statement delivered pursuant to Section 2.6, setting forth Parent's good-faith calculations of the Merger Cash Consideration and the Post-Closing Adjustment prepared in accordance with GAAP, and each component thereof, including as separate line items (a) Net Working Capital (which shall be prepared in a manner consistent with Exhibit F) and the amount of the

Net Working Capital Difference, (b) Cash on Hand, (c) Closing Indebtedness, and (d) Company Transaction Expenses (including Company Transaction Expenses Paid by Parent).

- "Pre-Closing Tax Period" means any taxable period ending on or before the Closing Date and, with respect to any taxable period beginning on or before and ending after the Closing Date, the portion of such taxable period ending on and including the Closing Date.
 - "Pre-Closing Tax Return" has the meaning set forth in Section 5.14(a).
 - "Privacy and Security Obligations" has the meaning set forth in Section 3.17(I).
- "Proceeding" means any action, cause of action, claim, demand, litigation, suit, investigation, grievance, citation, summons, subpoena, inquiry, audit, hearing, originating application to a tribunal, arbitration or other similar proceeding of any nature, civil, criminal, regulatory, administrative or otherwise, whether in equity or at Law, in Contract, in tort or otherwise.
 - "Private Placement Shares" has the meaning set forth in Section 3.24.
- "<u>Pro Rata Share</u>" means, with respect to a Company Stockholder, the percentage equal to the quotient of (a) the total number of shares of Company Capital Stock held by such Person immediately prior to the Initial Effective Time (as described opposite such Person's name in the Funds Flow Memorandum) divided by (b) the total number of Fully Diluted Non-Option Shares.
 - "Purchase Price Adjustment Holdback" has the meaning set forth in Section 2.4.
- "R&W Insurance Policy" means the buyer-side representations and warranties insurance policy being obtained by Parent in connection with the Transactions.
 - "Registrable Shares" has the meaning set forth in Section 5.17(a).
- "Registered" means issued by, registered with, renewed by or the subject of a pending application before any Governmental Entity.
- "Related Party" means (i) any current or former director or manager (or nominee), or officer of the Company or any of its Subsidiaries, (ii) any five percent (5%) or greater Company Stockholder on a fully-diluted basis and (iii) any first-degree relative, spouse, officer, director or Affiliate of any of the foregoing Persons.
 - "Remedy Action" has the meaning set forth in Section 5.3(b).
 - "Representations Survival Period" has the meaning set forth in Section 8.1(b)(ii).
- "Representative" of a Person means any Affiliate, stockholder, member, partner, officer, director, officer or employee of such Person or any investment banker, attorney, accountant, broker or other advisor, agent or representative of such Person.
 - "Required Stockholder Approval" has the meaning set forth in Section 3.3(a).
 - "Requisite Regulatory Approvals" has the meaning set forth in Section 6.1(b).
 - "Resale Registration Statement" has the meaning set forth in Section 5.17(a).

- "Reserve Holdback" has the meaning set forth in Section 2.4(c).
- "Reserve Holdback Amount" has the meaning set forth in Section 2.4(c).
- "Reserve Release Date" has the meaning set forth in Section 2.4(c).
- "Reserved Matter" has the meaning set forth in Section 2.4(c).
- "Retention Holdback" has the meaning set forth in Section 2.4(b).
- "Retention Holdback Amount" has the meaning set forth in Section 2.4(b).
- "Review Period" has the meaning set forth in Section 2.6(c)(i).
- "SEC" means the United States Securities and Exchange Commission.
- "Second Release Date" has the meaning set forth in Section 2.4(b).
- "Securities Act" means the Securities Act of 1933.
- "Seed Stock" has the meaning set forth in the Recitals.
- "Seller Group" has the meaning set forth in Section 10.13.

"Seller Indemnified Taxes" means, without duplication (i) any Taxes of the Company or any of its Subsidiaries relating or attributable to any Pre-Closing Tax Period (including such Taxes that are not yet due and payable; and including any Taxes the payment of which has been deferred (including any payroll taxes with respect to any Pre-Closing Tax Period deferred under the CARES Act or any Taxes deferred under Section 965(h) of the Code) as of the Closing Date), determined with respect to any Straddle Period, in accordance with Section 5.14(b); (ii) any Taxes of any Person (other than the Company or any of its Subsidiaries) for which the Company or any of its Subsidiaries is liable as a result of (x) having been a member of an affiliated, consolidated, combined, unitary or similar group prior to the Closing and (y) being a transferee or successor, or as a result of a transaction or Contract entered into prior to the Closing (other than commercial Contracts not primarily related to Taxes (such as financing or employment Contracts with Tax gross-up obligations or leases with Tax escalation provisions)); and (iii) any Transfer Taxes for which the Company Securityholders are responsible pursuant to Section 5.14(d); provided that, notwithstanding anything to the contrary herein, Seller Indemnified Taxes shall (A) not include (1) Taxes reflected Net Working Capital or Indebtedness, (2) any Taxes attributable to actions of the DK Parties, their Affiliates, the Initial Surviving Company, or the Subsequent Surviving Company on the Closing Date after the occurrence of the Initial Effective Time that occur outside the Ordinary Course, other than as explicitly contemplated by this Agreement, (3) any Taxes resulting from any breach by Parent of Section 5.14(j), (4) any Transfer Taxes that Parent is responsible for pursuant to Section 5.14(d), (5) any Excluded Payroll Taxes, and (6) any Taxes resulting from an election under Section 338 of the Code (or any similar provision of state, local or non-U.S. Law) with respect to the transactions contemplated by this Agreement and (B) be calculated by including as deductions against taxable income in the Pre-Closing Tax Period, all Transaction Deductions to the extent (i) deductible at a "more likely than not" (or higher) level of comfort in such Pre-Closing Tax Period (ii) such Transaction Deductions were not taken into account as a reduction to the Unpaid Pre-Closing Income Tax Amount in the Final Closing Statement.

"Selling Stockholder" has the meaning set forth in Section 5.17(a).

- "Software" means any computer program, application, middleware, firmware, microcode and other software, including operating systems, software implementations of algorithms, models and methodologies, in each case, whether in source code, object code, binary code, or other form or format, including libraries, subroutines and other components thereof, and all documentation relating thereto.
- "Specified Options" means those certain Company Options listed on Section 3.2 of the Company Disclosure Letter.
 - "Stock Plan" has the meaning set forth in Section 2.10(a).
 - "Stockholder Representative" has the meaning set forth in the Preamble.
 - "Stockholder Representative Expense Amount" means an amount in cash equal to \$100,000.
 - "Straddle Period" has the meaning set forth in Section 5.14(b).
 - "Subsequent Effective Time" has the meaning set forth in Section 1.4.
 - "Subsequent Merger" has the meaning set forth in the Recitals.
 - "Subsequent Surviving Company" has the meaning set forth in the Recitals.
- "Subsidiary" means, with respect to any Person, any other Person of which at least a majority of the securities or ownership interests having by their terms ordinary voting power to elect a majority of the board of directors or other Persons performing similar functions is directly or indirectly owned or controlled by such Person and/or by one or more of its Subsidiaries.
 - "Support Agreement" has the meaning set forth in the Recitals.
 - "Takeover Statute" has the meaning set forth in Section 3.12.
- "<u>Tax</u>" or "<u>Taxes</u>" means taxes including all federal, state, local and foreign income, profits, franchise, gross receipts, environmental, customs duty, capital stock, severances, stamp, payroll, sales, employment, unemployment, disability, use, property, withholding, excise, production, value-added, occupancy, escheat, unclaimed property, and other taxes, duties or assessments of any nature whatsoever, together with all interest, penalties and additions imposed with respect to such amounts and any interest in respect of such penalties and additions.
- "<u>Tax Proceeding</u>" means any inquiry, claim, audit, action, suit, Proceeding, examination, contest, litigation or investigation by any Tax authority in respect of Taxes.
- "<u>Tax Return</u>" means any returns and reports (including elections, declarations, disclosures, schedules, estimates and information returns) supplied or required to be supplied to a Tax authority relating to Taxes.
 - "Third Party Claim" has the meaning set forth in Section 8.3(b).
 - "Trade Secrets" has the meaning set forth in the definition of "Intellectual Property Rights."
- "<u>Transaction Deductions</u>" means all items of loss or deduction for U.S. federal income and state and local Tax purposes resulting from or attributable to: (i) Company Transaction Expenses or any compensatory payments made in connection with the Mergers and the transactions contemplated by this

Agreement and (ii) any fees, expenses, premiums and penalties with respect to the prepayment of debt and the write-off or acceleration of the amortization of deferred financing; provided that, to the extent applicable, the parties to this Agreement agree to make the safe harbor election set forth in IRS Revenue Procedure 2011-29 to determine the amount of deductions attributable to the payment of any success-based fees within the scope of such revenue procedure.

"Transactions" has the meaning set forth in the Recitals.

"Transfer Taxes" means all U.S. federal, state and local sales, use, value added, transfer (including real property transfer), stamp, documentary, filing, recordation and other similar taxes and fees that may be imposed or assessed with respect to the transactions undertaken pursuant to this Agreement, together with any interest, additions or penalties with respect thereto and any interest in respect of such additions or penalties. For the avoidance of doubt, Transfer Taxes shall not include any Taxes measured, in whole or in part, by reference to income or gain.

"Underwater Option" has the meaning set forth in Section 2.10(a)(ii).

"Undisclosed Information" has the meaning set forth in Section 3.29.

"Unpaid Pre-Closing Income Tax Amount" means the amount, if any, of income Taxes of the Company and its Subsidiaries described in clause (i) of the definition of "Seller Indemnified Taxes" for the current Pre-Closing Tax Period that are not yet paid (including such Taxes that are accrued but not yet due and payable) as of immediately prior to the Closing and that are not paid by the Company concurrently with the Closing, determined (a) in the case of the pre-closing portion of any Straddle Period, in accordance with Section 5.14(b), (b) except as otherwise provided in this definition, in accordance with the past practices (including reporting positions, elections and accounting methods) of the Company and its Subsidiaries in preparing its Tax Returns, (c) by including as deductions against taxable income in the Pre-Closing Tax Period, all Transaction Deductions to the extent deductible at a "more likely than not" (or higher) level of comfort in such Pre-Closing Tax Period, (d) by taking into account, to the extent applicable, any estimated or other Tax payments for any Pre-Closing Tax Period as a reduction of the liability for Taxes for such period, (e) by excluding any liabilities, accruals or reserves for contingent Taxes or uncertain Tax positions and (f) by excluding any Taxes attributable to any election under Section 338 of the Code made in connection with the Mergers or attributable to actions of the DK Parties, their Affiliates, the Initial Surviving Company, or the Subsequent Surviving Company on the Closing Date after the occurrence of the Initial Effective Time that occur outside the Ordinary Course, other than as explicitly contemplated by this Agreement.

"<u>Unvested Company Option</u>" means any Company Option that is outstanding and unexercised immediately prior to the Initial Effective Time that is not a Vested Company Option.

"UPE" has the meaning set forth in Section 5.2(a).

"Vested Company Option" means any vested and exercisable Company Option that is outstanding and unexercised immediately prior to the Initial Effective Time. For purposes of clarification, for any outstanding award of Company Options that is partially vested immediately prior to the Initial Effective Time, only the vested portion of such award shall be considered a Vested Company Option, and the unvested portion of such award shall be considered an Unvested Company Option.

"Willful Breach" means an intentional and willful material breach, or an intentional and willful material failure to perform, in each case that is the consequence of an act or omission by a Party with the

Agreer		'oncont'' has	tha maanina	ant fouth in	the Desitele		
	viitteii C	consent" has	the meaning	set forth in	the Recitais		



DraftKings Reports Fourth Quarter Revenue of \$1,231 Million; Raises 2024 Revenue Guidance Midpoint to \$4.775 Billion and 2024 Adjusted EBITDA Guidance Midpoint to \$460 Million

Boston, MA – February 15, 2024 — DraftKings Inc. (Nasdaq: DKNG) ("DraftKings" or the "Company") today announced its fourth quarter and fiscal year 2023 financial results. The Company also posted a letter to shareholders and an earnings presentation on the Investor Relations section of its website at investors.draftkings.com.

Fourth Quarter 2023 Highlights

For the three months ended December 31, 2023, DraftKings reported revenue of \$1,231 million, an increase of \$376 million, or 44%, compared to \$855 million during the same period in 2022. The increase in the Company's fourth quarter 2023 revenue was driven primarily by continued healthy customer engagement, efficient acquisition of new customers, the expansion of the Company's Sportsbook product offering into new jurisdictions, and product innovation leading to increased parlay mix and thus higher structural sportsbook hold percentage. Customer-friendly sport outcomes primarily in the final two weeks of November impacted DraftKings' revenue and Adjusted EBITDA in the fourth quarter of 2023 by approximately (\$175) million and approximately (\$126) million, respectively.

"DraftKings ended 2023 with excellent performance across customer acquisition, retention and engagement as well as structural sportsbook hold percentage despite the worst stretch of sport outcomes we have seen as a public company in the fourth quarter," said Jason Robins, DraftKings' Chief Executive Officer and Co-founder. "Looking ahead to 2024 and beyond, our focus remains on disciplined execution against our core value drivers, an unwavering commitment to customer centricity, and fulfilling our product roadmap to consistently differentiate ourselves competitively."

"In 2023 we delivered on our commitments to generate outstanding revenue growth and drive significant operating efficiencies," said Jason Park, DraftKings' Chief Financial Officer. "Based on continued strong underlying fundamentals through the first six weeks of 2024 on top of excellent customer acquisition in the fourth quarter, we are raising the midpoint of our fiscal year 2024 revenue guidance range to \$4.775 billion from \$4.65 billion and the midpoint of our fiscal year 2024 Adjusted EBITDA guidance range to \$460 million from \$400 million. We expect 2024 to mark our first full year of positive Adjusted EBITDA, demonstrating clear progress toward the goals we presented at our November 2023 Investor Day."

Continued Healthy Growth in Customer Retention, Acquisition, and Engagement

- Monthly Unique Payers ("MUPs") increased to 3.5 million average monthly unique paying customers in the fourth quarter of 2023, representing an increase of 37% compared to the fourth quarter of 2022. This increase reflects strong customer acquisition and retention across DraftKings' Sportsbook and iGaming products as well as the expansion of its Sportsbook product into new jurisdictions.
- Average Revenue per MUP ("ARPMUP") was \$116 in the fourth quarter of 2023, representing a 6% increase compared to the same period in 2022. This increase was primarily due to an increase in the Company's structural sportsbook hold rate, partially offset by customer-friendly sport outcomes. The increase in ARPMUP in the fourth quarter of 2023 would have been 22% when adjusting for customer-friendly sport outcomes.
- Detailed financial data and other information for the fourth quarter of 2023 is available in the financial statements set forth below under the caption "Financial Results."

Raising Fiscal Year 2024 Revenue and Adjusted EBITDA Guidance

• DraftKings is raising its fiscal year 2024 revenue guidance to a range of \$4.65 billion to \$4.90 billion from the range of \$4.50 billion to \$4.80 billion, which the Company previously announced on November 2,



2023. The Company's updated 2024 revenue guidance range equates to year-over-year growth of 27% to 34%.

- DraftKings is also increasing its fiscal year 2024 Adjusted EBITDA guidance. The Company now expects fiscal year 2024 Adjusted EBITDA of between \$410 million and \$510 million compared to its prior fiscal year 2024 Adjusted EBITDA guidance of between \$350 million and \$450 million, which the Company previously announced on November 2, 2023.
- The Company's revenue and Adjusted EBITDA guidance for fiscal year 2024 includes all of its existing jurisdictions as well as mobile sports betting in Puerto Rico and North Carolina, which have authorized mobile sports betting and collectively represent approximately 4% of the U.S. population. DraftKings expects to launch its Sportsbook product in North Carolina on March 11, 2024 pending market access, licensure, regulatory approvals, and contractual approvals where applicable.
- DraftKings' revenue and Adjusted EBITDA guidance for fiscal year 2024 excludes the estimated impact of the Company's proposed acquisition of Jackpocket, which DraftKings will incorporate into its guidance following the close of the proposed acquisition.

Mobile Sports Betting and iGaming Footprint

- Following the launches of its Sportsbook product in Maine on November 3, 2023 and in Vermont on January 11, 2024, DraftKings is live with mobile sports betting in 24 states that collectively represent approximately 46% of the U.S. population.
- DraftKings is also live with iGaming in 5 states, representing approximately 11% of the U.S. population.
- DraftKings is live with its Sportsbook and iGaming products in Ontario, Canada, which represents approximately 40% of Canada's population.
- North Carolina and Puerto Rico have authorized mobile sports betting and collectively represent approximately 4% of the U.S. population. DraftKings expects to launch its Sportsbook product in North Carolina on March 11, 2024 pending market access, licensure, regulatory approvals, and contractual approvals where applicable.
- To date in 2024, 7 states that collectively represent approximately 11% of the U.S. population have either introduced legislation to legalize mobile sports betting or introduced a bill that may result in a mobile sports betting referendum during an upcoming election. In addition, 5 states that collectively represent approximately 12% of the U.S. population have either introduced legislation to legalize iGaming or introduced a bill that may result in an iGaming referendum during an upcoming election.

Webcast and Conference Call Details

As previously announced, DraftKings will host a conference call and audio webcast tomorrow, Friday, February 16, 2024, at 8:30 a.m. ET, during which management will discuss the Company's results for the quarter and provide commentary on business performance. A question-and-answer session will follow the prepared remarks.

To listen to the audio webcast and live question and answer session, please visit DraftKings' investor relations website at investors.draftkings.com. A live audio webcast of the earnings conference call will be available on the Company's website at investors.draftkings.com, along with a copy of this press release, the Company's Annual Report on Form 10-K, an earnings presentation and a letter to shareholders. The audio webcast will be available on the Company's investor relations website until 11:59 p.m. ET on March 31,



Financial Results

DraftKings' fourth quarter and full-year 2023 financial results, as well as the financial results for their respective comparative periods, are presented below:



DRAFTKINGS INC. CONSOLIDATED BALANCE SHEETS

(Unaudited) (Amounts in thousands, except par value)

December 31. 2023 2022 Assets Current assets: Cash and cash equivalents 1,270,503 \$ 1,309,172 Restricted cash 11,700 Cash reserved for users 341,290 469,653 Receivables reserved for users 301.770 160.083 51 097 Accounts receivables 47 539 94,836 Prepaid expenses and other current assets 98,565 2,071,367 2,084,841 Total current assets 60.102 Property and equipment, net 60.695 Intangible assets, net 690,620 776,934 Goodwill 886,373 886,373 Operating lease right-of-use assets 93,985 65,957 Equity method investments 10,280 10,080 Deposits and other non-current assets 131,546 155,865 Total assets 3,944,866 4,040,152 Liabilities and Stockholders' equity Current liabilities: Ś 639,599 517.587 Accounts payable and accrued expenses 851,898 686,173 Liabilities to users Operating lease liabilities, current portion 11,499 4,253 Other current liabilities 46.624 38,444 Total current liabilities 1.549.620 1.246.457 Convertible notes, net of issuance costs 1,253,760 1,251,103 Non-current operating lease liabilities 80.827 69.332 Warrant liabilities 63,568 10,680 Long-term income tax liabilities 69,858 72,810 Other long-term liabilities 83,975 70,029 Total liabilities 3,104,560 2,717,459 Commitments and contingent liabilities (Note 7 and Note 15) Stockholders' equity: Class A common stock, \$0.0001 par value; 900,000 shares authorized as of December 31, 2023 and December 31, 2022, respectively; 484,598 and 459,265 shares issued and 472,697 and 450,575 outstanding at December 31, 2023 and December 31, 2022, respectively

Class B common stock, \$0.0001 par value; 900,000 shares authorized as of December 31, 2023 and December 31, 2022; 393,014 shares issued and outstanding at December 31, 2023 and December 31, 2022 is 30,014 shares issued and outstanding at December 31, 2023 in December 31, 2022 is 30,014 shares issued and outstanding at December 31, 2023 in December 31, 2022 is 30,014 shares issued and outstanding at December 31, 2023 in December 31, 2022 is 30,014 shares issued and outstanding at December 31, 2023 in December 31, 2023 in December 31, 2022 is 30,014 shares issued and outstanding at December 31, 2023 in Decemb \$ 46 \$ 45 39 Treasury stock, at cost; 11,901 and 8,690 shares as of December 31, 2023 and December 31, 2022, respectively (412,182) (332,133) Additional paid-in capital 7,149,858 6,750,055 Accumulated deficit (5,933,943) (5,131,801) 36,488 Accumulated other comprehensive income 36,488 1.322.693 Total stockholders' equity 840.306 Total liabilities and stockholders' equity \$ 3,944,866 4,040,152

Due to the timing of the consummation of the Company's acquisition of Golden Nugget Online Gaming, Inc. ("GNOG" and such acquisition, the "GNOG Transaction"), the above periods, to the extent applicable, exclude GNOG's operations prior to the closing date of May 5, 2022.



DRAFTKINGS INC. CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited) (Amounts in thousands, except loss per share data)

	Three Months Ended December 31,			Year ended I	Year ended December 31,		
		2023	2022	2023	2022		
Revenue	\$	1,230,857	\$ 855,133	\$ 3,665,393	\$ 2,240,461		
Cost of revenue		716,658	485,435	2,292,175	1,484,273		
Sales and marketing		290,775	345,282	1,200,718	1,185,977		
Product and technology		88,157	83,394	355,156	318,247		
General and administrative		179,076	173,244	606,569	763,720		
Loss from operations		(43,809)	(232,222)	(789,225)	(1,511,756)		
Other income (expense):							
Interest income		18,792	10,992	58,418	21,353		
Interest expense		(688)	(668)	(2,679)	(2,651)		
(Loss) gain on remeasurement of warrant liabilities		(12,716)	9,197	(57,543)	29,396		
Other (loss) gain, net		929	(19,866)	(224)	20,700		
Loss before income tax provision (benefit) and loss (income) from equity method investment		(37,492)	(232,567)	(791,253)	(1,442,958)		
Income tax (benefit) provision		6,860	9,714	10,170	(67,866)		
Loss (income) from equity method investment		269	416	719	2,895		
Net loss attributable to common stockholders	\$	(44,621)	\$ (242,697)	\$ (802,142)	\$ (1,377,987)		
Loss per share attributable to common stockholders:							
Basic and diluted	\$	(0.10)	\$ (0.53)	\$ (1.73)	\$ (3.16)		

Due to the timing of the consummation of the GNOG Transaction, the above periods, to the extent applicable, exclude GNOG's operations prior to the closing date of May 5, 2022.

DRAFTKINGS INC.

NON-GAAP FINANCIAL MEASURES
(Unaudited)
(Amounts in thousands, except earnings (loss) per share data)

	Three Months Ended December 31,			Year ended December 31,			
	 2023		2022		2023		2022
Adjusted EBITDA	\$ 151,018	\$	(49,927)	\$	(151,035)	\$	(721,781)
Adjusted Earnings (Loss) Per Share*	\$ 0.29	\$	(0.14)	\$	(0.41)	\$	(1.77)

Due to the timing of the consummation of the GNOG Transaction, the above periods, to the extent applicable, exclude GNOG's operations prior to the closing date of May 5, 2022.



DRAFTKINGS INC. CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited)

(Amounts in thousands)

2023 2022 Cash Flows from Operating Activities: Net loss attributable to common shareholders
Adjustments to reconcile net loss to net cash flows used in operating activities: \$ (802,142) \$ (1,377,987) Depreciation and amortization
Non-cash interest expense, net 201,920 169,252 870 386 Stock-based compensation expense
Loss (gain) on remeasurement of warrant liabilities 398,463 578,799 (29,396) 57 543

	====		:	
Cash and cash equivalents, restricted cash, and cash reserved for users at the end of period	\$	1,623,493	\$	1,778,825
Cash reserved for users		341,290		469,653
Restricted cash		11,700		_
Cash and cash equivalents	\$	1,270,503	\$	1,309,172
Disclosure of cash and cash equivalents, restricted cash, and cash reserved for users				
Cash and cash equivalents, restricted cash, and cash reserved for users at the end of period	\$	1,623,493	\$	1,778,825
Cash and cash equivalents, restricted cash, and cash reserved for users at the beginning of period		1,778,825		2,629,842
Net decrease in cash and cash equivalents, restricted cash, and cash reserved for users		(155,332)		(851,017)
Net cash flows (used in) provided by financing activities		(63,221)		(16,732)
Proceeds from exercise of stock options		16,540		8,743
Purchase of treasury stock		(80,049)		(25,519)
Proceeds from exercise of warrants		288		44
Cash Flow from Financing Activities:		(,,		(11, 11,
Net cash flows used in investing activities		(90,360)		(208,766)
Other investing activities, net		(1,400)		(8,614)
Cash paid for acquisitions, net of cash acquired		, .25		(96,507)
Proceeds from (purchase of) marketable equity securities and other financial assets		24,425		(7,213)
Acquisition of gaming licenses		(12,105)		(7,213)
Cash paid for internally developed software costs		(80,378)		(64,030)
Purchases of property and equipment		(20,902)		(32,402)
Cash Flows from Investing Activities:		(1,751)		(023,319)
Net cash flows used in operating activities		(1,751)		(625,519)
Other long-term liabilities		11,087		15,383
Liabilities to users Long-term income tax liability		2,952		(9,267)
Accounts payable and accrued expenses Liabilities to users		103,593 165,725		95,269 152,985
Operating leases, net		6,558		1,304
Deposits and other non-current assets		(19,355)		(4,921)
Prepaid expenses and other current assets		2,451		(26,217)
Accounts receivables		3,558		2,506
Receivables reserved for users		(141,687)		(105,320)
Change in operating assets and liabilities, net of effect of acquisitions:				
Other expenses, net		554		(7,268)
Deferred income taxes		5,849		(73,407)
Loss (gain) on marketable equity securities and other financial assets, net		75		(10,999)
Loss (gain) from equity method investment		719		2,895
Loss (gain) on remeasurement of warrant liabilities		57,543		(29,396)



Supplemental Disclosure of Noncash Investing and Financing Activities:		
Investing activities included in accounts payable and accrued expenses	569	9,155
Equity consideration issued for acquisitions	_	460,128
Decrease in warrant liabilities from cashless exercise of warrants	4,654	_
Supplemental Disclosure of Cash Activities:		
(Decrease) increase in cash reserved for users	(128,363)	(7,297)
Cash paid for interest	_	_
Cash paid for income taxes, net of refunds	8,341	10,366

Due to the timing of the consummation of the GNOG Transaction, the above periods, to the extent applicable, exclude GNOG's operations prior to the closing date of May 5, 2022.

Non-GAAP Financial Measures

This press release includes Adjusted EBITDA and Adjusted Earnings (Loss) Per Share, which are non-GAAP financial measures that DraftKings uses to supplement its results presented in accordance with U.S. generally accepted accounting principles ("GAAP"). The Company believes Adjusted EBITDA and Adjusted Earnings (Loss) Per Share are useful in evaluating its operating performance, similar to measures reported by its publicly-listed U.S. competitors, and regularly used by security analysts, institutional investors and other interested parties in analyzing operating performance and prospects. Adjusted EBITDA and Adjusted Earnings (Loss) Per Share are not intended to be substitutes for any GAAP financial measures, and, as calculated, may not be comparable to other similarly titled measures of performance of other companies in other industries or within the same industry.

DraftKings defines and calculates Adjusted EBITDA as net income (loss) before the impact of interest income or expense (net), income tax provision or benefit, and depreciation and amortization, and further adjusted for the following items: stock-based compensation; transaction-related costs; litigation, settlement and related costs; advocacy and other related legal expenses; gain or loss on remeasurement of warrant liabilities; and other non-recurring and non-operating costs or income, as described in the reconciliation below.

DraftKings defines and calculates Adjusted Earnings (Loss) Per Share as basic and diluted loss per share attributable to common stockholders before the impact of amortization of acquired intangible assets; stock-based compensation; transaction-related costs; litigation, settlement and related costs; advocacy and other related legal expenses; gain or loss on remeasurement of warrant liabilities; and other non-recurring and non-operating costs or income, as described in the reconciliation below.

DraftKings includes these non-GAAP financial measures because they are used by management to evaluate the Company's core operating performance and trends and to make strategic decisions regarding the allocation of capital and new investments. Adjusted EBITDA and Adjusted Earnings (Loss) Per Share exclude certain expenses that are required in accordance with GAAP because they are non-recurring items (for example, in the case of transaction-related costs and advocacy and other related legal expenses), non-cash expenditures (for example, in the case of amortization of acquired intangible assets, depreciation and amortization, remeasurement of warrant liabilities and stockbased compensation), or non-operating items which are not related to the Company's underlying business performance (for example, in the case of interest income and expense and litigation, settlement and related costs).



The unaudited table below presents the Company's Adjusted EBITDA reconciled to its net loss, which is the most directly comparable financial measure calculated in accordance with GAAP, for the periods indicated:

	Three Months Ended December 31,		Year ended [December 31,	
(amounts in thousands)	2023	2022	2023	2022	
Net Loss	\$ (44,621)	\$ (242,697)	\$ (802,142)	\$ (1,377,987)	
Adjusted for:					
Depreciation and amortization (1)	55,198	48,623	201,920	169,252	
Interest (income) expense, net	(18,104)	(10,324)	(55,739)	(18,702)	
Income tax (benefit) provision	6,860	9,714	10,170	(67,866)	
Stock-based compensation (2)	113,517	130,161	398,463	578,799	
Transaction-related costs (3)	1,954	2,285	3,060	17,315	
Litigation, settlement, and related costs (4)	23,910	1,224	34,500	7,010	
Advocacy and other related legal expenses (5)	_	_	-	16,558	
Loss (gain) on remeasurement of warrant liabilities	12,716	(9,197)	57,543	(29,396)	
Other non-recurring, special project and non-operating (income) costs (6)	(412)	20,284	1,190	(16,764)	
Adjusted EBITDA	\$ 151,018	\$ (49,927)	\$ (151,035)	\$ (721,781)	

Due to the timing of the consummation of the GNOG Transaction, the above periods, to the extent applicable, exclude GNOG's operations prior to the closing date of May 5, 2022.

- (1) The amounts include the amortization of acquired intangible assets of \$29.3 million and \$30.0 million for the three months ended December 31, 2023 and 2022, respectively, and \$117.3 million and \$106.1 million for the years ended 2023 and 2022, respectively.
- Reflects stock-based compensation expenses resulting from the issuance of awards under incentive plans.
 Includes capital markets advisory, consulting, accounting and legal expenses related to evaluation, negotiation and integration costs incurred in connection with proposed, pending or completed transactions and offerings, including costs related to the GNOG Transaction.
- (4) Primarily includes external legal costs related to litigation and litigation settlement costs deemed unrelated to DraftKings' core business operations.
- Reflects non-recurring and non-ordinary course costs relating to advocacy efforts and other legal expenses in jurisdictions where DraftKings does not operate certain product offerings and is actively seeking licensure, or similar approval, for those product offerings. This adjustment excludes (i) costs relating to advocacy efforts and other legal expenses in jurisdictions where DraftKings does not operate that are incurred in the ordinary course of business and (ii) costs relating to advocacy efforts and other legal expenses incurred in jurisdictions where related legislation has been passed and DraftKings currently operates. For 2022, those costs primarily related to our support of Proposition 27 in California.

 Primarily includes the change in fair value of certain financial assets, as well as the Company's equity method share of the investee's losses and other costs relating to non-recurring and non-operating items.

The unaudited table below presents the Company's Adjusted Earnings (Loss) Per Share reconciled to its basic loss per share attributable to common stockholders, which is the most directly comparable financial measure calculated in accordance with GAAP, for the periods indicated:

		Three Months Ende	Year ended December 31,		
	2	2023	2022	2023	2022
Basic loss per share attributable to common stockholders	\$	(0.10)	\$ (0.54)	\$ (1.73)	\$ (3.16)
Adjusted for:					
Amortization of acquired intangible assets		0.06	0.06	0.25	0.24
Discrete tax provision (benefit) attributed to the GNOG acquisition		_	0.02	_	(0.16)
Stock-based compensation (1)		0.24	0.29	0.86	1.33
Transaction-related costs (2)		_	_	0.01	0.03
Litigation, settlement, and related costs (3)		0.05	0.01	0.07	0.02
Advocacy and other related legal expenses (4)		_	_	_	0.04
Loss (gain) on remeasurement of warrant liabilities		0.03	(0.02)	0.12	(0.07)
Other non-recurring costs and non-operating (income) costs (5)		_	0.04	_	(0.04)
Adjusted Earnings (Loss) Per Share*	\$	0.29	\$ (0.14)	\$ (0.41)	\$ (1.77)



Due to the timing of the consummation of the GNOG Transaction, the above periods, to the extent applicable, exclude GNOG's operations prior to the closing date of May 5, 2022.

- * Weighted average number of shares used to calculate Adjusted Earnings (Loss) Per Share for the fourth quarter and year ending December 31, 2023 was 468.1 million and 462.6 million, respectively; totals may not sum due to rounding.
- 1) Reflects stock-based compensation expenses per share resulting from the issuance of awards under incentive plans.
- (2) Reflects capital markets advisory, consulting, accounting and legal expenses per share related to evaluation, negotiation and integration costs incurred in connection with proposed, pending or completed transactions and offerings, including costs relating to DraftKings' acquisition of Golden Nugget Online Gaming, Inc. in 2022.
- (3) Primarily reflects external legal costs related to litigation and litigation settlement costs, in each case per share, deemed unrelated to DraftKings' core business.
- (4) Reflects non-recurring and non-ordinary course costs per share relating to advocacy efforts and other legal expenses in jurisdictions where DraftKings does not operate certain product offerings and is actively seeking licensure, or similar approval, for those product offerings. This adjustment excludes (i) costs relating to advocacy efforts and other legal expenses in jurisdictions where DraftKings does not operate that are incurred in the ordinary course of business and (ii) costs relating to advocacy efforts and other legal expenses incurred in jurisdictions where related legislation has been passed and DraftKings currently operates. For 2022, those costs primarily related to our support of Proposition 27 in California.
- (5) Primarily includes the change in fair value of certain financial assets, as well as the Company's equity method share of the investee's losses and other costs relating to non-recurring and non-operating items, in each case per share.

Information reconciling forward-looking fiscal year 2024 Adjusted EBITDA to its most directly comparable GAAP financial measure, net income (loss), is unavailable to DraftKings without unreasonable effort due to, among other things, certain items required for such reconciliations being outside of DraftKings' control and/or not being able to be reasonably predicted. Preparation of such reconciliations would require a forward-looking balance sheet, statement of income and statement of cash flow, prepared in accordance with GAAP, and such forward-looking financial statements are unavailable to the Company without unreasonable effort. DraftKings provides a range for its Adjusted EBITDA forecast that it believes will be achieved; however, the Company cannot provide any assurance that it can predict all of the components of the Adjusted EBITDA calculation. DraftKings provides a forecast for Adjusted EBITDA because it believes that Adjusted EBITDA, when viewed with DraftKings' results calculated in accordance with GAAP, provides useful information for the reasons noted above. However, Adjusted EBITDA is not a measure of financial performance or liquidity under GAAP and, accordingly, should not be considered as an alternative to net income (loss) or cash flow from operating activities or as an indicator of operating performance or liquidity.

About DraftKings

DraftKings Inc. is a digital sports entertainment and gaming company created to be the Ultimate Host and fuel the competitive spirit of sports fans with products that range across daily fantasy, regulated gaming and digital media. Headquartered in Boston and launched in 2012 by Jason Robins, Matt Kalish and Paul Liberman, DraftKings is the only U.S.-based vertically integrated sports betting operator. DraftKings' mission is to make life more exciting by responsibly creating the world's favorite real-money games and betting experiences. DraftKings Sportsbook is live with mobile and/or retail sports betting operations pursuant to regulations in 26 states and in Ontario, Canada. The Company operates iGaming pursuant to regulations in five states and in Ontario, Canada under its DraftKings brand and pursuant to regulations in three states under its Golden Nugget Online Gaming brand. DraftKings' daily fantasy sports product is available in 44 states, certain Canadian provinces, and the United Kingdom. DraftKings is both an official daily fantasy and sports betting partner of the NFL, NHL, PGA TOUR, and UFC, as well as an official daily fantasy partner of NASCAR, an official sports betting partner of the NBA and an authorized gaming operator of MLB. In addition, DraftKings owns and operates both DraftKings Network and Vegas Sports Information Network (VSiN), to provide a multiplatform content ecosystem with original programming. DraftKings is committed to being a responsible steward of this new era in real-money gaming with a Company-wide focus on responsible gaming and corporate social responsibility.



Forward-Looking Statements

This press release contains forward-looking statements within the meaning of Section 21E of the Securities Exchange Act of 1934, as amended, and the Private Securities Litigation Reform Act of 1995, including statements about the Company, Jackpocket and their respective industries that involve substantial risks and uncertainties. All statements, other than statements of historical fact, contained in this press release, including statements regarding guidance, DraftKings' and Jackpocket's consummation of the proposed transaction and future results of operations or financial condition, strategic plans and focus, user growth and engagement, product initiatives, and the objectives and expectations of management for future operations (including launches in new jurisdictions and the expected timing thereof), are forward-looking statements. In some cases, you can identify forward-looking statements because they contain words such as "anticipate," "believe," "confident," "contemplate," "continue," "could," "estimate," "expect," "forecast," "going to," "intend," "may," "plan," "poised," "potential," "predict," "propose," "should," "target," "will," or "would" or the negative of these words or other similar terms or expressions. DraftKings cautions you that the foregoing may not include all of the forward-looking statements made in this press release.

You should not rely on forward-looking statements as predictions of future events. DraftKings has based the forward-looking statements contained in this press release primarily on its current expectations and projections about future events and trends, including the current macroeconomic environment, that it believes may affect its and Jackpocket's business, financial condition, results of operations, and prospects. These forward-looking statements are not guarantees of future performance, conditions or results, and involve a number of known and unknown risks, uncertainties, assumptions and other important factors, many of which are outside DraftKings' and Jackpocket's control and that could cause actual results or outcomes to differ materially from those discussed in the forward-looking statements. Important factors, among others, that may affect actual results or outcomes include, but are not limited to, the outcome of any legal proceedings that may be instituted against DraftKings and Jackpocket following the announcement of DraftKings' proposed acquisition of Jackpocket; the inability to complete DraftKings' proposed acquisition of Jackpocket, including the failure to obtain the requisite approvals of applicable regulatory authorities or the failure to satisfy certain other conditions to closing in the merger agreement for the proposed transaction; the occurrence of any event, change or other circumstance that could give rise to the termination of the merger agreement or could otherwise cause DraftKings' proposed acquisition of Jackpocket to fail to close; the risk that DraftKings' proposed acquisition of Jackpocket disrupts current plans or operations as a result of the announcement and consummation of the proposed transaction; the ability to recognize the anticipated benefits of DraftKings' proposed acquisition of Jackpocket, which may be affected by among other things, competition and the ability of DraftKings and Jackpocket to manage growth and retain its key employees; costs related to the proposed transactions; DraftKings' and Jackpocket's abilities to execute their respective business plans and meet their respective projections; potential litigation involving DraftKings or Jackpocket; changes in applicable laws or regulations, particularly with respect to online gaming, digital lottery courier or similar businesses; general economic and market conditions impacting demand for DraftKings' and Jackpocket's products and services; economic and market conditions in the media, entertainment, gaming, lottery and software industries in the jurisdictions in which DraftKings and Jackpocket operates; market and global conditions and economic factors, as well as the potential impact of general economic conditions, including inflation, rising interest rates and instability in the banking system, on DraftKings' and Jackpocket's liquidity, operations and personnel, as well as the risks, uncertainties, and other factors described in "Risk Factors" in DraftKings' fillings with the Securities and Exchange Commission (the "SEC"), which are available on the SEC's website at www.sec.gov. Additional information will be made available in other filings that DraftKings makes from time to time with the SEC. The forward-looking statements contained herein are based on DraftKings management's current expectations and beliefs and speak only as of the date hereof, and neither DraftKings nor Jackpocket makes any commitment to update or publicly release any revisions to forward-looking statements in order to reflect new information or subsequent events, circumstances or changes in expectations, except as required by law.



Media:

Media@draftkings.com

@DraftKingsNews

Investors:

Investors@draftkings.com



DraftKings Reaches Agreement to Acquire Jackpocket for \$750 Million

Transaction Expected to Drive \$60 Million to \$100 Million of Adjusted EBITDA in Fiscal Year 2026

BOSTON & NEW YORK – Feb. 15, 2024 — DraftKings Inc. (Nasdaq: DKNG) ("DraftKings" and the "Company") today announced that it has reached an agreement to acquire Jackpocket, the leading lottery app* in the United States, for total consideration of approximately \$750 million, with approximately 55 percent of the consideration payable in cash funded from the Company's balance sheet with no capital raise required and approximately 45 percent of the consideration payable in the Company's Class A common stock, subject to customary purchase price adjustments and the collar mechanism described below (the "Proposed Transaction").

Jackpocket is the leading provider of digital lottery services in the U.S. with proprietary and highly-scalable technology, a strong brand, and an outstanding founder-led management team. The Proposed Transaction will enable DraftKings to access and grow into the massive U.S. lottery industry, but more importantly strengthen its position in Sportsbook and iGaming through higher customer lifetime value – based on demonstrated cross-sell capabilities – and an enhanced customer acquisition engine.

"We are very excited to enter the rapidly growing U.S. digital lottery vertical with our acquisition of Jackpocket," said Jason Robins, Co-founder and CEO of DraftKings. "This transaction will create significant value for DraftKings not only by giving our customers another differentiated product to enjoy but also by improving our overall marketing efficiency similar to how our daily fantasy sports database created an advantage for DraftKings in OSB and iGaming."

"Together with DraftKings, we will be able to bring tremendous value to our customer base as we advance our mission to create a more convenient, fun, and responsible way to take part in the lottery," said Peter Sullivan, CEO of Jackpocket. "DraftKings' broad footprint and exceptional mobile products present an opportunity to meaningfully expand the digital lottery vertical, and we could not be more excited to come together with DraftKings."

Transaction Financial Impact

Conservatively assuming no additional OSB and iGaming legalization in the U.S., DraftKings expects the Proposed Transaction to drive \$260 million to \$340 million of incremental revenue and \$60 million to \$100 million of incremental Adjusted EBITDA in fiscal year 2026. On the same basis, assuming no additional OSB and iGaming legalization in the U.S., DraftKings expects the Proposed Transaction to drive \$350 million to \$450 million of incremental revenue and \$100 million to \$150 million of incremental Adjusted EBITDA in fiscal year 2028.

Details of the Transaction

Under the terms of the merger agreement entered into on February 11, 2024 (the "Merger Agreement"), Jackpocket stockholders will receive total consideration of approximately \$750 million on a fully diluted basis, consisting of approximately \$412.5 million in cash, subject to certain customary purchase price adjustments, and approximately \$337.5 million in the Company's Class A common stock, subject to the collar mechanism described below.

The stock consideration will be subject to a collar pursuant to which a variable number of shares of DraftKings' Class A common stock will be issued to Jackpocket stockholders in order to deliver a value of

approximately \$337.5 million, so long as the 30-trading-day volume weighted average price of DraftKings' Class A common stock as of the second trading day immediately preceding the closing of the Proposed Transaction (the "Closing Stock Price") remains between \$31.68 and \$42.86. In the event that DraftKings' Closing Stock Price is above \$42.86 or below \$31.68, Jackpocket stockholders will receive a fixed number of approximately 7,874,806 shares or approximately 10,654,149 shares, respectively, of DraftKings' Class A common stock, representing approximately 2% of the outstanding shares of DraftKings' Class A common stock.

The Merger Agreement and the Proposed Transaction have been approved by the Boards of Directors of each of DraftKings and Jackpocket, as well as Jackpocket's stockholders.

The Proposed Transaction is subject to the receipt of required regulatory approvals and other customary closing conditions and is expected to close by the second half of 2024. Additional details and information about the terms and conditions of the Proposed Transaction are included in a Current Report on Form 8-K, filed by DraftKings with the Securities and Exchange Commission (the "SEC").

DraftKings' top priority is stewarding safe and responsible play. This core tenet aligns closely with Jackpocket as a fellow member of the National Council on Problem Gambling ("NCPG"). Jackpocket is certified for responsible gaming through NCPG's assessment program (iCAP), and both DraftKings and Jackpocket provide tools and resources to encourage responsible gaming behavior, including spending and funding limits as well as self-exclusion. DraftKings believes the Proposed Transaction will enable the Company to safely and responsibly expand the lottery category, generating more revenue for states to fund important programs.

Advisors

Goldman Sachs & Co. LLC served as exclusive financial advisor to DraftKings, and Sullivan & Cromwell LLP served as legal counsel to DraftKings. The Raine Group served as exclusive financial advisor to Jackpocket, and Cooley LLP served as legal counsel to Jackpocket.

About DraftKings

DraftKings Inc. is a digital sports entertainment and gaming company created to be the Ultimate Host and fuel the competitive spirit of sports fans with products that range across daily fantasy, regulated gaming and digital media. Headquartered in Boston and launched in 2012 by Jason Robins, Matt Kalish and Paul Liberman, DraftKings is the only U.S.-based vertically integrated sports betting operator. DraftKings' mission is to make life more exciting by responsibly creating the world's favorite real-money games and betting experiences. DraftKings Sportsbook is live with mobile and/or retail sports betting operations pursuant to regulations in 26 states and in Ontario, Canada. The Company operates iGaming pursuant to regulations in five states and in Ontario, Canada under its DraftKings brand and pursuant to regulations in three states under its Golden Nugget Online Gaming brand. DraftKings' daily fantasy sports product is available in 44 states, certain Canadian provinces, and the United Kingdom. DraftKings is both an official daily fantasy and sports betting partner of the NFL, NHL, PGA TOUR, and UFC, as well as an official daily fantasy partner of NASCAR, an official sports betting partner of the NBA and an authorized gaming operator of MLB. In addition, DraftKings owns and operates both DraftKings Network and Vegas Sports Information Network (VSiN), to provide a multiplatform content ecosystem with original programming. DraftKings is committed to being a responsible steward of this new era in real-money gaming with a Company-wide focus on responsible gaming and corporate social responsibility.

About Jackpocket



Jackpocket is on a mission to create a more convenient, fun, and responsible way to participate in the lottery. The first licensed third-party lottery app in the United States, Jackpocket provides an easy, secure way to order official state lottery tickets. Jackpocket is currently available in Arizona, Arkansas, Colorado, Idaho, Massachusetts, Minnesota, Montana, Nebraska, New Hampshire, New Jersey, New Mexico, New York, Ohio, Oregon, Puerto Rico, Texas, Washington D.C., and West Virginia.

*Jackpocket was downloaded 9x more than its closest competitor in the digital lottery app category in fiscal year 2023 according to Sensor Tower estimates.

No Offer or Solicitation

This communication is not intended to and shall not constitute an offer to buy or sell or the solicitation of an offer to buy or sell any securities, or a solicitation of any vote or approval, nor shall there be any offer, solicitation or sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offer of securities shall be made in the United States absent registration under the U.S. Securities Act of 1933, as amended, or pursuant to an exemption from, or in a transaction not subject to, such registration requirements.

Forward-Looking Statements

This press release contains forward-looking statements within the meaning of Section 21E of the Securities Exchange Act of 1934, as amended, and the Private Securities Litigation Reform Act of 1995, including statements about the Company, Jackpocket and their respective industries that involve substantial risks and uncertainties. All statements, other than statements of historical fact, contained in this press release, including statements regarding guidance, DraftKings' and Jackpocket's consummation of the Proposed Transaction and future results of operations or financial condition, strategic plans and focus, user growth and engagement, product initiatives, and the objectives and expectations of management for future operations (including launches in new jurisdictions and the expected timing thereof), are forward-looking statements. In some cases, you can identify forward-looking statements because they contain words such as "anticipate," "believe," "confident," "contemplate," "continue," "could," "estimate," "expect," "forecast," "going to," "intend," "may," "plan," "poised," "potential," "predict," "project," "propose," "should," "target," "will," or "would" or the negative of these words or other similar terms or expressions. DraftKings cautions you that the foregoing may not include all of the forward-looking statements made in this press release.

You should not rely on forward-looking statements as predictions of future events. DraftKings has based the forward-looking statements contained in this press release primarily on its current expectations and projections about future events and trends, including the current macroeconomic environment, that it believes may affect its and Jackpocket's business, financial condition, results of operations, and prospects. These forward-looking statements are not guarantees of future performance, conditions or results, and involve a number of known and unknown risks, uncertainties, assumptions and other important factors, many of which are outside DraftKings' and Jackpocket's control and that could cause actual results or outcomes to differ materially from those discussed in the forward-looking statements. Important factors, among others, that may affect actual results or outcomes include, but are not limited to, the outcome of any legal proceedings that may be instituted against DraftKings and Jackpocket following the announcement of the Merger Agreement and the Proposed Transaction; the inability to complete the Proposed Transaction, including due to failure to obtain the requisite approvals of applicable regulatory authorities or the failure to satisfy certain other conditions to closing in the Merger Agreement; the occurrence of any event, change or other circumstance that could give rise to the termination of the Merger Agreement or could otherwise cause the Proposed Transaction to fail to



close; the risk that the Proposed Transaction disrupts current plans or operations as a result of the announcement and consummation of the Proposed Transaction; the ability to recognize the anticipated benefits of the Proposed Transaction, which may be affected by, among other things, competition and the ability of DraftKings and Jackpocket to grow and manage growth and retain its key employees; costs related to the Proposed Transaction; DraftKings' and Jackpocket's abilities to execute their respective business plans and meet their respective projections; potential litigation involving DraftKings or Jackpocket; changes in applicable laws or regulations, particularly with respect to online gaming, digital lottery courier or similar businesses; general economic and market conditions impacting demand for DraftKings' and Jackpocket's products and services; economic and market conditions in the media, entertainment, gaming, lottery and software industries in the jurisdictions in which DraftKings and Jackpocket operate; global conditions and economic factors, as well as the potential impact of general economic conditions, including inflation, rising interest rates and instability in the banking system, on DraftKings' and Jackpocket's liquidity, operations and personnel, as well as the risks, uncertainties, and other factors described in "Risk Factors" in DraftKings with the SEC, which are available on the SEC's website at www.sec.gov. Additional information will be made available in other filings that DraftKings makes from time to time with the SEC. The forward-looking statements contained herein are based on the DraftKings management's current expectations and beliefs and speak only as of the date hereof, and neither DraftKings nor Jackpocket makes any commitment to update or publicly release any revisions to forward-looking statements in order to reflect new information or subsequent events, circumstances or changes in expectations, except as required by law.

Non-GAAP Financial Measures

This press release includes Adjusted EBITDA, which is a non-GAAP financial measure that DraftKings uses to supplement its results presented in accordance with U.S. generally accepted accounting principles ("GAAP"). The Company believes Adjusted EBITDA is useful in evaluating its operating performance, similar to measures reported by its publicly-listed U.S. competitors, and regularly used by security analysts, institutional investors and other interested parties in analyzing operating performance and prospects. Adjusted EBITDA is not intended to be a substitute for any GAAP financial measure, and, as calculated, may not be comparable to other similarly titled measures of performance of other companies in other industries or within the same industry.

DraftKings defines and calculates Adjusted EBITDA as net income (loss) before the impact of interest income or expense (net), income tax provision or benefit, and depreciation and amortization, and further adjusted for the following items: stock-based compensation; transaction-related costs; litigation, settlement and related costs; advocacy and other related legal expenses; gain or loss on remeasurement of warrant liabilities; and other non-recurring and nonoperating costs or income.

DraftKings includes non-GAAP financial measures because they are used by management to evaluate the Company's core operating performance and trends and to make strategic decisions regarding the allocation of capital and new investments. Adjusted EBITDA excludes certain expenses that are required in accordance with GAAP because they are non-recurring items (for example, in the case of transaction-related costs and advocacy and other related legal expenses), non-cash expenditures (for example, in the case of amortization of acquired intangible assets, depreciation and amortization, remeasurement of warrant liabilities and stock-based compensation), or non-operating items which are not related to the Company's underlying business performance (for example, in the case of interest income and expense and litigation, settlement and related costs).

Information reconciling forward-looking fiscal year 2026 Adjusted EBITDA and fiscal year 2028 Adjusted EBITDA to its most directly comparable GAAP financial measure is unavailable to DraftKings without unreasonable effort due to, among other things, certain items required for such reconciliation being outside of DraftKings' control and/or not being able to be reasonably predicted. Preparation of such reconciliation would require a forward-looking balance sheet, statement of income and statement of



cash flow, prepared in accordance with GAAP, and such forward-looking financial statements are unavailable to the Company without unreasonable effort. DraftKings provides a range for its Adjusted EBITDA forecast that it believes will be achieved; however, the Company cannot provide any assurance that it can predict all of the components of the Adjusted EBITDA calculation. DraftKings provides a forecast for Adjusted EBITDA because it believes that Adjusted EBITDA, when viewed with DraftKings' results calculated in accordance with GAAP, provide useful information for the reasons noted above. However, Adjusted EBITDA is not a measure of financial performance or liquidity under GAAP and, accordingly, should not be considered as an alternative to net income (loss), gross profit, or cash flow from operating activities or as an indicator of operating performance or liquidity.

Contacts

Media:

Media@DraftKings.com @DraftKingsNews

Investors:

Investors@DraftKings.com